| 272-101 | | | · · · · · · · · · · · · · · · · · · · |
|---|--|--|---|
| PAGE | 1. NEPONT NO. NSF/CEE-83013 | 3 | PB8 4 118272 |
| . Title and Subtitle Dynamic Analysis o | of Multiply Tuned and Ar | sitrarily Supported | July 1983 |
| Secondary Systems | | | |
| Author(s) Takony Tours and A | man Dan Kiumakina | | Performing Organization Rost. No. |
| Performing Organization Name | and Address | | DUD/EERC-03/U/ |
| Earthquake Enginee | ring Research Center | | |
| University of Cali 47th Street & Hoff | fornia, Berkeley Fman Blud | 1 | I. Contract(C) or Grant(G) No. |
| Richmond, Calif. 9 | 14804 | a | > |
| | | | * CEE-8105790 |
| . Sponsoring Organization Nam | ve and Address | 1 | 1. Type of Report & Periori Covered |
| National Science F | oundation | | |
| 100 G Street, N.W | 20550 | 1 | L. |
| washington, D.C. | | | |
| · 34.)piementary notes | | | |
| • | | | |
| Abstract (Lunit: 200 words) | | | |
| between motions of resonance or tunin or more frequencie ing ratio of the t consider the combi Such an approach i resonance, and non complexity of the classically damped example systems an | the support points and ig phenomena when a set is of the other system; wo subsystems are diffe- ned primary and seconda mplicitly includes the i-classical damping, but resulting eigenvalue pi I. The main results of id compared with results | of modal responses for stoc of frequencies of one sys and non-classical damping erent. The basic approach by subsystems as a single effects of interaction, m t was avoided in the past roblem and the fact that s the analysis are applied s obtained from numerical | hastic input; tem is tuned with one effects when the damp- of the analysis is to dynamic assemblage. ultiple support motions, due to the size and uch systems are non- to several representative analysis. |
| Document Analysis 4. Desc | riptors | | |
| b. Identifiers/Open-Ended Te | ***** | | |
| e. COSATI Field/Group | | 1 | |
| Availability Statement | · · · · · · | 23. Scourtly Close (This S | lopert) 21. No. of Pages 24.() |
| · Release U | Inlimited | 38. Security Class (This F | vge) 22. Price |
| e AMS-239.18) | See Inc | trustions on Revense | OPTIONAL FORM 272 (4-77) |

Department of Commerce

ED34-178-12

REPORT NO. UCB/EERC-83/07 JULY 1983

EARTHQUAKE ENGINEERING RESEARCH CENTER

DYNAMIC ANALYSIS OF MULTIPLY TUNED AND ARBITRARILY SUPPORTED SECONDARY SYSTEMS

by

TAKERU IGUSA ARMEN DER KIUREGHIAN

Report to the National Science Foundation and the Electric Power Research Institute



UNIVERSITY OF CALIFORNIA · Berkeley, California

NATIONAL TECHNICAL INFORMATION SERVICE US DEPARTMENT OF COMMERCE SPRINGER LD. VA 22161

DYNAMIC ANALYSIS OF MULTIPLY TUNED AND ARBITRARILY SUPPORTED SECONDARY SYSTEMS

Ву

Takeru Igusa

Assistant Research Engineer University of California, Berkeley

Armen Der Kiureghian

Associate Professor of Civil Engineering University of California, Berkeley

A report on research sponsored by the National Science Foundation and the Electric Power Research Institute

Report No. UCB/EERC-83/07

Earthquake Engineering Research Center College of Engineering University of California, Berkeley

July 1983

1.12.

DYNAMIC ANALYSIS OF MULTIPLY TUNED AND ARBITRARILY SUPPORTED SECONDARY SYSTEMS

ABSTRACT

The subject of this study is the dynamic analysis of linear, multi-degree-of-freedom (MDOF) secondary subsystems with multiple support points attached to linear MDOF primary subsystems. It is known that such systems possess a number of important and complicated dynamic characteristics. These characteristics include: interaction between the two subsystems; cross-correlations between motions of the support points and modal responses for stochastic input; resonance or tuning phenomena when a set of frequencies of one system is tuned with one or more frequencies of the other system; and non-classical damping effects when the damping ratio of the two subsystems are different. In past research, one or more of these dynamic characteristics have not been given full or adequate attention; the objective of this study is to include all of these characteristics in a dynamic analysis of the complete system.

The basic approach of the analysis is to consider the combined primary and secondary subsystems as a single dynamic assemblage. Such an approach implicitly includes the effects of interaction, multiple support motions, resonance, and non-classical damping, but was avoided in the past due to the size and complexity of the resulting eigenvalue problem and the fact that such systems are non-classically damped. These problems are resolved in this study in the following manner: (1) A modal decomposition method is developed for non-classically damped systems with closed-form expressions for combining modal responses. Derivations are included for stationary stochastic input specified by the power spectral density or response spectra and indications are given for considering non-stationary input. (2) Perturbation methods are systematically applied to the analysis of the complex-valued eigenvalue problem to reduce the analysis into a physically meaningful and mathematically manageable form. Expressions are

subsequently derived for the modal properties, which are in closed form for all but multiply tuned modes

The analysis of the systems follows a logical development, beginning with the simplest 2-DOF system and progressing to the most general MDOF system. For completeness, frequency response function analysis is presented and compared with modal decomposition results. Also, as one extension of the theory, the results are simplified to the important case of noninteracting subsystems where the secondary subsystem masses are sufficiently small in comparison with the primary subsystem masses. Finally, the main results of the analysis are applied to several representative example systems and compared with results obtained from numerical analysis. Favorable agreement is found for all cases.

ACKNOWLEDGMENTS

The authors would like to express special thanks to Professor Jerome L. Sackman for his providing the initial impetus for our interest in this problem and for many resourceful and enlightening discussions that we have had with him during the course of this study.

This research work was sponsored by the National Science Foundation under Grant No. CEE-8105790 and the Electric Power Research Institute under Grant No. RP964-8. This sponsorship is gratefully acknowledged.

TABLE OF CONTENTS

| ABSTRACT | i |
|--|--|
| ACKNOWLEDGEMENTS | 131 |
| TABLE OF CONTENTS | 1N |
| LIST OF TABLES | NI. |
| LIST OF FIGURES | vń |
| 1. INTRODUCTION | I |
| 1.1 General Remarks 1.2 Description of the Problem 1.3 Literature Survey 1.4 The Perturbation Approach 1.5 Scope and Limitations | 1 4 7 10 |
| 1.6 Approach of Analysis | 11 |
| 2. MODEL DECOMPOSITION METHOD FOR STOCHASTIC RESPONSE OF NON-CLASSICALLY DAMPED SYSTEMS | 14 |
| 2.1 Introduction 2.2 Response to Base Input in Time Domain 2.3 Response to Base Input in Frequency Domain 2.4 Spectral Moments of Response 2.5 Response to White-Noise Input 2.6 Examples 2.7 Development of the Response Spectrum Method 2.8 Summary and Conclusions | 14 17 22 23 27 30 32 35 |
| 3. ANALYSIS OF THE BASIC 2-DEGREE-OF-FREEDOM EQUIPMENT- STRUCTURE SYSTEM | 45 |
| 3.1 Introduction 3.2 Definitions 3.3 Frequency Response Function Approach for Tuned Systems 3.4 Modal Decomposition Approach 3.5 Non-Interaction Results 3.6 Summary | 45 46 49 59 70 73 |
| 4. ANALYSIS OF A 3DOF EQUIPMENT ATTACHED TO A MDOF STRUCTURE | 99 |
| 4.1 Introduction | 99 |

| 4.2 Definitions | 100 |
|--|-----|
| 4.3 Frequency Response Function Approach | 106 |
| 4.4 Modal Decomposition Approach | 111 |
| 4.5 Non-Interaction Results | 122 |
| 4.6 Floor Spectra | 128 |

| 5. ANALYSIS OF MOOF SECONDARY SYSTEMS ATTACHED TO SDO | |
|---|-----|
| PRIMARY SYSTEMS | 150 |
| S.1 Introduction | 150 |
| 5.2 Definitions | 150 |
| 5.3 Frequency Response Results | 153 |
| 5.4 Modal Decomposition Results | 154 |
| 5.5 Non-Interaction Results | 159 |

6. ANALYSIS OF MOOF SECONDARY SYSTEMS ATTACHED TO MOOF PRIMARY SYSTEMS 182 182 6.1 Introduction 182 6.2 Definitions 6.3 Frequency Response Function Approach 186 6.4 Modal Decomposition Approach 186 6.5 Non-Interaction Results 193 7. SUMMARY AND CONCLUSIONS 212 7.1 Summary of Thesis 212 7.2 Conclusions and Recommendations 213

215

8. REFERENCES

• v -

LIST OF TABLES

| Table | Description | Page |
|-------|--|------|
| 2.1 | Physical Properties of Example System | 37 |
| 2.2 | Modal Properties of Example System | 37 |
| 23 | Analysis Results: Equipment Response | 37 |
| 3.1 | Frequencies for Example System | 76 |
| 3.2 | Mode Shape Component α of Example System | 76 |
| 4 } | Physical Properties of the MDOF/SDOF Example System | 130 |
| 4.2 | Modal Properties of the Fixed Base Subsystems | 130 |
| 4.3 | Frequencies for Example System (ϵ_{11} =0.01) | 130 |
| 4.4 | Mode Shapes of Example System (ϵ_{11} =0.01, ω_{11} =1.0) | 131 |
| 5.1 | Physical Properties of the SDOF/MDOF Example System | 163 |
| 5.2 | Modal Properties of the Fixed Base Subsystems | 163 |
| 5.3 | Frequencies for Example System ($\epsilon_{11}=0.01$) | 163 |
| 5.4 | Mode Shapes of Example System (ϵ_{11} =0.01, ω_{∞} =1.0) | 164 |
| 6.1 | Physical Properties of the MDOF/MDOF Example System | 198 |
| 6.2 | Modal Properties of the Fixed Base Subsystems | 198 |
| 6.3 | Frequencies for Example System (ϵ_{11} =0.01) | 198 |
| 6.4 | Mode Shapes of Example System (ϵ_{11} =0.01, ω_{c1} =1.0) | 199 |

- vi -

LIST OF FIGURES

| Figure | Description | Page |
|--------|---|------|
| 2.1 | One Cycle of an Example Complex-Valued Mode Shape | 38 |
| 2.2 | Coefficients $p_{0,q}, p'_{0,q}$, and $\eta_{1,q}$ for Response to White-Noise Input | 39 |
| 2.3 | Two Degree-of-Freedom Example System | 40 |
| 2.4 | Response Parameters δ and λ_{0} for Soil-Structure Study | 41 |
| 2.5 | Response Parameters δ and $\lambda_{\rm b}$ for Equipment-Structure Study | \$2 |
| 2.6 | Contribution of Terms to λ_0 in Example Studies | 43 |
| 2.7 | Example System for Response Spectrum Study | 44 |
| 3.1 | Two Degree-of-Freedom Equipment-Structure System | 77 |
| 3.2 | Perturbation Approximations | 78 |
| 3.3a | Tuned Complex Frequencies: General Configuration | 79 |
| 3.3b | Tuned Complex Frequencies: Perfect Tuning | 80 |
| 3.3c | Tuned Complex Frequencies: Equal Damping | 81 |
| 3.4 | Transfer Function $T_{ij}(\omega)$: Tuned System | 82 |
| 3.5.a | Spectral Moment λ_0 | 83 |
| 3.5 b | Spectral Moment A | 84 |
| 3.5.c | Spectral Moment λ_2 | 85 |
| 3.5.d | Spectral Moment λ_0 versus Mass Ratio ϵ | 86 |
| 3.5.e | Spectral Moment λ_0 versus Damping Difference ξ_d | 87 |
| 3.6a | Mean Zero-Crossing Rate ν | 88 |
| 3.6b | Shape Factor 8 versus Mass Ratio e | 89 |
| 3.6c | Shape Factor δ versus Detuning β | 90 |
| 3.7a | Mode Shape Component α_i : General Configuration | 91 |
| 3.7b | Mode Shape Component α_i : Perfect Tuning | 92 |

| 3.7c | Mode Shape Component α : Equal Damping | 93 |
|-------|--|------------|
| 3.8 | Spectral Moment λ_0 : Detuned System | 94 |
| 3.9 | General Formula for Spectral Moments | 9 5 |
| 3.10 | Error for Defuned Spectral Moments | 96 |
| 3.11 | Spectral Moments for Non-Interaction | 97 |
| 3.12 | Non-interaction Error in Spectral Moment λ_{μ} | 98 |
| 4 1 | Example of ζ_{\pm} Calculation | 131 |
| 4.2 | Example MDOF/SDOF System | 132 |
| 43 | Distribution of Subsystem Free Vibration Frequencies | 133 |
| 4.4.4 | Transfer Function $T_{ii}(\omega)$: Singly Tuned System | 134 |
| 4.4b | Transfer Function $T_{ii}(l\omega)$: Multiply Tuned System | 135 |
| 4.5a | Spectral Moment λ_0 derived from $H_0(\omega)$ | 136 |
| 4.5b | Spectral Moment λ_1 derived from $H_{i,i}(\omega)$ | 137 |
| 4.5c | Spectral Moment λ_2 derived from $H_{ij}(\omega)$ | 158 |
| 4.6a | Complex Frequencies: Singly Tuned System | 139 |
| 4.6b | Complex Frequencies. Multiply Tuned System | 140 |
| 4.6c | Complex Frequencies: Deturied System | 141 |
| 4.7a | Spectral Moment λ_0 | 142 |
| 4 7b | Spectral Moment λ_3 | 143 |
| 4.7c | Spectral Moment λ_2 | 144 |
| 4.8a | Mean Zero-Crossing Rate v | 145 |
| 4.8b | Shape Factor 8 | 146 |
| 4.9 | Transfer Function $\mathcal{T}_{\Omega}^{(non)}(\omega)$ without Interaction | 147 |
| 4.10 | Spectral Moment $\lambda_{tt}^{(mw)}$ without Interaction | 148 |
| 4.]] | Spectral Moment $\lambda_0^{(max)}$ without Interaction | 149 |
| 5.1 | Example SDOF/MDOF System | 165 |
| 5.2 | Distribution of Subsystem Free Vibration Frequencies | 166 |

a same

- vili-

| 5.3a | Transfer Function $T_{\mu}(\omega)$: Singly Tuned System | 167 |
|-------------------|--|-----|
| 5.3b | Transfer Function T_{ij} (ω): Multiply Tuned System | 168 |
| 5.4a | Spectral Moment λ_0 derived from $H_{\mu}(\omega)$ | 169 |
| 5.4b | Spectral Moment λ_1 derived from $H_{ij}(\omega)$ | 170 |
| 5.4c | Spectral Moment λ_2 derived from $H_{j1}(\omega)$ | 171 |
| 5.5a | Complex Frequencies: Singly Tuned System | 172 |
| 5.5b | Complex Frequencies: Multiply Tuned System | 173 |
| 5. 6 a | Spectral Moment λ_0 | 174 |
| 5.6b | Spectral Moment λ_1 | 175 |
| 5.6c | Spectral Moment λ_2 | 176 |
| 5.7a | Mean Zero-Crossing Rate v | 177 |
| 5.7ь | Shape Factor 8 | 178 |
| 5.8 | Transfer Function $T_{ii}^{(non)}(\omega)$ without Interaction | 179 |
| 5.9 | Spectral Moment $\lambda_0^{(non)}$ without Interaction | 180 |
| 5.10 | Spectral Moment $\lambda_0^{(min)}$ without Interaction | 181 |
| 6.1 | Example MDOF/SDOF System | 200 |
| 6.2 | Distribution of Subsystem Free Vibration Frequencies | 201 |
| 6.3 a | Complex Frequencies: Singly Tuned System | 202 |
| 6.3b | Complex Frequencies: Multiply Tuned System | 203 |
| 6. 4 a | Spectral Momen: λ_0 | 204 |
| 6.4b | Spectral Moment λ_1 | 205 |
| 6.4c | Spectral Moment λ_2 | 206 |
| 6.5a | Mean Zero-Crossing Rate ν | 207 |
| 6.5b | Shape Factor 8 | 208 |
| 6.6 | Transfer Function $T_{in}^{(non)}(\omega)$ without Interaction | 209 |
| 6.7 | Spectral Moment $\lambda_0^{(non)}$ without Interaction | 210 |
| 6.8 | Spectral Moment $\lambda_0^{(mn)}$ without Interaction | 211 |

-

CHAPTER 1

INTRODUCTION

1.1 General Remarks

Complex structural systems composed of light secondary components or subsystems attached to heavier primary subsystems are frequently encountered in civil engineering. Piping systems, electrical equipment, laboratory instruments, antennas, computer hardware, and safety devices are a few examples of the attachments which may be found in multi-story buildings and industrial plants and facilities. In many cases the secondary subsystems perform vital tasks or are valuable in themseives and the response of these systems to base excitations have long been of engineering interest.

In the following, the properties of such systems are described, and a detailed literature survey of the studies on these systems is presented, concentrating on the methodology used and the difficulties encountered in the analysis. Then a method to overcome these difficulties is introduced and a plan of analysis, which constitutes the main body of this study, is outlined.

1.2 Description of the Problem

1.2.1 Characteristics of a General Primacy-Secondary System

Primary-Secondary (PS) systems have a variety of different forms and characteristics, and in this thesis, those systems that possess certain well-defined, standard properties are studied. These properties are as follows:

- 1. The primary and secondary subsystems are viscously and classically damped, linear-elastic systems.
- 2. The mass of the secondary subsystem is considerably smaller than that of the primary subsystem (mathematical criteria are given in Chapter 3).

- 3. The system does not include the interaction effects of a foundation, cascaded tertiary or higher level subsystems, or other separate secondary subsystems that may be attached to the primary subsystem. It is assumed that any such effects, if significant, are accounted for by properly modifying the input excitation (in the case of interacting foundation) or the properties of the primary subsystem (in the case of other secondary subsystems).
- 4. All attachment points of the PS system to the base are subjected to the same input excitation.

PS systems that have the above properties are quite general and have a number of important dynamic characteristics which have been topics of intense study in the past. The main dynamic characteristics are:

- Multi-degree-of-freedom (MDOF) subsystems: Both primary and secondary subsystems are in general MDOF systems. Either subsystem can be composed of more than one independent subsystem.
- General attachment configuration: The secondary subsystem may be attached to the primary subsystem in an arbitrary number of locations and may also be attached to the base of the combined system. The secondary subsystem may be attached to more than one primary subsystem.
- 3. General resonance (tuning) characteristics: Any number of the frequencies of one subsystem may be arbitrarily close to or coincident with the frequencies of the other subsystem. This condition is known as tuning. When a group of closely spaced frequencies of any subsystem is tuned with one or more frequencies of the other subsystem, the frequencies are said to be in multiple tuning.
- 4. Dynamic interaction: For arbitrary PS systems, the primary and secondary subsystems interact with each other, particularly if the secondary subsystem modal masses are non-negligible and the frequencies of the two subsystems are juned.
- 5. Stochastic correlation: For stochastic input, the effects of correlation cannot be neglected,

والمراجع المراجع المراجع

and for PS systems, this effect is an important, integral part of their dynamic behavior. For tuned systems, the cross-correlations between modul responses is significant and must be accurately determined in any dynamic analysis. Also, motions at various support points of the secondary subsystem are, in general, correlated.

 Non-classical damping: This effect occurs in systems which have different damping ratios in the primary and secondary subsystems and is particularly significant at tuning.

1.2.2 Common Restrictions and Approximations

In principle, the theoretically exact response of a general secondary subsystem which includes the above effects can be obtained using standard methods of analysis on the combined **PS** system. However, this procedure presents a number of difficulties. The number of degrees of freedom (DOF) of the combined system is usually prohibitively large and the differences in the magnitudes of the stiffness, damping, and mass terms between the primary and secondary subsystems pose serious numerical problems [48]. Also, the analysis and design of the secondary subsystem is usually performed well after the design and analysis of the primary subsystem. Consequently, alternative methods of analysis for this problem are required.

In general, isosearchers have used several well-defined approximations and restrictions on the physical and dynamic properties of PS systems to reduce the analysis to a simpler and more manageable form. In these restrictions and approximations, one or more of the six dynamic characteristics of PC against listed above are not given full or accurate consideration, or are neglected altogether. For instance, some studies neglect interaction, some consider secondary systems with only a single DOF, and others assume that none of the modes are tuned. Many times, a combination of restrictions are used.

In order to present an organized review of the literature, four categories are considered. Studies which make the non-interaction approximation are reviewed first, followed by those works which include the interaction effect. Then, works which consider multiply supported systems are discussed. Finally, in Section 1.4, several recent studies which provide the background to the theory and methods of analysis in this thesis are examined.

1.3 Literature Survey

1.3.1 Non-Interaction Studies

A commonly used method of analysis which neglects interaction is the floor response spectrum method. In this method, the motions of the support points of the secondary subsystem are calculated by time history analyses of the primary subsystem. The response spectra of the support motions are known as floor spectra which are used as input to the secondary subsystem. Realizing that this method is lengthy and inefficient, several authors have developed more direct methods of finding floor spectra using the modal properties of the primary subsystem and the ground response spectrum. Biggs and Roesset [6] developed an empirical rule for finding the floor spectra using the modal properties of the primary subsystem, which was later modified by Kapur and Shao [25] using more inathematically based premises. An alternate approach based on Fourier transforms was developed by Scanlan and Sachs [38]. Singh [40], Chakravarti and Vantuarke [9]. Singh [41], and Vanmarke [46] incorporated ideas from random vibration theory.

All of these methods have been shown to give reasonable accuracy for single-degree-offreedom (SDOF) secondary subsystems with relatively small masses and frequencies that are not tuned to a frequency of the primary subsystem. However, when the two subsystems are tuned to each other, the methods consistently fail. One problem is that the effect of interaction becomes important and as pointed out by Crandall and Mark [11], Singh [41], and Kapur and Shao [25], significant error is introduced in the analysis of such secondary subsystems if this effect is neglected. The other problem is inherent in the analysis. Many formulations that were derived in the above works yield infinite results at tuning. Singh [42] and Peters, et. al. [35] provide rough approximations for perfectly tuned systems, but these results continue to ignore interaction and are inaccurate for nearly juned systems, where the tuned frequencies are not exactly coincident. Since tuning is considered important and, frequently, critical in the analysis of secondary systems [2], it is clear that the above non-interaction studies are not adequate for a proper analysis of such systems.

1.3.2 Studies Considering Interaction

A number of analytical methods have been developed to account for the dynamic interaction between the primary and secondary subsystem. In these methods, the combined equations of motion must be analyzed without decoupling to account for this effect. Crandall and Mark [11] used the exact equations for a combined 2-DOF system to compute the root-mean-square response to stationary excitation. In other studies, approximations are made to reduce and simplify the analysis of the combined system. Penzien and Chopra [34] reduce a system composed of an N-DOF primary subsystem and a single DOF (SDOF) secondary subsystem to a series of N 2-DOF subsystems, however, the analysis is in terms of 2-DOF response spectra, which are not generally available, and the responses are combined with the square-root-of-sum-of-squares (SRSS) rule which is inaccurate at tuning.

Newmark [33] used a notion of effective mass ratio to obtain approximate mode shapes and frequencies and combined modal responses with the conservative absolute sum rule. This method was improved by Nakhata et. al. [31]. These two modal approaches have sound theoretical bases, however a certain level of accuracy is required both in the formulations for the modal properties and the combination of modal response quantities to obtain good approximations for the system response [48]. This accuracy was not attained until later works, as detailed in Section 1.4.

1.3.3 Multiple Support Excitations

As stated earlier, a common assumption in the analysis of PS systems is that the secondary subsystem has only one DOF or is attached to the primary subsystem at a single point. This assumption is satisfactory for simple systems or for obtaining floor spectra. However, for more general secondary subsystems, the important effect of multiple support excitations must be analyzed.

In the earliest research efforts the effect of interaction was not included in the analysis so that simplified decoupled equations of motion could be used. A straightforward method for analyzing decoupled multiply-supported secondary subsystems is the time history approach. The base ame history is used to find the resulting motions of the support points and these motions are subsequently input into the equations of motion for the sciondary subsystem [22]. This method is both inefficient and cumbersome and researchers have investigated alternate methods based on response spectra and random vibration techniques.

Amin. et al. [4]. Shaw [39], and Vashi [47] formulated the response analysis of the secondary subsystem in terms of the motions of the individual support points using heuri-tic procedures for combining these motions. Vashi provides three methods for combining responses one based on the absolute sum method for support motions which are in phase, another based on the SRSS rule for motions which are independent, and the third based on a combination of the two methods for motions which are neither in phase nor independent. No precise procedure is given to determine the classification of the support motions; therefore the analysis becomes subjective in nature.

In the industry, it is considered standard to decompose the response of the secondary subsystem into "inertial" or "dynamic" effects and effects due to "relative seismic support displacements" or "pseudo-static motion" [2]. However, this artificial formulation complicates the analysis, particularly in the consideration of correlation between the numerous decomposed response quantities.

Recently, Lee [26] introduced a method based on frequency response analysis to include the correlation between support motions and modal responses. However, the frequency response formulations are not readily useful in the analysis of secondary subsystems. Der Kiureghian, et. al. [18, 19] used a modal approach to develop a practical response spectrum method of analysis. The basis of their method is the theory of perturbations, which is explained in terms of secondary subsystems in the next section.

Reproduced from best available copy

1.4 The Perturbation Approach

1.4.1 Overview

The perturbation method and its associated theory of asymptotic expansions are long established tools of mathematics [20]. During the past quarter century, applications of these methods have been made in fluid dynamics and related branches of hydraulic engineering [32], which have resulted in considerable developments in these fields. These methods have also been used in various branches of mechanical and structural engineering, however systematic applications of these methods to PS systems has not been carried out until very recently.

1.4.2 Past Applications in PS Systems

Perturbation methods are useful in solving differential equations which contain small parameters. Such equations occur in the analysis of PS systems; the small parameters are the mass, stiffness, and damping terms of the secondary subsystem. Sackman and Kelly [37] were among the first to recognize this and they were able to cast a new form to the equations describing the PS system using perturbation methods. With this new formulation, they were able to accurately analyze tuned 2-DOF systems which, in previous works, were treated in a rough, approximate manner, and obtain closed form expressions for the frequencies. These results were subsequently used in an analysis of more general PS systems with a single DOF secondary subsystem using Laplace transforms.

Other research work in this area has used the modal approach to the archesis of PS systems [36, 48, 18, 19]. Basically, the modal approach is as follows: Perturbation techniques are used to obtain approximations for the mode shapes and frequencies of the combined system. From these modal properties, the modal responses are found which are subsequently combined using a modal combination rule. This approach is standard in the analysis of most linear structural systems [10], however for PS systems special problems are encountered. For systems with even slight differences of the damping ratios in the primary and secondary subsystems, the mode shapes are complex-valued. Also, for tuned subsystems, some of the frequencies of the combined system are very closely-spaced and the corresponding mode shapes are large and nearly opposite in sign. Consequently, the modal responses must be combined in a procise and theoretically sound manner. In the following, a survey of several recent studies based on perturbation applications to modal analysis of PS systems is presented and the difficulties encountered are discussed.

Ruricka and Robinson [36] obtained expressions for the modal properties of fauly general PS systems, however, the important non-classical damping characteristic was given only an approximate consideration. Also, the response analysis was in terms of the Fourier transform of the groand motion, which is generally not available. Villaverde and Newmark [48] concentrated their efforts on systems with only one or two attachment points, which restricted their analysis. Also, the non-classical damping effect was not given precise treatment in their work. Der Kiureghian, et. al. [18, 19] studied systems restricted to single DOF secondary subsystems and classical damping and obtained relatively good results for the modal properties of these systems and derived the response to ground motions specified by their response spectra. Der Kiureghian, et. al. [17] extended this method to account for multiple support excitations by defining cross-floor spectra which are in terms of the cross-correlations, between two SDOF oscillators attached to the same primary subsystem. However, this latter approach in its present formulation does not account for interaction. Also, none of the above itentioned studies considered the important case of multiple tuning.

From the above works, it is clear that the perturbation approach is a powerful tool in analyzing PS systems. However, its use has been limited and its capabilities have only partially been utilized. The important effects of non-classical damping, multiple tuning, and interaction are yet to be given accurate consideration.

1.4.3 Plan of Analysis

In the present thesis, the methods of perturbation analysis are applied to general PS systems in a togical, systematic manner. The emphasis is on completeness and the limitations imposed on the previous works discussed above are fifted. Thus, all of the important characteristics listed in Section 1.2.1 are included in the analysis. The equations of motion are set up for interacting, multi-degree-of-freedom subsystems with general attachment configuration and the modal properties are obtained from a complex eigenvalue formulation which explicitly accounts for non-classical damping and general tuning characteristics of the combined system. Finally, methods of random vibrations are used in combining modal responses to include the effects of correlation between modal responses and between the motions of the attachment points. This analysis procedure appears to be straightforward, however, several important rules must be closely and care/ully followed in the plan of analysis to insure accuracy and completeness in the results

All of the analysis revolves around the complete equations of motion of the combined PS system. The separation of the coordinates into dynamic and static parts are not necessary since all of the coordinates of the primary and secondary subsystems are considered together as *dynamic* responses to the base motion. Also, the behavior at the support point - need not be explicitly analyzed: this complex effect is included implicitly in the analysis since the PS system is considered as a single dynamic assemblage. Thus, by treating the PS system as a whole, the analysis becomes, in many ways, simpler and more direct. This approach was avoided until recently partly because the applicability of perturbation methods in such an analysis was not exploited. However, this application of perturbation methods is by no means trivial, and the studies discussed above in Section 1.4.2 did not fully and accurately apply these methods to the analysis of general PS systems.

The thosis is organized to follow a logical plan of analysis beginning with the simplest PS system where both primary and secondary subsystems have only a single DOF, and progressing to the most general case where both subsystems have multiple DOF. The reason for this is clear, the most general systems are also the most complex and many of the important dynamic characteristics of these systems which must be included in the analysis is obscured by the complexity of the problem. Thus, the simplest PS system is examined first, where most of the results of the analysis is formulated in simple closed form expressions. The important dynamic

characteristics are readily seen and are thoroughly studied before more general systems are studred. In the analysis of more general systems, these same dynamic properties are recognized from the a priori knowledge gained from the analysis of the simpler systems. In this way, the properties listed in Section 1.2.1 are identified in the most general PS systems, studied in detail, and the results of the analysis presented using theoretically derived formulations.

The secondary subsystem is constantly interpreted as a perturbation of the primary subsystem in order to make full use of the perturbation methods of analysis. To achieve this level of analysis, precise and clear definitions are used for the relationships between the terms used in the perturbation approach. Also, the criteria for the application of the perturbation methods are explicitly marked out and strictly adhered to. This helps in maintaining consistency in all of the resulting expressions of the analysis.

Finally, no approximations are used except those that are allowed in the mathematical framework of the theory of perturbation methods. Thus, within this framework, the analysis is exact. Consequently, all of the dynamic properties, such as general tuning characteristics, multiple attachment configuration, non-classical damping, interaction, and all correlations, which were not all adequately accounted for in previous works, are included in the analysis in a mathematically precise manner. This final premise avoids some of the inaccuracies and inherent deliciencies found in earlier works and provides results that are comprehensive and theoretically sound within the size of the analysis.

1.5 Scope and Limitations

In Section 1.2.1, the scope of the type of PS systems that are being considered was described in detail. In this section, the scope of the type of input and the corresponding response quantities is discussed.

In the analysis of PS systems presented in this study, two alternate formulations describing the dynamic properties of these systems are presented. One is in terms of the frequency response function of the response of the secondary subsystem and the other is a modal description based on the mode shapes and frequencies of the combined system. These formulations have applications in a variety of methods of dynamic analysis. Attention here is focused on the response of PS systems to stabonary stochastic input and response to seismic input described by the ground response spectrum.

A method for obtaining the power spectral density (PSD) is provided and expressions for the first few moments of the PSD are derived. Most practical response quantities can be found from these moments, including the mean square of the response and its time derivative, and for Gaussian excitation, the response mean frequency, distribution of the peak response over a specified duration and its mean and variance, and the distribution of the peaks of the response. These results form the basis for the response spectrum method for seismic analysis of PS systems.

Once the characteristics of the combined systems are determined, the response of PS systems can be found for more general inputs such as non-stationary inputs and procedures for these extensions are indicated in the text, however, a detailed analysis for such inputs is beyond the intended scope of this report.

1.6 Approach of Analysis

From the discussion in Section 3.4.2, it is clear that a modal combination rule which correctly accounts for correlation between modes with closely spaced frequencies and is general enough to analyze non-classically damped systems is necessary in the modal analysis of PS systems. Consequently, in the first step of this study, a modal combination rule for stationary response satisfying these properties is developed. The method is a generalization of that for classically damped systems [19]. The remainder of the study follows the ideas outlined in Section 1.4.3. A synopsis of the approach of the analysis follows:

Chapter 2. Modal Decomposition Method for Stochastic Response of Non-Classically Damped Systems

Based on the complex mode shapes and frequencies of non-classically damped systems, general formulae for the response PSD and its moments are derived. The expressions are in terms of the cross-spectral moments between the modes of the system. Comparisons are made with results for classically damped systems and closed form solutions for the moments are obtained for white-noise input. These results are used to formulate a response spectrum method for seismic analysis of general systems with nonclassical damping.

Chapter 3. Analysis of the Basic 2-DOF PS System

The basic 2-DOF system is the simplest PS system and consists of a SDOF secondary subsystem attached to a SDOF primary subsystem. The frequency response approach is used to derive accurate closed form expressions for the PSD and its first three moments. After a careful derivation and investigation of the mode shapes and frequencies, both tuned and defuned systems are analyzed using the modal decomposition approach and a criterion for tuning is mathematically derived. A similar analysis for very light equipment is carried out and a mathematical criterion for decoupling between the primary and secondary subsystems is derived.

Chapter 4. Analysis of SDOF Equipment Attached to MDOF Structures (MDOF/SDOF PS System)

The methods developed for the analysis of the 2-DOF system are generalized for MDOF/SDOF systems. There are more complex problems to be handled such as multiple support excitations and multiple tuning, however the framework established through the perturbation approach effectively solves these problems, as mentioned earlier.

Chapter 5. Analysis of MDOF Secondary Subsystems Attached to SDOF Primary Subsystems (SDOF/MDOF PS System)

The results for MDOF/SDOF systems are found to be closely related to those for SDOF/MDOF systems. This relation is explored and utilized to derive expressions for

the response of SDOF/MDOF systems based on the results of Chapter 4.

Chapter 6. Analysis of MDOF Secondary Subsystems Attached to MDOF Primary Subsystems (MDOF/MDOF PS System)

All of the results derived in the previous chapters form the theoretical background for analyzing this most general and complex class of PS systems. By adhering to the rules laid out in Section 1.4.3, the analysis encompasses all c, the dynamic characteristics described above. Multiple support configurations, general tuning, non-classical damping, and interaction are implicitly included in the equations of motion and the corresponding eigenvalue problem, and all correlation effects are explicitly accounted for in the combination of modal responses.

Numerical examples are presented for each of the above types of PS systems to illustrate the accuracy of the proposed methods.

CHAPTER 2

MODAL DECOMPOSITION METHOD FOR STOCHASTIC RESPONSE OF NON-CLASSICALLY DAMPED SYSTEM5

2.1 Introduction

In dynamic analysis of linear systems, such as structures subjected to seismic excitations, it is common to assume that the system is classically damped. Under this assumption, the equations of motion can be transformed into a set of independent modal equations using the real-valued eigenvectors and eigenvalues of the undamped system [10]. However, in most real systems the modal equations are coupled by the damping matrix [49]; these systems are defined to be non-classically damped. In practice, non-classically damped systems can be approximated by a classically damped system, and the results are usually of sufficient accuracy. One common approximation is to neglect coupling damping terms in the modal equations. There are important situations, however, where the effect of non-classical damping is essential and must be included in the analysis [49]. As indicated in the introduction, this effect is found in PS systems and numerical examples are presented in this chapter.

A classical method for analyzing non-classically damped systems is to use modal decomposition employing the complex-valued eigenvectors and eigenvalues [21]. This method has been used in certain applications of deterministic dynamic analysis such as in soil-dynamics where the effect of non-classical damping is significant. Investigators have also studied response to random input. Caughey [8] and Masti [30] considered non-stationary input and responses and Debchandhury and Gasparini [14] used a state space approach to the problem. Many of the results of the analysis of PS systems in the succeeding chapters can be used in the above mentioned methods for finding the response to deterministic or non-stationary random input, however the final results are relatively complex and are not obtainable in closed form. Singh [43] applied the modal decomposition approach to the analysis of the response of non-classically damped systems and was able to derive a response spectrum method for such systems. However, his formulations are in terms of a set of four simultaneous equations which are difficult to interpret physically and are unsuitable for analytical purposes where closed form solutions are sought. In this chapter, a medal decomposition approach is used to develop an alternate response spectrum method that is more simple and direct than Singh's method in that the response quantities are given in closed form expressions directly in terms of the modal properties of the system. This formulation is particularly well-suited for the analysis of general PS systems which is carried out in the succeeding chapters. Another advantage of this method is that a full probabilistic description of the response is given rather than only the mean of the peak response as given in Singh's work

The emphasis in this study is on a second-moment characterization of the stationary response, v(t), of non-classically damped systems under random excitation which is given by the one-sided power spectral density function, $G_{ii}(\omega)$. It is well known [15,44] that most response quantities of engineering interest can be expressed in terms of the first few spectral moments

$$\lambda_{m} = \int_{0}^{\infty} \omega^{m} G_{m} d\omega, \quad m = 0, 1, 2, \cdots$$
 (1)

For example, λ_0 and λ_2 are the mean squares of the response and its time-rate respectively [28]. If the excitation is Gaussian, as is assumed for most engineering applications, the power spectral density provides a complete characterization of the response process. Many additional response quantities in that case are given in terms of the spectral moments λ_{in} . For example, $\overline{\omega} = \sqrt{\lambda_2/\lambda_0}$ denotes the mean frequency of the response process [28] and $\delta = \sqrt{1 - \lambda_1/(\lambda_0 \lambda_2)}$, known as the shape factor, is a measure of the narrow-bandedness of the response power spectral density shape [44]. (δ ranges between 0 and 1 and decreases with increasing narrowbandedness.) The three moments λ_{0x} , λ_1 , and λ_2 also describe several properties of the envelope process associated with the response [29]. Using the envelope process, expressions have been derived for the distribution of the peak response over a specified duration [45] and its mean and variance [13, 15] in terms of these three moments. These expressions are used in deriving the response spectrum method for non-classically damped systems. For fatigue study applications, the moments λ_0 , λ_2 , and λ_4 may be used to find the distribution of the peaks of the response [7].

Expressions for the spectral moments of the response of a multi-degree-of-freedom system subjected to stationary input have been derived [15] for the case of classical damping using a modal decomposition approach. In this chapter, these results are generalized to systems with non-classical damping. First, the basic equations of motion and their modal coordinates are laid out using standard methods of analysis. These equations are integrated and solved in terms of Duhamel integrals and their derivatives. From this formulation, the auto-correlation function of the response process is obtained which is used to derive the power spectral density function and its moments in terms of the modal properties of the original system. Attention is focused on the important case of white-noise input, for which exact closed form expressions and simple first-order approximations are derived. These expressions are subsequently used to formulate a response spectrum method using the theory of stationary random vibrations [22]. Throughout the analysis, results for non-classical damping are compared with previous results for classical damping.

Example studies for simple systems have shown that the effect of non-classical damping can be highly significant in two types of systems: soil-structure systems, where the difference of damping ratios is very large; and nearly tuned equipment-structure systems, where the mass ratio is small and the damping ratios are unequal. It is found that in each case as the difference of the damping ratios of the two subsystems increases or the mass ratio decreases, the classically damped approximation to the response becomes less accurate. Numerical examples for several typical systems are presented which indicate significant errors associated with neglecting the effect of non-classical damping.

The results of this chapter form the foundation of the modal analysis of PS systems in the remainder of this study. The various formulations derived here which are in terms of the com-

plex eigenvalue analysis of the system are particularly well-suited to the analysis of general multiply-supported, multiply-tuned secondary systems.

2.2 Response to Base Input in Time Domain

First, the equations of motion are laid out and, for the sake of review, the solution is derived for classically damped systems. Let **M**, **C**, and **K** be the mass, damping, and stiffness matrices of an *n*-degree-of-freedom, viscously damped linear system. For a base acceleration $\tilde{x}_{c}(t)$, the equation for the system relative displacement response $\mathbf{x}(t)$ is

$$\mathbf{M}\ddot{\mathbf{x}} + \mathbf{C}\dot{\mathbf{x}} + \mathbf{K}\mathbf{x} = -\mathbf{M}\mathbf{r}\dot{\mathbf{x}}_{0}(t) \tag{2}$$

where **r** is the influence vector that couples the ground motion to the degrees of freedom of the structure. The mode shapes, ϕ , and natural frequencies, ω , $i=1,2,\cdots,n$, associated with the undamped system are found from the following $n \times n$ eigenvalue problem

$$\mathbf{K}\boldsymbol{\phi}_{1} = \boldsymbol{\omega}_{1}^{2}\mathbf{M}\boldsymbol{\phi}_{1} \tag{3}$$

If the damping matrix C is orthogonal with respect to the undamped mode shapes ϕ , the system is said to be classically damped, and the equations of motion can be decoupled into *n* modal equations. This is accomplished by using the transformation $\mathbf{x} = \Phi \mathbf{u}$ in Eq.2, where $\Phi = [\phi_1 \phi_2 \cdots \phi_n]$, and in turn premultiplying that equation by $\phi_1^{(i)}$. The *i*th decoupled modal equation is of the form

$$\ddot{u}_i + 2\xi_i \omega_i \dot{u}_i + \omega_i^2 u = -\Gamma_i \ddot{\chi}_i (I)$$
(4)

in which $\Gamma_{i} = (\phi_{i}^{j} \mathbf{M} \mathbf{r}) / M_{i}$ is the modal participation factor and $\xi_{i} = \phi_{i}^{j} \mathbf{C} \phi_{i} / 2\omega_{i} M_{i}$ is the estimated damping ratio for mode *i*, where $M_{i} = \phi_{i}^{j} \mathbf{M} \phi_{i}$ is the modal mass. The homogeneous solution of Eq.4 is of the form $\exp(s_{i} t)$ where

$$s_i = -\xi_i \omega_i \pm i \omega_{D_i}, \quad \omega_{D_i} = \omega_i \sqrt{1 - \xi_i^2}$$
 (5)

and the solution for an input $\tilde{x}_{c}(t)$ is given by

$$u_{i}(t) = -\frac{\Gamma_{i}}{\omega_{D_{i}}} \int_{0}^{t} \dot{x}_{k}(\tau) \exp\{-\xi_{i}\omega_{i}(t-\tau)\} \sin\omega_{D_{i}}(t-\tau) d\tau = -\Gamma_{i}h_{i}(\tau)$$
(6)

where $h_i(t)$ is the well-known Duhamel integral

$$h_i(i) = \frac{1}{\omega_D} \int_0^t \widetilde{X}_k(\tau) \exp[-\xi_i \omega_i(t-\tau)] \sin \omega_D(t-\tau) d\tau$$
(7)

In general, the response quantity of interest is a linear combination of the components of the displacement vector \mathbf{x} ,

$$v = \mathbf{q}^T \mathbf{x} = \sum_{i=1}^{n} \mathbf{q}^T \boldsymbol{\phi} \ u = \sum_{i=1}^{n} \mathbf{q}^T \boldsymbol{\phi} \ \Gamma \ h_i(t) = \sum_{i=1}^{n} \psi_i h_i(t)$$
(8)

where **q** is an *n*-vector of constants and $\psi \equiv \mathbf{q}^T \phi$, Γ is known as the effective modal participation factor.

For the general case of non-classical damping, the above modal equations cannot be used and the classical mathematical approach $\{21\}$ to solve Eq.2 is to reformulate it into a first-order 2n-dimensional equation

$$\mathbf{A}\mathbf{\dot{y}} + \mathbf{B}\mathbf{y} = \mathbf{F}\mathbf{\ddot{x}}_{c}(t) \tag{9}$$

where,

$$\mathbf{A} = \begin{bmatrix} 0 & \mathbf{M} \\ \mathbf{M} & \mathbf{C} \end{bmatrix}, \ \mathbf{B} = \begin{bmatrix} -\mathbf{M} & 0 \\ 0 & \mathbf{K} \end{bmatrix}, \ \mathbf{F} = \begin{bmatrix} 0 \\ -\mathbf{M}\mathbf{r} \end{bmatrix}, \text{ and } \mathbf{y} = \begin{bmatrix} \dot{\mathbf{x}} \\ \mathbf{x} \end{bmatrix}$$
(10)

The associated eigenvalue equation is

$$\mathbf{B}\hat{\boldsymbol{\phi}}_{1} = -\mathbf{s}_{1}\mathbf{A}\hat{\boldsymbol{\phi}} \tag{11}$$

Due to the symmetry of the matrices **A** and **B**, the solutions $\{\hat{\phi}_{i}, s_{i}\}$ occur in conjugate pairs. From the definition of y it is clear that $\hat{\phi}_{i}$ is of the form

$$\hat{\boldsymbol{\phi}}_{i} = \begin{cases} s_{i} \boldsymbol{\phi}_{i} \\ \boldsymbol{\phi}_{i} \end{cases}$$
(12)

Also, for the sake of comparison with the case of classical damping, the following notation, which was also used by Singh [43], is introduced:

$$\mathbf{s}_{i} = -\boldsymbol{\xi}_{i}\boldsymbol{\omega}_{i} \pm i\boldsymbol{\omega}_{D_{i}} \tag{13}$$

where $\omega_{i}, \omega_{D_{i}}$, and ξ_{i} are determined by

$$\omega_{i} = |s_{i}|, \quad \xi_{i} = -\operatorname{Re}(s_{i})/|s_{i}|, \text{ and } \omega_{Iii} = \omega_{i}\sqrt{1-\xi_{i}}$$
(14)

In summary, for non-classically damped systems, the eigenvalue problem is complex and of order 2n and is given by Eq.11. The resulting mode shapes and frequencies $\{\phi_{i}, s_{i}\}$ occur in conjugate pairs and can be expressed as in Eqs.12 and 13. For non-classical damping, ϕ_{i} are complex vectors, in the special case of classical damping, ϕ_{i} are real.

At this point, a physical interpretation of the meaning of complex mode shapes and frequencies is in order. The terms ω and ξ are the natural frequencies and damping ratios of the system with the same meaning as for classically damped systems. The complex values for the coordinates of a mode shape lead to different phase angles of the free vibration harmonic motions of the various degrees of freedom of the system [21]. This concept is illustrated in Fig.2.1 which shows a 2-DOF system with a mode shape given by $\phi_1 = [1/i]^i$ and period T_1 at different time intervals during free vibration. It can be observed that the motion of each degree of freedom is out-of-phase with the other.

For a classically damped system, Eq.2 was reduced to *n* decoupled modal equations by transforming to modal coordinates. Likewise, for non-classically damped systems, Eq.9 can be reduced to 2n decoupled modal equations which occur in conjugate pairs by using the modal coordinates z obtained from the transformation $y = \hat{\Phi} z$ where $\hat{\Phi} = [\hat{\phi}_1 \hat{\phi}_2 \cdots \hat{\phi}_{2n}]$. The *i*th modal equation is

$$A\dot{z}_{c} + Bz = \hat{\Phi}^{T} F \ddot{x}_{c}(t)$$
(15)

or alternatively

-

3 1- 3

$$\dot{z} = s_i z_i = r_i \ddot{x}_i(t) \tag{16}$$

where

$$A_{i} = \hat{\phi}_{i}^{\prime} A \hat{\phi}_{i}$$
(17a)

/ . . .

$$B_{i} = \phi_{i} B_{i} \phi_{i} = -s_{i} A_{i}$$

$$(17b)$$

$$\mathbf{r}_{i} = \boldsymbol{\phi}_{i}' \mathbf{F} / A_{i} = -\boldsymbol{\phi}_{i}' \mathbf{M} \mathbf{r} / A \tag{17c}$$

The solution of the first-order equation Eq.16, is the integral

$$z_{i} = r_{i} \int_{0}^{t} \tilde{x}_{s}(\tau) \exp[s_{i}(\tau-\tau)] d\tau$$
(18)

which is analogous to Eq.6 for classically damped systems. Combining modal response quantities as before, one obtains

$$\mathbf{v} = \mathbf{q}^T \mathbf{x} = \sum_{i=1}^{2n} \mathbf{q}^T \boldsymbol{\phi}_i z_i = \sum_{i=1}^{2n} b_i \int_0^t \tilde{x}_v(\tau) \exp[s_i(t-\tau)] d\tau$$
(19)

where

$$b = \mathbf{q}^T \boldsymbol{\phi}_{t} t_{t} \tag{29}$$

Since b and s occur in conjugate pairs, v(t) is always real.

In general, b is complex, however, if the system is classically damped, b is pure imaginary. In fact, for such systems

$$A = -\frac{1}{s}B = -\frac{1}{s}\left[\phi^{2}K\phi - s^{2}\phi^{2}M\phi\right] = \left[\frac{\omega^{2}-s^{2}}{-s}\right]M$$
(21)

where $M = \phi^{t} M \phi^{t}$ is the conventional modal mass. Using Eq.13 for s and substituting the above result in the expression for b yields

$$b = \frac{(\mathbf{q}^T \boldsymbol{\phi}_{-})(\boldsymbol{\phi}_{-}^T \mathbf{M} \mathbf{r})}{\pm 2/\omega_D} = \pm \frac{i\psi_{-}}{2\omega_D}$$
(22)

for systems with classical damping

Up to this point, the analysis has been a review, and all of the above equations can be found in standard textbooks such as Hurty and Rubinstein [21]. However, the subsequent treatment of these equation is open to invention. Singh [43] used a straightforward method of analysis which led to a set of four simultaneous equations in the derivation of the power spectral density function of response to stationary random input. Here, these equations are expressed in terms of the Duhamel integral, which is helpful in deriving simpler and more direct expressions for the PSD and its moments.

Using the fact that b and s occur in conjugate pairs, and letting $b_i \neq \vec{b}_{+n}$ and $s_i = \vec{s}_{+n}$, where the superposed bar denotes the complex conjugate, Eq.19 can be written in the expanded form

$$v(t) = \sum_{i=1}^{n} \left\{ b_{i} \int_{0}^{t} \ddot{x}_{v}(\tau) \exp[(-\xi_{i}\omega_{i} + i\omega_{D_{i}})(t-\tau)] d\tau + \bar{b} \int_{0}^{t} \ddot{x}_{v}(\tau) \exp[(-\xi_{i}\omega_{i} - i\omega_{D_{i}})(t-\tau)] d\tau \right\}$$

$$= 2\sum_{i=1}^{n} \left\{ -imb_{i} \int_{0}^{t} \ddot{x}_{v}(\tau) \exp[-\xi_{i}\omega_{i}(t-\tau)] \sin\omega_{D_{i}}(t-\tau) d\tau + \operatorname{Re}b_{i} \int_{0}^{t} \ddot{x}_{v}(\tau) \exp[-\xi_{i}\omega_{i}(t-\tau)] \cos\omega_{D_{i}}(t-\tau) d\tau \right\}$$
(23)

For the special case of classical damping, it was noted that b_i is pure imaginary, therefore only

the first integral term appears in Eq.23, which is in agreement with Fq.8. For the general case of non-classical damping, a second integral expression with a cosine term appears in Eq.23 which is due to the phase differences arising from the complex mode shapes. The first integral can be directly expressed in terms of the Duhamel integral. The appearance of the cosine term in the second integral suggests that this expression can be obtained in terms of the derivative of the Duhamel integral

$$\dot{h}(t) = \int_{0}^{t} \ddot{x}_{v}(\tau) \exp[-\xi \ \omega \ (t-\tau)] \cos \omega_{D_{v}}(t-\tau) d\tau \qquad \xi \ \omega \ h(t)$$
(24)

Substituting into Eq.23 yields

$$v(t) = 2\sum_{i=1}^{n} \left\{ -\ln b \left[\omega_{t}, h(t) \right] + \operatorname{Re} b \left[\dot{h}_{i}(t) + \xi_{i} \omega_{i} h(t) \right] \right\}$$

=
$$\sum_{i=1}^{n} \left[a_{i} h(t) + c_{i} \dot{h}(t) \right]$$
(25)

where,

$$a = -2\operatorname{Re}(b,\overline{s}), \quad c_i = 2\operatorname{Re}(b). \tag{26}$$

This formulation is considerably simpler than the earlier expression, Eq.23. It can readily be evaluated in the time domain for deterministic input using several well-known numerical methods for evaluating h(t) and $\hat{h}_i(t)$ [5]. It is also in a form suitable for random vibration analysis, as shown in the next section.

Before continuing, several important features of Eq.25 should be pointed out. For classical damping, we recall that $b_i = i\psi_i/2\omega_{D_i}$ so that $c_i=0$ and $a_i=\psi_i$, and Eq.25 reduces to Eq.8, as expected. Also, a relationship between c_i can be found by observing that for an impulsive load $\bar{x}_c(t)=\delta(t)$, where $\delta(t)$ is the Dirac delta function, all displacements must be zero immediately after start at time $t=0^+$. It follows that $v(0^+)=0$ and $h_i(0^+)=0$. Also, $\bar{h}_i(0^+)=1$, as can be seen from Eq.24. Substituting into Eq.25

$$v(0^{+}) - \sum_{i=1}^{n} \left\{ a_{i} h_{i}(0^{+}) + c_{i} \dot{h}_{i}(0^{+}) \right\} = \sum_{i=1}^{n} c_{i} = 0$$
(27)

This fact will be important in subsequent developments in this paper.

2.3 Response to Base Input in Frequency Domain

In this section the autocorrelation function and PSD are derived. The simplicity of the expression for the response, Eq.25, allows for the derivation of closed form results with physical interpretations. This allows for a greater insight in the behavior of non-classically damped systems and is particularly suited for the analysis of PS systems in the next chapters. If the ground motion $\vec{x}_{i}(t)$ is a stationary random process, the autocorrelation function of the stationary response v(t) can be obtained as follows:

$$R_{i,i}(\tau) = E[v(t)v(t+\tau)]$$

= $\sum_{i}^{n} \sum_{j=1}^{n} \left\{ a_{i}a_{j}E[h(t)h_{i}(t+\tau)] + a_{i}c_{j}E[h(t)\dot{h}_{i}(t+\tau)] + a_{i}c_{j}E[\dot{h}(t)\dot{h}_{i}(t+\tau)] + a_{i}c_{j}E[\dot{h}(t)\dot{h}_{i}(t+\tau)] + a_{i}c_{j}E[\dot{h}(t)\dot{h}_{i}(t+\tau)] \right\}$
(28)

It is well known [28] that if $R_{i}(\tau) \equiv E[h(t)h_{i}(t+\tau)]$ represents the cross-correlation function of h(t) and $h_{i}(t)$, then

$$E[h(t)\dot{h}_{t}(t+\tau)] = R'_{tt}(\tau)$$
(29a)

$$E[\hat{h}(t)h_{i}(t+\tau)] = -R^{2}_{ij}(\tau)$$
(29b)

$$E[h_{i}(t)h_{i}(t+\tau)] = -R''_{i}(\tau)$$
(29c)

where a prime denotes differentiation with respect to τ . Thus the autocorrelation function simplifies to

$$R_{i,i}(\tau) = \sum_{j=1}^{n} \sum_{i=1}^{n} [a | a_i R_{i,i}(\tau) + a | c_i R_{i,i}^{*}(\tau) - a_i c_i R_{i,j}^{*}(\tau) - c_i c_i R_{i,i}^{**}(\tau)]$$
(30)

In order to find the power spectral density function of the response, the Fourier transforms of the above expressions must be derived. The transform $G_{ij}(\omega)$ of the cross-correlation functic $-\mathbf{R}_{ij}(\tau)$ is the cross spectral density of $h_i(t)$ and $h_i(t)$ and is given by

$$G_{ij}(\omega) = G_{0}(\omega)H_{i}(\omega)\overline{H}_{i}(\omega)$$
(31)

where $H_{\varepsilon}(\omega)$ is the frequency response function for an oscillator of frequency ω_{ε} and damping ξ_{ε} .

$$H_{i}(\omega) = \frac{1}{\omega_{i}^{2} - \omega^{2} + 2i\xi_{i}\omega_{i}\omega}$$
(32)

and $G_0(\omega)$ is the power spectral density function of the input process $\tilde{x}_c(t)$. From the basic

properties of Fourier transforms, $(i\omega G_{i}(\omega), R_{i,i}^{*}(\tau))$ and $(-\omega^{2}G_{i,i}(\omega), R_{i,i}^{*}(\tau))$ are Fourier transform pairs. Thus, taking the Fourier transform of Eq.30, the power spectral density function for the response $v(\tau)$ is obtained:

$$G_{ij}(\boldsymbol{\omega}) = \sum_{1}^{n} \sum_{i=1}^{n} [C_{i} + i\boldsymbol{\omega}D_{j} + \boldsymbol{\omega}^{2}E_{j}]G_{j}(\boldsymbol{\omega})$$
(33)

where,

$$C_{i} = a_{i}a_{i}, D_{i} = (a_{i}c_{i} - a_{i}c_{i}), \text{ and } E_{i} = c_{i}c_{i}.$$
 (34)

This is a generalization of the power spectral density function of a system with classical damping. As was demonstrated earlier, b_i is a pure imaginary number for classically damped systems; therefore, $c = c_i = 0$, and $D_i = E_i = 0$. The remaining coefficient C_i is evaluated using Eq.20 and Eq.26. It turns out for this case that Eq.33 reduces to the well known power spectral density function for classically damped systems [10, 15]

$$G_{ij}(\boldsymbol{\omega}) = \sum_{i=1}^{N} \sum_{j=1}^{N} \boldsymbol{\psi}_{ij} G_{ij}(\boldsymbol{\omega})$$
(35)

The difference between the above and the more general expression in Eq.33 for non-classically damped systems is the inclusion of extra terms which arise due to the phase effects in the modal responses.

2.4 Spectral Moments of Response

As stated in the Introduction, most statistical measures of the response that are of engineering interest are obtained in terms of the first few spectral moments λ_{min} , i.e. for m=0,1,2, and 4. Substituting Eq.33 into the integral in Eq.1, the following expression is obtained for the *m*-th moment

$$\lambda_{m} = \sum_{i=1}^{n} \sum_{j=1}^{n} \{ C_{ij} \operatorname{Re} \lambda_{m,ij} = D_{ij} \operatorname{Im} \lambda_{m(i,1,j)} + E_{ij} \operatorname{Re} \lambda_{m(i,2,j)} \}$$
(36)

where,

$$\lambda_{m,n} = \int_{0}^{\infty} \omega^{m} G_{n}(\omega) d\omega = \int_{0}^{\infty} \omega^{m} G_{0}(\omega) H_{i}(\omega) \overline{H_{i}(\omega)} d\omega$$
(37)

are cross spectral moments associated with $h_i(t)$ and $h_i(t)$.

For the case of classical damping $D_1 = E_2 = 0$ and Eq.36 reduces to

$$\lambda_{\mu} = \sum_{i=1}^{n} \sum_{j=1}^{n} \psi_{ij} Re \lambda_{\mu} q_{ij}$$
(38)

which is the same formulation derived by Der Kiureghian [15]. The phase shift in the eigenvectors caused by non-classical damping results in the appearance of higher order moment terms in Eq.36.

The next topic to consider is the important question of convergence of the integral in Eq.37. In previous work, this topic did not receive adequate mathematical clarification, and divergent expressions were left unresolved (e.g., Vanmarke, Ref. 44, pp.437, Eq.44). If the power spectral density function is band limited, i.e. if there is a cutoff frequency ω_0 such that $G_1(\omega)=0$ for all frequencies ω greater than ω_0 , then it is clear that the integral for the cross spectral moments in Eq.37 converges for all positive values of m. Thus the moments λ_0 , which is a sum of terms involving λ_{m-1} , $\lambda_{m+1,m}$, and $\lambda_{m+2,m}$ can be found for any order. However, many theoretical power spectral density functions are not band limited. For large ω , the integrand terms $H_1(\omega)$ and $\overline{H_1(\omega)}$ have the following behavior:

$$\operatorname{Re}\omega^{n} H(\omega) \overline{H_{i}(\omega)} \rightarrow \omega^{n-4}$$
(39a)

$$\operatorname{Im}\omega^{n}H_{i}(\omega)\overline{H_{i}(\omega)} \to \omega^{n-1}$$
(39b)

Since the entire integrand must be of order smaller than ω^{-1} for large ω to insure convergence, it follows that the integral for λ_{m-n} will converge only if $G_0(\omega)$ is of order smaller than $\omega^{-(m-1)}$ for $\omega \rightarrow \infty$. For instance, for white-noise input, where $G_0(\omega) \equiv G_0$ is a constant, λ_{m-n} will converge only for m < 3. It appears that for this input, the moments λ_m from Eq.36 would only exist for m=0. However, for classically damped systems, λ_m is a sum of terms only involving λ_{m-n} , so the moment exists for m=0,1, and 2. From intuitive considerations, the same moments which exist for classically damped systems should also exist for non-classically damped systems. In other words, if $m=m_0$ is the highest existing order for the cross spectral moments λ_{m-n} , then m_0 should also be the highest order for the moment λ_m for both classically damped and non-classically damped structures. This hypothesis will be proven, presently.

Eq.36 can be rewritten in the following form:

- 24 -
$$\lambda_{\mu} = \sum_{i=1}^{n} \sum_{i=1}^{n} \left\{ C_{i} \operatorname{Re} \lambda_{\mu-i} = D_{ii} \operatorname{Im} \lambda_{i-i} \right\} + \int_{0}^{1} \operatorname{Re} \sum_{i=1}^{n} \sum_{j=n}^{n} E_{j} \omega^{m+2} G_{ij}(\omega) d\omega$$
(40)

If λ_{m-1} exists, then $Im\lambda_{m+1}$, also exists from the relations Eqs.39a,b. Thus, only the last term in Eq.40 must be tested for convergence.

The problem is that the summation and the infinite integral cannot be interchanged; the form of the expression above converges, yet the alternate formulation with the summation outside the integral diverges. This same problem arises in the previously referenced equation in Vanmarke's study. The problem is solved by rewriting the integrand term $G_{i}(\omega)$ as a sum of two components, one which vanishes under the summation and the other, defined as a modified power spectral density $G'_{i,i}(\omega)$, which converges when the summation and integral is interchanged.

First, $G'_{-}(\omega)$ is found by subtracting the dominant term from $G_0(\omega)$. Since $H_{-}(\omega)\overline{H_{-}(\omega)}$ tends to ω^{-4} for large ω , the dominant term is simply $\omega^{-4}G_0(\omega)$. From partial fraction expansions, it can be shown that the difference,

$$G_{\mu}(\omega) = \omega^{-4}G_{\mu}(\omega) \rightarrow \omega^{-6}G_{\mu}(\omega)$$
 for large ω (41)

The above could be used for the modified power spectral density $G'_{\alpha}(\omega)$ except that the second term diverges at ω near 0. To correct for this, the following definition is used:

$$G'_{ij}(\omega) = \begin{cases} G_{ij}(\omega) - \omega^{-4} G_{ij}(\omega) & \omega \geqslant \omega_{0} \\ G_{ij}(\omega) & 0 \leqslant \omega < \omega_{0} \end{cases}$$
(42)

where ω_0 is any fixed, arbitrary positive frequency. The corresponding n -dified cross-spectral moment is

$$\boldsymbol{\lambda}_{n=n}^{*} = \int_{0}^{\infty} \boldsymbol{\omega}^{n} \boldsymbol{G}_{n}^{*}(\boldsymbol{\omega}) d\boldsymbol{\omega}$$
(43)

which converges because of the relation in Eq.41.

What remains to be shown is that the above expression for $\lambda_{m+2,n}^{*}$ can be substituted into Eq.36 without changing the value for λ_{m} . The last term in Eq.40 can be rewritten

$$\int_{0}^{\infty} \operatorname{Re} \sum_{j=1}^{n} \sum_{\ell=1}^{n} E_{\ell \ell} \omega^{n+2} G_{\ell j}(\omega) d\omega$$

$$= \int_{0}^{m_{0}} \operatorname{Re} \sum_{\ell=1}^{n} \sum_{j=1}^{n} E_{\ell j} \omega^{n+2} G_{\ell j}^{\prime}(\omega) d\omega + \int_{m_{0}}^{\infty} \operatorname{Re} \sum_{\ell=1}^{n} \sum_{j=1}^{n} E_{\ell j} \omega^{n+2} [G_{\ell j}^{\prime}(\omega) + \omega^{-4} G_{0}(\omega)] d\omega$$

$$= \sum_{n=1}^{n} \sum_{i=1}^{n} E_{i} \operatorname{Re} \lambda_{in+2,i}^{i} + \int_{m_{0}}^{1} \operatorname{Re} \omega^{n-2} G_{0}(\omega) \sum_{i=1}^{n} \sum_{j=1}^{n} E_{ij} d\omega$$
(44)

Since $\sum_{i=1}^{r} \sum_{j=1}^{r} E_{ij} = \sum_{j=1}^{r} c_{j} \sum_{j=1}^{r} c_{j} = 0$ (see Eq.27), the last integral in the above is zero, and it has been shown that the remaining expression is well-defined. Thus, the original hypothesis is proven, and by combining the above with Eq.40, the final expression for the moment λ_{ij} is obtained:

$$\lambda_{\mu} = \sum_{i=1}^{2} \left\{ C_i \mathbf{R} \mathbf{e} \lambda_{\mu} - D_i \mathbf{I} \mathbf{m} \lambda_{\mu+1,\mu} + E_i \mathbf{R} \mathbf{e} \lambda_{\mu+2,\mu}^{\dagger} \right\}$$
(45)

which exists for the same values for m as the corresponding expression for classically damped systems, Eq.38. Note the similarity in form of the expressions in Eqs.45 and 36. In the following, the preceding equation will be used in general with the understanding that the primed moments need be used only when the unprimed moments do not exist.

The above expressions for the spectral moments of response are general. Also, the form of these expressions allow for further analysis and insight into the properties and dynamic response of non-classically damped systems. The evaluation of λ_{in} has been separated to two independent problems: the calculation of the coefficients C_{in} , D_{in} , and E_{in} , and the determination of the generic cross spectral moments $\lambda_{inn,in}$. The first problem is one of dynamics and is concerned entirely with the properties of the structural system. Using tools of dynamic analysis, this problem can be routinely solved; for PS systems, procedures are developed for finding analytical and closed form solutions in the remaining chapters of this study. The second problem is one of random vibrations and is concerned with the properties of the input excitation and the frequencies and damping ratios of the system. For any stationary input process with power spectral density function $G_n(\omega)$, the generic integrals in Eqs.37 and 43 can be calculated either by the method of residues, partial fractions, or by numerical integration. Below, closed form solutions are found for these cross-moments for the important case of white noise input. These results form the basis for a response spectrum method for seismic response of non-classically damped systems which is presented at the end of this chapter. The expression in Eq.25 can also be used for the study of nonstationary input with results that are more general and complicated than those for stationary input. Clearly, the generic cross-correlation function $R_{ii}(t_1, t_2) = E[h_i(t_1)h_i(t_2)]$, and its partial derivatives need to be evaluated. The results of such an approach would be in essence equivalent to the method of Debchandhury et. al. [14]. However, this formulation may yield simpler and more tractable results. This study will be reported in the future.

2.5 Response to White-Noise Input

Attention is drawn to the evaluation of structural response to white-noise input, where $G_0(\omega) \equiv G_0$ is a constant. This special case is important from an analytical viewpoint and is studied in most random vibrations textbooks [28]. The results of the analysis for white noise input are generally simple to interpret and are helpful in the study of more general forms of input. In addition, in this study, the results for the response to white noise input form the basis for the response spectrum method to be presented in Section 2.7.

Using the method of partial fractions and method of residues, closed form solutions for Eqs.37 and 43 have been obtained for all cross-spectral moments required for determining the first three spectral moments λ_{10} , λ_1 , and λ_2 of the response process. It is noted that moments higher than λ_2 for the response to white-noise input do not exist.

The cross spectral moments for response to white-noise input G_0 are

$$\operatorname{Re}_{\mathcal{D},\alpha} = \left\{ \omega_{i}\xi_{i} + \omega_{j}\xi_{j} \right\} \frac{2\pi G_{0}}{K_{j}}$$
(46a)

$$\operatorname{Re}_{\lambda_{1,ij}} = \left\{ [(\omega_{i}^{2} + \omega_{j}^{2})\xi_{i} + 2\omega_{i}\omega_{j}\xi_{j}] \frac{1}{\beta_{i}} \tan^{-1} \frac{\beta_{i}}{\xi_{i}} + [(\omega_{i}^{2} + \omega_{j}^{2})\xi_{j} + 2\omega_{i}\omega_{j}\xi_{j}] \frac{1}{\beta_{i}} \tan^{-1} \frac{\beta_{j}}{\xi_{i}} - (\omega_{j}^{2} - \omega_{j}^{2})\log \frac{\omega_{i}}{\xi_{i}} \right\} \frac{G_{0}}{\xi_{i}}$$

$$(46b)$$

$$\operatorname{Im} \lambda_{1,i_{\ell}} = \left\{ -\omega_{i}^{2} + \omega_{i}^{2} \right\} \frac{\pi G_{0}}{K_{i_{\ell}}}$$
(46c)

$$\operatorname{Re}_{\Sigma_{\alpha}} = \left\{ \omega_{\alpha} \xi_{\beta} + \omega_{\beta} \xi_{\alpha} \right\} \omega_{\alpha} \omega_{\beta} \frac{2\pi G_{\alpha}}{K_{\alpha}}$$
(46d)

$$\mathrm{Im}\lambda_{2,ij} = \left\{ \left[-\omega_i^2 + \omega_j^2 + 2\omega_j \xi_j (\omega_i \xi_j + \omega_j \xi_j) \right] \frac{\omega_i}{\beta_i} \tan^{-1} \frac{\beta_i}{\xi_j} \right\}$$

$$+ \left[-\omega_{i}^{2} + \omega_{j}^{2} - 2\omega_{i}\xi_{i}\left(\omega_{i}\xi_{j} + \omega_{i}\xi_{i}\right)\right]\frac{\omega_{i}}{\beta_{i}}\tan^{-\frac{1}{2}}\frac{\beta_{i}}{\xi_{i}}$$

$$- 2\omega_{i}\omega_{i}\left(\omega_{i}\xi_{j} + \omega_{i}\xi_{i}\right)\log\frac{\omega_{i}}{\omega_{i}}\right]\frac{G_{ii}}{K_{ij}}$$

$$(46e)$$

$$\operatorname{Re}\lambda^{2}\varepsilon_{ij} = \left\{\left[3\omega_{i}^{2}\xi_{j} + 2\omega_{i}\omega_{j}\xi_{j} - \omega_{i}^{2}\xi_{j} - 4\omega_{i}\xi_{i}^{2}\left(\omega_{i}\xi_{j} + \omega_{j}\xi_{i}\right)\right]\frac{\omega_{j}^{2}}{\beta_{j}}\tan^{-\frac{1}{2}}\frac{\beta_{i}}{\xi_{i}}$$

$$+ \left[3\omega_{j}^{2}\xi_{i} + 2\omega_{i}\omega_{j}\xi_{j} - \omega_{j}^{2}\xi_{i} - 4\omega_{i}\xi_{i}^{2}\left(\omega_{i}\xi_{j} + \omega_{j}\xi_{i}\right)\right]\frac{\omega_{j}^{2}}{\beta_{j}}\tan^{-\frac{1}{2}}\frac{\beta_{i}}{\xi_{i}}$$

$$= \left\{13\omega_{i}^{2}\xi_{i} + 2\omega_{i}\omega_{j}\xi_{j} - \omega_{j}^{2}\xi_{i} - 4\omega_{i}\xi_{i}^{2}\left(\omega_{i}\xi_{j} + \omega_{j}\xi_{i}\right)\right\}\frac{\omega_{j}^{2}}{\beta_{i}}\tan^{-\frac{1}{2}}\frac{\beta_{i}}{\xi_{i}}$$

$$+\frac{1}{2}\left[-\omega_{i}^{4}+\omega_{i}^{4}+4(\omega_{i}^{2}\xi_{i}^{2}-\omega_{i}^{2}\xi_{i}^{2})\right]\log\frac{\omega_{i}}{\omega_{i}}\left\{\frac{G_{0}}{K_{ii}}\right\}$$
(46f)

$$\operatorname{Im}\lambda_{3-i} = \left\{\omega_{i}^{4} - \omega_{i}^{4} - 4\omega_{i}\omega_{j}(\omega_{i}\xi_{i} - \omega_{j}\xi_{i})(\omega_{i}\xi_{j} + \omega_{j}\xi_{i})\right\} \frac{\pi G_{0}}{2K_{j}}$$
(46g)

$$\operatorname{Re} \lambda_{4,ij}^{i} = \left\{ \omega_{i} \xi_{i} \left[\omega_{j}^{4} + 2\omega_{i}^{2} \omega_{j}^{i} - \omega_{i}^{4} \right] + \omega_{i} \xi_{j} \left[-\omega_{j}^{4} + 2\omega_{i}^{2} \omega_{j}^{2} + \omega_{i}^{4} \right] - 4\omega_{i} \omega_{i} \left(\omega_{i} \xi_{j} + \omega_{i} \xi_{i} \right) \left(\omega_{i}^{2} \xi_{j}^{2} + \omega_{j}^{2} \xi_{j}^{2} \right) \right\} \frac{\pi G_{0}}{K_{ij}}$$

$$(46h)$$

where.

$$K_{ii} = (\omega_i^2 - \omega_i^2)^2 + 4\omega_i \omega_i (\omega_i \xi_i + \omega_j \xi_i) (\omega_i \xi_j + \omega_j \xi_i)$$

$$\beta_i = \sqrt{1 - \xi_i^2}$$
(47a)
(47a)

For the calculation of $\operatorname{Re}\lambda_{A,g}^{*}$ and $\operatorname{Re}\lambda_{A,g}^{*}$, the transition frequency ω_{0} used in Eq.42 was set to 1 and 0, respectively. Note that results for these cross-moment terms would be different for other selections of ω_{0n} but the final result for λ_{n0} would remain unchanged. When the indices iand j are equal, the imaginary parts of the moments become 0 and the real parts reduce to:

$$\lambda_{0,n} = \frac{\pi G_0}{4\xi_r \omega_r^3} \tag{48a}$$

$$\lambda_{1,\nu} = \frac{\pi G_0}{4\xi_1 \omega_1^2} \frac{2}{\pi \beta_1} \tan^{-1} \frac{\beta_1}{\xi_1}$$
(48b)

$$\lambda_{2,a} = \frac{\pi G_0}{4\xi_i \omega_i} \tag{48c}$$

$$\lambda'_{3,\mu} = \frac{\pi G_0}{4\xi_1} \frac{2(1-2\xi_1^2)}{\pi \beta_1} \tan^{-1} \frac{\beta_1}{\xi_1}$$
(48d)

$$\lambda'_{4,n} = \frac{\pi G_0 \omega_1}{4\xi_1} \left(1 - 4\xi_1^2\right) \tag{48c}$$

Good, simple approximations for the above expressions for the cross moments can be made if the damping terms are reasonably small. The error will be of the order ξ_{i}^{2} and ξ_{i}^{2} , e.g. if the damping is about 20% the error is only 4%. The approximations are

$$\operatorname{Re}_{W_{ij}} = \frac{2\pi G_0}{K_{ij}} \left[\omega_i \xi_j + \omega_j \xi_j \right]$$
(49a)

$$\operatorname{Re}_{\lambda_{1,j}} = \frac{2\pi G_0}{K_0} \left[\frac{1}{4} (\omega_i + \omega_j)^2 (\xi_i + \xi_j) - \frac{1}{2\pi} (\omega_i^2 - \omega_j^2) \log \frac{\omega_i}{\omega_j} \right]$$
(49b)

$$Im\lambda_{1,ii} = \frac{2\pi G_0}{K_{ij}} \left[\frac{-\omega_i^2 + \omega_j^2}{2} \right]$$
(49c)

$$\operatorname{Re} \lambda_{2,ij} = \frac{2\pi G_0}{K_{ij}} \left[\omega_i \xi_j + \omega_j \xi_j \right] \omega_i \omega_j$$
(49d)

$$Im\lambda_{2,ij} = \frac{2\pi G_0}{K_{ij}} \left[\frac{1}{4} \left(-\omega_i^2 + \omega_j^2 \right) \left(\omega_i + \omega_j \right) - \frac{\omega_i \omega_j}{\pi} \left(\omega_i \xi_j + \omega_j \xi_j \right) \log \frac{\omega_i}{\omega_j} \right]$$

$$(49e)$$

$$\operatorname{Re}\lambda_{3,\mu}^{\prime} = \frac{2\pi G_{0}}{K_{\mu}} \left\{ \frac{1}{4} \left[-\omega_{1}^{4}\xi_{1} + 2\omega_{1}^{3}\omega_{1}\xi_{1} + 3\omega_{1}^{2}\omega_{1}^{2}(\xi_{1} + \xi_{1}) + 2\omega_{1}^{3}\omega_{1}\xi_{1} - \omega_{1}^{4}\xi_{1} \right] + \frac{1}{4\pi} \left[-\omega_{1}^{4} + \omega_{1}^{4} \right] \log \frac{\omega_{1}}{\omega_{1}} \right\}$$

$$(49f)$$

$$\operatorname{Im}\lambda_{3,j} = \frac{2\pi G_0}{K_j} \left[\frac{\omega_j^4 - \omega_j^4}{4} \right]$$
(49g)

$$\operatorname{Re}\lambda_{4,ij}^{\prime} = \frac{2\pi G_0}{K_{ij}} \left[\frac{\omega_i \xi_i}{2} (\omega_i^4 + 2\omega_i^2 \omega_j^2 - \omega_i^4) + \frac{\omega_i \xi_j}{2} (-\omega_i^4 + 2\omega_i^2 \omega_i^2 + \omega^4) \right]$$
(49h)

 $Re\lambda_{0,n}$, $Re\lambda_{1,n}$, and $Re\lambda_{2,n}$ were previously given in the earlier report by Der Kiureghian [15] for systems with classical damping, as they are the only cross moments needed to evaluate λ_0 , λ_1 , and λ_2 for such systems. The results for $Re\lambda_{0,n}$, $Re\lambda_{1,n}$, and $Re\lambda_{2,n}$ were also previously known [15, 44]. The remaining solutions shown above, which are needed for nonclassically damped systems, are new results.

For further insight into the nature of the above terms, the following coefficients are introduced:

$$\rho_{m,n} = \frac{\operatorname{Re}\lambda_{m,n}}{\sqrt{\lambda_{m,n}\lambda_{m,n}}}$$
(50a)

$$\rho'_{m,n} = \frac{\operatorname{Re}\lambda'_{m,n}}{\sqrt{\lambda'_{m,n}\lambda'_{m,n}}}$$
(50b)

$$\eta_{m,\mu} = \frac{\mathrm{Im}\lambda_{m,\mu}}{\sqrt{\lambda_{m,\mu}\lambda_{m,\mu}}}$$
(50c)

Note that $\rho_{0,ij}$, $\rho_{2,ij}$, and $\rho_{4,ii}$ represent correlation coefficients between the zeroeth, first, and second derivatives of $h_i(t)$ and $h_j(t)$, respectively. The coefficients, $\rho_{0,ij}$, $\rho'_{3,ij}$ and $\eta_{1,ij}$, are plotted for various values of damping and frequency ratios in Fig.2.2. Results for other values

of *m* are similar and, therefore, at + not shown. As expected, $p = \frac{1}{2}$ and p'_{m-1} behave like correlation coefficients, diminishing rapidly as the two frequencies depart, particularly at small damping. Thus, cross modal terms $\text{Re}\lambda_m$ and $\text{Re}\lambda'_{m+2,m}$ in Eq.45 are significant only for modes with closely spaced frequencies. The plots indicate that η_m behaves quite differently from p_{m-2} . For moderately spaced frequencies, it can be seen that while p_{m-2} is negligible, the corresponding values for η_{m-2} are significant. It appears that the cross modal terms $\text{Im}\lambda_m \cdots_n$ in Eq.45 may be important with moderately spaced frequencies. However, preliminary example studies have shown that the coefficients D_{α} in Eq.45 are generally very small.

2.6 Examples

The simplest system having non-classical damping is the 2-degree-of-freedom system in Fig.2.3. This system has been the subject of considerable study in the past. Crandall and Mark [11] and Curtis and Boykin [12] used frequency response function approaches to find the response of this system to white noise input. Their methods, as well as the modal decomposition method developed here are exact, consequently the corresponding results of the analysis of the 2-DOF system agree.

The examples in this Section are chosen to illustrate the effect of non-classical damping. It can be shown that the 2-DOF system is classically damped if and only if the ratio of spring stiffnesses is equal to the ratio of damping coefficients:

$$\frac{k_1}{k_2} = \frac{c_1}{c_2}$$
(61)

If the two ratios are not equal, then in the mathematically strict sense, the system is nonclassically damped. As stated in the introduction, a common approximate approach in the analysis of non-classically damped structures is to use the free vibration mode shapes and to ignore the off-diagonal terms of the transformed damping matrix to eliminate modal coupling. For certain values of the parameters of the 2-degree-of-freedom system, this approximation appears reasonable. However, an analytical and numerical study of the system revealed two distinct and well-defined sets of parameter values where this classically damped approximation becomes inaccurate and inappropriate

The first case is the well-known soil-structure system where the damping ratios of the soil (substructure 1) and structure (substructure 2) are significantly different. The second case is the equipment-structure system where the mass ratio $\epsilon = m_2/m_1$ is small, the frequency ratio is close to 1 (tuning), and the difference of the damping ratios satisfies the following inequality:

$$(\xi_1 - \xi_2)^2 > \epsilon \tag{52}$$

To illustrate the differences between the approximate classically damped approach and the exact non-classically damped approach, a parameter study was made of the 2-degree-of-freedom soil-structure and equipment-structure systems using the 2 methods of analysis. In the soil structure study, the following fixed parameters were chosen: average damping ratio $\xi_a = (\xi_1 + \xi_2)/2 = 0.20$, mass ratio $\epsilon = m_2/m_1 = 0.3$, and frequency ratio $\omega_2/\omega_1 = 1.0$. In the equipment-structure study, the following parameters were chosen: average damping ratio $\xi_a = 0.04$, mass ratio $\epsilon = 0.001$, and the frequency ratio $\omega_2/\omega_1 = 1.0$. In both systems, the independent variable was the difference between the damping ratios $\xi_1 = \xi_2$. The relative displacement between the two masses was chosen as the response quantity of interest. The first three moments of response to white-moise input were computed using Eq.38 for the classically damped approximation and Eq.45 for the exact results.

From the first three moments, the mean square, λ_{01} , the mean zero-crossing rate, $\nu = \sqrt{\lambda_0/\lambda_0}/\pi$, and the shape factor, $\delta = \sqrt{1 - \lambda_0^2/(\lambda_0 \lambda_0)}$, were found and compared. The two approaches yielded very similar values for ν , however, a more significant difference of values were found for δ and λ_{00} and these variables are plotted for the various 2-degree-of-freedom systems in Figs.2.4 and 2.5. To gain further insight to the three sets of terms in the expression for λ_{00} in Eq.45, the percentage of the contributions of the terms:

$$\sum_{i=1}^{n} \sum_{j=1}^{n} C_{ij} \operatorname{Re} \lambda_{0,ij} , \quad \sum_{i=1}^{n} \sum_{j=1}^{n} D_{ij} \operatorname{Im} \lambda_{1,ij} , \text{ and } \sum_{i=1}^{n} \sum_{j=1}^{n} E_{ij} \operatorname{Re} \lambda_{2,ij}$$
(53)

to the total sum, λ_0 , is plotted in Fig.2.6.

For both systems, the condition Eq.51 is satisfied when the difference of damping ratios is

zero, therefore the systems are classically damped. Thus, the classically damped approximation yields exact results, as can be seen in Figs.2.4 and 2.5 and the D_0 and E_0 terms are zero, as shown in Fig.2.6. As the difference of damping ratios increases, the character of the system changes from classical damping to non-classical damping. From Figs.2.4 and 2.5 if is apparent that the effect of non-classical damping is not adequately accounted for in the classical damping approximations. This effect is also visible in Fig.2.6, where the E_0 terms become prominent for systems with increasing non-classical damping character. This is performed in equipmentic-structure systems with properties satisfying the inequality, Eq.52. This phenomena is discussed more fully in the next chapter.

It should be noted that for all 2-degree-of-freedom systems, the D terms in Eq.45 are essentially zero. This phenomenon is a characteristic of the simplicity of the system itself. A more complex 3-degree-of-freedom system was studied and compared with the results of the 2-degree-of-freedom system. Due to the larger number of parameters, the relationships between the parameters and the degree of non-classical damping was less clear. However, from the numerical results it could be generalized that the key parameters were the same as was observed for the 2-degree-of-freedom system, i.e., the difference of damping ratios and the mass ratio. However, unlike the 2-degree-of-freedom system, the D_{c} terms in Eq.45 were no longer negligible, particularly for soil-structure-type systems.

2.7 Development of the Response Spectrum Method

In this final section, the above results are extended to input specified by its response spectrum. Since the response spectrum is an incomplete description of the input, the method to be developed is necessarily approximate. The proposed method will be most accurate when the input is wide band, has a long stationary duration (several times longer than the fundamental period), and the significant modes of vibration are within the dominant frequencies of vibration,

In order to develop the response spectrum method, Eq.45 is rewritten in the form

$$\lambda_{m} = \sum_{i=1}^{n} \sum_{j=1}^{n} \left[C_{ij} \rho_{m,ij} w_{m,ij} - D_{ij} \eta_{m,ij} w_{m+1,ij} + E_{ij} \rho_{m+2,ij} w_{m+2,ij} \right] \sqrt{\lambda_{0,n} \lambda_{0,jj}}$$
(54)

where $p_{m,m}, p'_{m,m}, \eta_{m,m}$ are defined in Eqs.502-c, and

$$\mathbf{w}_{m,m} = \frac{\sqrt{\mathbf{\lambda}_{m,m} \mathbf{\lambda}_{m,m}}}{\sqrt{\mathbf{\lambda}_{m,m} \mathbf{\lambda}_{m,m}}}$$
(55)

Now we introduce two sets of approximations. The first approximation is to derive expressions for the coefficients $p_{m,\alpha}$, $\eta_{m,\alpha}$, and $w_{m,\alpha}$ in terms of the system parameters and independent of the input excitation. Let $\omega_{\alpha} = (\omega_{\alpha} + \omega_{\alpha})/2$, $\xi_{\alpha} = (\xi_{\alpha} + \xi_{\beta})/2$, $\omega_{\alpha} = \omega_{\alpha} - \omega_{\beta}$, and $\xi_{\alpha} = \xi_{\beta} - \xi_{\beta}$. Employing the assumption that the input is a wide-band process, and following the procedure in Ref. 16, it can be shown using results in Eqs.46 for white-noise input, that the following are good approximations:

$$\rho_{0,i} \approx R_{i} \{4\xi_{a} + \xi_{d} \omega_{d} / \omega_{a}\}, \quad \rho_{1,i} \approx R_{i} \{4\xi_{a} - 2\omega_{d}^{2} / (\omega_{a}^{2} \pi)\}$$
(56a)

$$\boldsymbol{\rho}_{2,q} \approx \boldsymbol{R}_{q} \left[4\boldsymbol{\xi}_{u} - \boldsymbol{\xi}_{d} \boldsymbol{\omega}_{d} / \boldsymbol{\omega}_{u} \right], \quad \boldsymbol{\rho}_{3,q} \approx \boldsymbol{R}_{q} \left[4\boldsymbol{\xi}_{u} + \boldsymbol{\omega}_{1}^{2} / (\boldsymbol{\omega}_{u}^{2}\boldsymbol{\pi}) \right]$$
(56b)

$$\rho'_{4,\mu} = \rho_{0,\mu} \tag{56c}$$

where $R_{ii} = \omega_a^2 \sqrt{\xi_i \xi_j} / [\omega_a^2 + 4\omega_a^2 \xi_a^2]$, and

$$\eta_{m,m} = 2R_m \frac{\omega_m}{\omega_m}, \quad w_{m,m} = \begin{cases} (\omega_m \omega_m)^{m/2} & \text{for } m \text{ even} \\ (\omega_m \omega_m)^{m/2} \left[\left(1 - \frac{2\xi_m}{\pi} \right) \left(1 - \frac{2\xi_m}{\pi} \right) \right] & \text{for } m \text{ odd} \end{cases}$$
(57)

The second approximation applied to Eq.54 is to estimate the modal spectral moments $\lambda_{0,\tau}$ in terms of the input response spectrum. Let $\bar{S}_{\tau}(\omega,\xi)$ be the mean response spectrum representing the mean value of the maximum absolute response of an oscillator over the duration τ of its the input excitation, where ω and ξ are the oscillator frequency and damping values, respectively. Then, for $\omega - \omega_{\tau}$ and $\xi - \xi_{\tau}$, the mean response spectrum ordinate can be given in terms of the spectral moment $\lambda_{0,\tau}$ [16]

$$\bar{S}_{i\tau} = \bar{S}_{\tau}(\omega_i, \xi_i) - \rho_i \sqrt{\lambda_{0,n}}$$
(58)

where p_i is a peak factor associated with mode *i* which can be derived in terms of system parameters as described below. Substituting the above into Eq.54 yields, for m=0,1,2,

$$\lambda_{m} = \sum_{i=1}^{n} \sum_{j=1}^{n} \left[C_{ij} \rho_{m-ij} w_{m,ij} - D_{ij} \eta_{m,ij} w_{m+1,ij} + E_{ij} \rho_{m+2,ij} w_{m+2,ij} \right] \frac{1}{p_i p_j} \widetilde{S}_{ij} \widetilde{S}_{jj},$$
(59)

The above equation provides a modal superposition rule for the spectral moments of response directly in terms of the response spectrum ordinates. These moments can be used to derive various important expressions of the response. In particular, the root-mean-square, and the mean and standard deviation of the peak response over duration τ are given, respectively, by $\sqrt{\lambda_{00}}$, $p\sqrt{\lambda_{00}}$, and $q\sqrt{\lambda_{00}}$, where p and q are peak factors given by [15]

$$p = \sqrt{2\ln\nu_c \tau} + \frac{0.5772}{\sqrt{2\ln\nu_c \tau}}$$
 and $q = \frac{1.2}{\sqrt{2\ln\nu_c \tau}} - \frac{5.4}{13 + (2\ln\nu_c \tau)^{3/2}}$ (60)

.n which

$$\nu_{\nu\tau} \tau = \begin{cases} \max(2.1, 2\delta\nu\tau), & 0 < \delta \le 0.1 \\ (1.63\delta^{0.45} - 0.38)\nu\tau, & 0.1 < \delta \le 0.69 \\ \nu\tau, & 0.69 < \delta \le 1 \end{cases}$$
(61)

and $\nu = \sqrt{\lambda_2/\lambda_0}/\pi$ and $\delta = \sqrt{1 - \lambda_1^2/\lambda_0\lambda_2}$. Eqs. 60 and 61 are also used in deriving the modal peak factors $p_{1,2}$ in which case $\nu = \omega_1/\pi$ and $\delta = 2\sqrt{\xi_1/\pi}$ [16]. Furthermore, having the three spectral moments one may derive the probability distribution of the peak response as given by Ref.45.

In practical applications, the mean of the peak response is of most interest. If the ratios of peak factors ρ/p_c are approximated by unity [16], a simplified expression for the mean peak response \overline{R} , is

$$\bar{R}_{\tau} = \left[\sum_{i=1}^{\tau} \sum_{j=1}^{n} \left[C_{ij} \rho_{0,ij} - D_{ij} \eta_{1,ij} w_{1,ij} + E_{ij} \rho_{2,ij} w_{2,ij} \right] \bar{S}_{i\tau} \bar{S}_{j\tau} \right]^{t}$$
(62)

This expression is a generalization of the CQC method introduced in Refs.16 and 50 for classically damped systems.

As an example application of the response spectrum method, a simple system composed of a foundation, structure, and light equipment, described in Fig.2.7 and Table 1, is studied. The modes of each fixed base subsystem have different damping ratios, as shown in Table 2, and as a result, the combined assemblage is characterized by non-classical damping.

Twenty simulated motions and their mean response spectrum were used for input in the analysis. The mean and standard deviation of the peak response of the equipment displacement relative to the foundation were obtained by using the modal combination rule of the proposed response spectrum method and the results compared with solutions obtained by the modal decomposition rule using a classical damping assumption (i.e., off-diagonal terms in the modal damping matrix were ignored) and exact solutions obtained from the numerical integration of the ground time histories. The results of the comparison, shown in Table 3, show that the response spectrum method is in close agreement with the simulation results and that the effect of non-classical damping can not be ignored in the analysis.

2.8 Summary and Conclusions

The response of multi-degree-of-freedom non-classically damped linear systems to stationary input excitation is examined and a modal decomposition method is developed employing the complex eigenvalues and eigenvectors of the system. A general formula for the spectral moments is derived and compared with the results for the special case of classical damping. The evaluation of these expressions involves two relatively independent problems. One is a dynamics problem in finding a set of coefficients in terms of the free vibration modal properties of the system under study. The other is a random vibrations problem in finding the crossspectral moments in terms of the input process. Procedures for finding the coefficients for PS systems are derived in the next chapters. For other systems, standard eigenvalue methods can be applied. The cross-spectral moments can be computed by the method of residues, partial fractions, or by direct integration for general input processes.

Closed form solutions for the cross-moments have been derived for the important case of white-noise input. These results were subsequently used to develop a response spectrum method of analysis for non-classically damped systems. This method is based on a set of approximations that were successfully used in previous works for classically damped systems [16, 50] and includes peak factors in the formulation, which are known to be of importance in the study of PS systems [18].

Example studies for simple systems have shown that non-classical damping occurs primarily in two types of systems: soil-structure-type systems, where the difference of damping ratios is large; and equipment-structure-type systems, where the mass ratio is small and the damping ratios are unequal. It is found that as the difference of the damping ratios increases or the mass ratio decreases, the classically damped approximation to the response becomes less accurate.

In general, for structures with classical damping or slight non-classical damping characteristics, the free vibration mode shapes and the decoupled modal equations, Eq.4, provide a suitable approximation to the true structural behavior. For such systems, the moments of the power spectral density function of response, Eq.38, have a relatively simple form. However, for structures which have predominantly non-classical damping characteristics, such as the soilstructure and equipment-structure systems, the damping matrix produces significant coupling of the free vibration mode shapes and the classically dampe'd approximation is no longer justifiable. This is shown numerically by the examples and mathematically by the more complex expressions for the moments of response, Eq.45.

| Subsystem | Frequency (rad/s) | Damping Ratio |
|------------|----------------------|------------------|
| Equipment | 12.00 | 0.0100 |
| Structure | 4.58 | 0.0191 |
| | 12.00 | 0.0500 |
| Foundation | 11.70 | 0.3000 |

Table 2.1. Physical Properties of Example System

Table 2.2. Modal Properties of Fixed Base Subsystems

| DOF | Mass (kg) | Interstory Stiffness (kN/m) | Interstory Damping (kN/(m/s)) |
|-----|--------------|-----------------------------------|-------------------------------------|
| 1 | 0.1 | 14.4 | 0.024 |
| 2 | 200.0 | 11000. | 916 |
| 3 | 200.0 | 11000. | 91.6 |
| 4 | 600.0 | 82100. | 4210.0 |

Table 3. Analysis Results: Equipment Response

| Response, m | Exact | Арргох. | Classical |
|------------------------|-------|---------|-----------|
| \overline{R} , | 0.106 | 0.113 | 0.046 |
| $\sigma_{\hat{R}_{i}}$ | 0.020 | 0.020 | 0.011 |



Fig.2.1. One Cycle of an Example Complex-Valued Mode Shape



Fig.2.2. Coefficients $\rho_{0,ij}$, $\rho'_{3,ij}$, and $\eta_{1,ij}$ for Response to White-Noise Input



Fig.2.3. Twe Degree-of-Freedom Example System



Fig.2.4. Response Parameters 8 and λ_0 for Soil-Structure Study



Fig.2.5. Response Parameters 8 and λ_0 for Equipment-Structure Study



Fig.2.6. Contribution of Terms to λ_0 in Example Studies

- 43 -



Fig.2.7. Example System for Response Spectrum Study

CHAPTER 3

ANALYSIS OF THE BASIC TWO-DEGREE-OF-FREEDOM EQUIPMENT-STRUCTURE SYSTEM

3.1 Introduction

The study of PS systems is best introduced with the simplest and most fundamental 2-DOF assemblage shown in Fig.3.1. This system is commonly referred to as an equipmentstructure (ES) system with the structure and equipment corresponding to the primary and secondary subsystems, respectively. A thorough study of this system is important for several reasons: the analysis of more general and complex MDOF systems is based on the results of the study of the 2-DOF system; the 2-DOF system contains the essential properties that characterize more general PS systems; and the simplicity of this system enables one to identify these properties easily. Also, simple closed form formulae are derived for the 2-DOF system which illustrate clearly and concisely the important relationships between the parameters of the system. The concepts and techniques developed in detail in this chapter will prepare the reader for the study of more complex MDOF PS systems in the following chapters.

As stated in the Introduct numerous studies have been made on equipment-structure systems. In this chapter, two different approaches will be developed. The first approach is based on the frequency response function of the system. Using perturbation methods based on the light equipment assumption it is possible to reduce the expressions derived by Crandall and Mark for the mean-square response of general 2-DOF systems [11] to rather simple formulae. Closed form expressions for other useful response quantities can be derived as well. The second approach utilizes the modal decomposition method developed in the preceding chapter. The results are a generalization of the previous study by Der Kiureghian, et. al. [18, 19]; the primary difference is that the important effect of non-classical damping, which occurs even as slight differences of damping ratios, is fully accounted for. The two approaches developed in

this study yield different formulations for the same response quantities, however it is shown that the two formulations are equivalent.

3.2 Definitions

3.2.1 Parameters

The analysis of the 2-DOF equipment-structure system in Fig.3.1 will be in terms of the parameters of the individual fixed base SDOF structure and equipment oscillators. The physical properties of the two sub-systems are: masses, m_i ; damping ratios, ξ_i ; and natural frequencies, ω ; where i=1 refers to the structure and i=2 refers to the equipment. The displacements x are relative to the ground. Using these properties, the following non-dimensional parameters are defined:

$$\xi_{\perp} \simeq \frac{\xi_{\perp} + \xi_{\perp}}{2} = \text{average damping}$$
(1a)

$$\xi_{i\ell} = \xi_1 - \xi_2 = \text{difference of damping} \tag{1b}$$

$$\epsilon = \frac{m_2}{m_1} = \text{mass ratio} \tag{1c}$$

$$\beta = \frac{\omega_1 - \omega_2}{\omega_a} = \text{detuning parameter}$$
(1d)

where

$$\omega_{o} = \frac{\omega_{1} + \omega_{2}}{2} - \text{average frequency}$$
(1e)

3.2.2 Review of perturbation methods

A non-dimensional parameter β is defined to be small if its absolute value is much smaller than 1; this is written symbolically as

$$|\boldsymbol{\beta}| << 1 \tag{2}$$

For example, the damping ratios in structural systems are generally small parameters. Similarly, a parameter β_1 is defined to be of a smaller order of magnitude than another parameter β_2 if the ratio $\{\beta_1|/|\beta_2\}$ satisfies

$$\frac{|\boldsymbol{\beta}_1|}{|\boldsymbol{\beta}_2|} \ll 1 \tag{3}$$

By multiplying both sides of the above inequality by $\{\beta_{i}\}$, the order relationship can be rewritten symbolically as

$$|\boldsymbol{\beta}_1| << |\boldsymbol{\beta}_2| \tag{4}$$

Finally, two parameters β_1 and β_2 are defined to be of the same order of magnitude if neither

$$|\boldsymbol{\beta}_1| \ll |\boldsymbol{\beta}_2| \quad \text{nor} \quad |\boldsymbol{\beta}_2| \ll |\boldsymbol{\beta}_1| \tag{5a}$$

are true. This is written symbolically as

. .

$$\boldsymbol{\beta}_1 = \boldsymbol{O}(\boldsymbol{\beta}_2) \quad \text{or} \quad \boldsymbol{\beta}_2 = \boldsymbol{O}(\boldsymbol{\beta}_1) \tag{5b}$$

where $O(\beta_1)$ denotes a term of the order of magnitude of the parameter β_1 . Note that the above relationship does not imply equality. For example, if the damping ratios ξ_1 =5% and ξ_2 =2%, they are not close to equality, yet they are considered to be of the same order of magnitude.

In engineering applications, the above relationships are put in the context of relative errors. In this study, the parameter e denotes the order of magnitude of the relative error that is allowed in the approximations.

Using the above definitions, a mathematical description can be given to equipmentstructure systems. The system in Fig.3.1 is defined to be an equipment-structure system if the parameters ξ_{μ} and $\sqrt{\epsilon}$ are small. The system is tuned if β is also small, otherwise the system is detuned. A more precise definition of detuning will be developed later in this chapter.

As mentioned in the introduction, the key to the analysis of the equipment-structure systems is the use of perturbation methods. Only the most elementary techniques of perturbation theory are used in this study. For instance, using Taylor's series the quantity $\xi_{2}(1+\xi_{d})^{-1}$ can be expanded

$$\frac{\xi_a}{1+\xi_a} = \xi_a \left(1-\xi_a+\xi_a^2-\xi_a^3+\cdots\right)$$
(6a)

For small ξ_{a} , the above can be approximated to the following degrees of accuracy:

$$\frac{\xi_{a}}{1+\xi_{a}} \approx \xi_{a}(1-\xi_{a}+\xi_{a}^{2}) \quad \text{second order}$$
(6b)

$$\approx \xi_a (1 - \xi_a)$$
 first order (6c)

$$= \xi_{\mu}$$
 lowest order (6d)

The accuracy of the above three approximations can be observed from the graph in Fig.3.2. Clearly the second order approximation is the most accurate; however the algebraic expression is also the most complicated. Since this is a study for engineering applications, a high level of accuracy is not required and simple, manageable results are sought, therefore lowest order approximations are used. Occasionally firs, and second order expressions are derived for obtaining intermediate results.

Most expressions encountered in this study involve a combination of several small parameters, and it is important to note the order relationships between the various parameters. In this chapter, it is assumed that ϵ and ξ_a are of the same order of magnitude, i.e.

$$\xi_{\mu} = O(\epsilon_{\mu}) \tag{7a}$$

and if the system is tuned, then

$$\boldsymbol{\beta} = \boldsymbol{O}(\boldsymbol{\epsilon}) \tag{7b}$$

Thus, an expression such as

$$\frac{\beta}{1+\xi_o} + \epsilon = \epsilon + \beta (1-\xi_o + \xi_o^2 - \cdots)$$
(8a)

in the case of tuning can be approximated to

$$\frac{\beta}{1+\xi_u} + \epsilon \approx \epsilon' + \beta(1-\xi_u + \xi_u^2) \quad \text{second order}$$
(8b)

 $\approx \epsilon' + \beta(1-\xi_a)$ first order (8c)

$$\approx \epsilon + \beta$$
 lowest order (8d)

One final note: In a real problem, the order relationships in Eqs.7a, b may not always appear to be valid. For instance, the detuning parameter β may happen to be exactly 0, in which case the order relationships would seem to be

$$\beta << \epsilon^{\circ}$$
 and $\beta << \xi_a$ (9)

Consequently, the lowest order approximation for the expression in Eq.8a would be

$$\frac{\beta}{1+\xi_{\mu}} + \epsilon^{\prime} \approx \epsilon^{\prime \prime}$$
(10)

The difference between Eq.8d and the above expression lies in the generality of the approximation. Eq.8d is valid if either of the relationships Eq.9 or Eqs.7a,b are satisfied, however the above approximation is valid only if Eq.9 is satisfied. Therefore, in order to obtain the most general results, it is assumed that Eq.7a applies to detuned systems and Eqs.7a,b applies to tuned systems.

3.3 Frequency Response Function Approach for Tuned Systems

3.3.1 Introduction

a 1

The frequency response function approach is well-suited for deriving the response of the equipment for tuned systems. By using perturbation methods, second-order results are obtained. These results are the basis of the response of all tuned PS systems and are used in the remainder of this study. Detuned systems are evaluated more easily using the modal approach and are studied in the next section.

3.3.2 Frequencies of the System

Let M,C, and K be the mass, damping, and stiffness matrices of the system in Fig.3.1. Using the parameters defined previously,

$$\mathbf{M} = m_1 \begin{bmatrix} 1 & 0 \\ 0 & \epsilon \end{bmatrix} \quad \mathbf{C} = 2m_1 \begin{bmatrix} \boldsymbol{\xi}_1 \boldsymbol{\omega}_1 + \epsilon \boldsymbol{\xi}_2 \boldsymbol{\omega}_2 & -\epsilon \boldsymbol{\xi}_2 \boldsymbol{\omega}_2 \\ -\epsilon \boldsymbol{\xi}_2 \boldsymbol{\omega}_2 & \epsilon \boldsymbol{\xi}_2 \boldsymbol{\omega}_2 \end{bmatrix} \quad \mathbf{K} = m_1 \begin{bmatrix} \boldsymbol{\omega}_1^2 + \epsilon \boldsymbol{\omega}_2^2 & -\epsilon \boldsymbol{\omega}_2^2 \\ -\epsilon \boldsymbol{\omega}_2^2 & \epsilon \boldsymbol{\omega}_2^2 \end{bmatrix}$$
(12)

The equation of the response of the system to base input is

$$\mathbf{M}\ddot{\mathbf{x}} + \mathbf{C}\dot{\mathbf{x}} + \mathbf{K}\mathbf{x} = -\mathbf{M}\mathbf{r}\ddot{x}_{\mathbf{y}}(t) \tag{13}$$

where $\mathbf{x} = [x_1 x_2]^T$ is the vector of displacements relative to the base and $\mathbf{r} = [1\,1]^T$ is the influence vector coupling the input to the DOF of the system. Taking the Fourier transform of Eq.13 and rearranging terms

$$\mathbf{X}(\boldsymbol{\omega}) = -\mathbf{H}(\boldsymbol{\omega})\mathbf{M}_{\mathbf{I}}X_{\boldsymbol{\mu}}(\boldsymbol{\omega}) \tag{14}$$

where $X(\omega)$ and $X_g(\omega)$ are Fourier transforms of x(t) and $\dot{x}_g(t)$, respectively, and $H(\omega)$ is the complex frequency response matrix

$$\mathbf{H}(\boldsymbol{\omega}) = [-\boldsymbol{\omega}^2 \mathbf{M} + i\boldsymbol{\omega}\mathbf{C} + \mathbf{K}]^{-1}$$
(15)

Substituting Eq.12 into the above,

1.....

$$\mathbf{H}(\omega) = \begin{bmatrix} G_1(\omega) & f_1(\omega) \\ f_1(\omega) & g_1(\omega) \end{bmatrix}^{-1} = \frac{1}{d(\omega)} \begin{bmatrix} g_1(\omega) & -f_1(\omega) \\ -f_1(\omega) & (f_1(\omega)) \end{bmatrix}$$
(16)

where

$$(r_1(\omega) = m_1[-\omega^2 + 2i(\xi_1\omega_1 + \epsilon\xi_1\omega_2)\omega + (\omega_1 + \epsilon\omega_2)]$$
(17a)

$$g_1(\omega) = m_1 \{-\omega^2 + 2i\xi_0\omega_0\omega + \omega_0\}$$
(17b)

. .

••

$$J_{1}(\omega) = -m_{2} [2i\xi_{2}\omega_{3}\omega + \omega_{3}]$$
(17c)

$$d(\omega) = g_1(\omega)(f_1(\omega) - f_1(\omega))$$
(1/d)

Note that $g_{\ell}(\omega)$ is the reciprocal of the complex frequency response function of the secondary subsystem. The function $G_{1}(\omega)$ closely resembles the reciprocal of the complex frequency response function of the primary subsystem. The difference is in the terms with the parameter ϵ which arise from the small additional stiffness and damping contributed from the secondary subsystem. The function $f_1(\omega)$ represents the coupling between the two subsystems. Also, $d(\omega)$ is the characteristic polynomial of the system.

Using the frequency response function $H(\omega)$ defined above many properties of the system can be derived. The first set of properties that will be investigated are the frequencies of the system, which are found by solving the quartic equation

$$d(\omega) = 0 \tag{18}$$

It is possible to find second-order approximations to this equation by a straightforward yet tedious application of Ferrari's formula for solving quartic equations [3]. However, a more elegant approach based on perturbation methods can be used to solve Eq.18. The derivation is presented in detail to illustrate how perturbation methods can be effectively applied to the equipment-structure problem. Other researchers have found other forms for the expressions for the frequencies of the 2-DOF ES system [37, 36], how wer the perturbation method, as it is applied in this chapter, yields expressions which are of the same level of accuracy, yet algebraically simpler than the previous formulations. These simple forms of the frequencies will be used to derive several new expressions of the response of the system.

It was stated that the parameters ξ_a , β , and ϵ' are small for tuned systems. It is instruc-

tive to examine the properties of the system when the mass ratio ϵ is of a smaller order of magnitude than the other parameters, i.e.

$$\epsilon \ll \max(\xi_{\alpha},\beta) \tag{19}$$

In this case, the first term in Eq.17d dominates the expression for $d(\omega)$ and the second term can be neglected

$$d(\omega) \approx g_1(\omega) G_1(\omega) \tag{20}$$

It is well known [11] that this is equivalent to ignoring the interaction forces between the equipment and structure and this point will be discussed later in this chapter. The low-order approximation for the frequencies of this system can be obtained by solving Eq.18. Using the above approximation for $d(\omega)$, Eq.18 is equivalent to

$$G_1(\omega) = 0 \quad \text{or} \quad g_1(\omega) = 0 \tag{21}$$

The roots of $G_1(\omega)$ are approximately equal to the frequencies of the primary substructure and the frequency with positive real part can be written to first order as

$$\omega_1 \approx \omega_1 (1+\xi_1) \tag{22a}$$

The superposed asterisk is used hereafter to indicate that these properties are associated with the combined system. Similarly, the roots of $g_1(\omega)$ are the frequencies of the secondary substructure and the frequency with positive real part can be written to first order as

$$\omega_2 \simeq \omega_2 (1+i\xi_2) \tag{22b}$$

The frequencies in the left half plane, which are located symmetrically with respect to the imaginary axis, are not explicitly written for clarity of notation. This convention will be used throughout the study.

On physical grounds, the above result is not surprising: For very small values of the mass ratio, the secondary system would be very light relative to the primary system and the dynamic behavior of the two subsystems would be essentially independent of each other. Therefore, the frequencies of the combined ES system would be very close to the frequencies of each subsystem, which is the result obtained above.

If ϵ is of the same order of magnitude as ξ_{α} or β , the above analysis would no longer be

valid. Both terms of the characteristic polynomial $d(\omega)$ must be included in Eq.17d. As a result, the solutions for the frequencies of the system would be slightly shifted or perturbed from the above values. To preserve the inherent symmetry of the problem, this perturbation will be measured from the average of the frequencies in Eqs.22a,b. If μ is defined to be the perturbation variable, the frequency ω can be written in the following form

$$\omega = \omega_u (1 + i\xi_u + \mu) \tag{23}$$

By rearranging terms, the variable μ can be expressed in terms of ω

$$\mu = \frac{\omega}{\omega_a} - 1 - i\xi_a \tag{24}$$

Reformulating Eqs. 17a-c in terms of μ and retaining only the lowest order terms, the new expressions are

$$G_1(\omega) \approx \omega_d^2 m_1 (\beta + i\xi_d - 2\mu)$$
(25a)

$$g_1(\omega) \approx -\omega_a^2 m_2 (\beta + i\xi_d + 2\mu)$$
(25b)

$$f_1(\omega) \approx -\omega_a^2 m_2 \tag{25c}$$

Substitution of the above into Eq.17d yields

$$d(\omega) \approx \omega_{\sigma}^{4} m_{1} m_{2} [(-\beta - i\xi_{\sigma} - 2\mu) (\beta + i\xi_{\sigma} - 2\mu) - \epsilon]$$

= $\omega_{\sigma}^{4} m_{1} m_{2} [4\mu^{2} - (\beta + i\xi_{\sigma})^{2} - \epsilon]$ (26)

and its roots are readily found. The usefulness of the perturbation analysis becomes clear: The original quartic polynomial $d(\omega)$ has been reduced to a simple quadratic polynomial in terms of μ . Denoting the solutions for μ by μ_{μ}^{-} , the lowest order result is

$$\mu_{i} = \pm \frac{1}{2} \sqrt{\epsilon + (i\xi_{d} + \beta)^{2}}$$
(27)

It follows from Eq.23 that the frequencies ω_i are

$$\omega_{i} \approx \omega_{a} [1 + i\xi_{a} \pm \frac{1}{2} \sqrt{\epsilon + (i\xi_{d} + \beta)^{2}}]$$
(28)

The accuracy of the above solution is shown numerically in Table 1.

In Figs.3.3a-c, the exact values of ω_1^* which are found from solving Eq.18 are plotted along with the above approximation for various values of the parameters. Figure 3.3a represents a general illustration of the behavior of the frequencies. The natural frequencies and damping ratios of the two subsystems are chosen to be unequal, and the location of these frequencies on the complex plane are indicated by the solid squares. Then, for several values of the mass ratio ϵ , the corresponding pairs of frequencies of the complete system are plotted and hibeled by the letter A-D. Figures 3.3b and 3.3c are similar, except in the former the frequencies ω_1 and ω_2 are chosen to be equal, and in the latter the damping ratios ξ_3 and ξ_2 are equal.

It is instructive to explore the meaning of Fq.28 and the corresponding characteristics that can be observed in Figs.3.3a-c. The following are some of the more revealing facts

- 1. The frequencies of the combined system are located symmetrically with respect to the average frequency $\omega_0 (1 + i\xi_0)$ on the complex plane. This is apparent in all of the Figures 3 3a-c.
- 2. For very small values of ϵ , the value of μ is approximately

$$\mu' \approx \pm \frac{1}{2} \sqrt{(i\xi_d + \beta)^2} = \pm \frac{1}{2} (i\xi_d + \beta)$$
(29a)

which, when substituted in Eq.23, yields the same result obtained earlier in Eqs.22a,b. However, if ϵ is not negligible, it was explained earlier in hueristic terms that the frequencies become displaced or perturbed from the frequencies in Eqs.22a,b. This perturbation is accounted for mathematically by the presence of the parameter ϵ in the radical in Eq.28. This phenomena can be observed in Figs.3.3a-c, where for ϵ =.0005, the frequencies of the combined system nearly coincide with the subsystem frequencies, whereas they become displaced for larger values of the mass ratio.

- 3. If the frequencies ω_{\perp} and ω_{2} are equal, then $\beta=0$ which is usually referred to as the "perfect tuning" condition [37]. In this case, the frequencies of the combined system can be in one of two configurations.
 - a. If $\xi_i^2 < \epsilon$, then the frequencies would be

$$\omega_{c}^{2} \approx \omega_{a} \left[1 \pm \frac{1}{2} \sqrt{\epsilon - \xi_{d}^{2}} + i\xi_{a} \right]$$
(29b)

which, on the complex plane, would lie to the right and left of the average at a distance of $\sqrt{\epsilon - \xi_a^2}$. The damping ratios would both be equal to ξ_a and the natural frequencies would be unequal

b. If $\mathcal{E} > \epsilon$, then the frequencies would be

$$\omega' \approx \omega_{\alpha} \left[1 + i \left(\xi_{\alpha} \pm \frac{1}{2} \sqrt{-\epsilon + \xi_{\alpha}} \right) \right]$$
(29c)

which, on the complex plane, would lie above and below the average at a distance $\sqrt{-\epsilon + \epsilon}$. In this case, the frequencies would both be equal to ω_n and the damping ratios would be unequal

These characteristics are visible in Fig.3.3b; for points A and B, Eq.29b applies, and for points C and D, Eq.29c applies.

4. If the damping ratios ξ_{\pm} and ξ_{\pm} are equal, then the frequencies would be

$$\boldsymbol{\omega}^{\dagger} \approx \boldsymbol{\omega}_{0} \left[1 \pm \frac{1}{2} \sqrt{\epsilon} + \beta^{2} + i \boldsymbol{\xi}_{0} \right]$$
(29d)

which, on the complex plane would always lie parallel to the real axis at a distance $\sqrt{\pi + \beta}$ from the average frequency, as shown in Fig.3.3c.

3.3.3 Spectral Density of Equipment Response

Next consider the PSD of the response to stationary input. The primary response variable of interest is the displacement v(t) of the equipment relative to the structure and is given by

$$y(t) = x_1(t) - x_1(t) = \mathbf{q}^T \mathbf{x}(t)$$
(30)

where $q = [-1, 1]^T$. The Fourier transform $f(\omega)$ of the response y(r) is found by taking the transform of both sides of the above equation and using Eq.14

$$Y(\omega) = q^{T} X(\omega) = -q^{T} H(\omega) Mr X_{\nu}(\omega)$$
(31)

By definition, the frequency response function $H_i(\omega)$ for the response y(i) to the input $\vec{x}_i(i)$ satisfies the relation:

$$Y(\omega) = \prod_{i} (\omega) X_{i}(\omega)$$
(32)

It follows from the previous two equations that $H_{i}(\omega)$ is given by

$$H_{\lambda}(\omega) = -\mathbf{q}^{T} \mathbf{H}(\omega) \mathbf{M} \mathbf{r}$$
(33)

It is well known [10, 28] that the PSD function $G_{ii}(\omega)$ of the response can be found in terms of the frequency response function by the relation

$$G_{ij}(\omega) = T_{ij}(\omega)G_{ij}(\omega)$$
(34a)

where $T_{ij}(\omega)$ is the transfer function

$$T_{ij}(\omega) = |H_j(\omega)|^2$$
(34b)

and $G_{x_{1}}(\omega)$ is the PSD of the input $\tilde{x}_{0}(t)$. By expanding $T_{x_{1}}(\omega)$ through the use of Eqs.16 and 33 and keeping second order terms, the following is obtained

$$T_{ii}(\omega) \approx \frac{\omega_i^2 + 4\xi_i^2 \omega_j^2 \omega_i^2}{d(\omega) d(-\omega)} m_i^2 m_i^$$

A plot for $T_{i_1}(\omega)$ is shown in Fig 3.4 for $\omega_1=1.0$, $\omega_2=1.04$, $\epsilon=.00^3$, and two sets of values for the damping ratios. It is apparent that $T_{i_1}(\omega)$ is highly peaked for values of ω that lie in a small neighborhood of ω_{i_1} and that the peak is higher for smaller damping values. The explanation of this phenomena can be seen by examining the denominator of the transfer function. The polynomial $d(\omega)$ can be factored

$$d(\boldsymbol{\omega}) = [\boldsymbol{\omega}_{\alpha}(1+i\boldsymbol{\xi}_{\alpha}+\boldsymbol{\mu}_{\lambda})-\boldsymbol{\omega}][\boldsymbol{\omega}_{\alpha}(1+i\boldsymbol{\xi}_{\alpha}+\boldsymbol{\mu}_{\lambda})-\boldsymbol{\omega}] [\boldsymbol{\omega}_{\alpha}(-1+i\boldsymbol{\xi}_{\lambda}-\boldsymbol{\mu}_{\lambda})-\boldsymbol{\omega}][\boldsymbol{\omega}_{\alpha}(-1+i\boldsymbol{\xi}_{\alpha}-\boldsymbol{\mu}_{\lambda})-\boldsymbol{\omega}]\boldsymbol{m}_{1}\boldsymbol{m}_{2}$$
(36)

When ω lies near the average frequency ω_{ω} , then the order of $d(\omega)$ is

$$d(\omega) = O(\xi_{\mu}^{2})\omega_{\mu}^{2}m_{\mu}m_{\mu}$$
(37)

From Eq.36, it is clear that the same order relationship holds for $d(-\omega)$. Returning to the expression for the transfer function, it follows that

$$T_{ij}(\omega) = O(\xi_{ij}^{-4}) \quad \text{for } \omega \approx \omega_{ij}$$
(38)

This characteristic of $T_{ii}(\omega)$ will be useful in making approximations

3.3.4 Spectral Moments of Response

The spectral moments $\lambda_{\mu\nu}$ of the response of the secondary system can be obtained by integrating the PSD function $G_{\mu\nu}(\omega)$

$$\lambda_{m} = \int_{0}^{m} \omega^{m} G_{\mu\nu}(\omega) d\omega = \int_{0}^{1} \frac{\omega_{1}^{4} + 4\xi_{1}^{2} \omega_{1}^{2} \omega^{2}}{d(\omega) d(-\omega)} \omega^{m} G_{\mu\nu}(\omega) d\omega$$
(39)

For general forms of the input PSD $G_{cc}(\omega)$, the above integral can be evaluated numerically. However, for the important case of white-noise input closed form second-order approximations can be derived. For m=0 and 2 the following integration formula is used [11]:

$$\int_{0} \frac{p(\omega) d\omega}{q(\omega)q(-\omega)} = \frac{\pi_{1} a_{4}(-a_{3}b_{1}+a_{1}b_{2}) + b_{3}(a_{3}-a_{1}a_{2})}{a_{4}(a_{3}^{2}+a_{1}a_{4}-a_{1}a_{2}a_{3})}$$
(40a)

where the polynomials $p(\omega)$ and $q(\omega)$ are given by

$$p(\omega) = b_1 \omega^2 + b_2 \omega^2 + b_3$$
(40b)

$$q(\omega) = a_1 \omega^3 + a_2 \omega^2 + a_3 \omega + a_4$$
(40c)

and the roots of $q(\omega)$ are required to be in the upper half plane. Since $G_{i,i}(\omega) \equiv G_{i,i}$ is a constant for white-noise input and the roots of $d(\omega)$ all lie in the upper half plane, the formula above can be used directly to solve Eq.39 for λ_0 and λ_2 . The calculations are fairly extensive yet straightforward with the following results:

$$\lambda_{ii} = \frac{\pi G_{av}}{\omega_a^3 D} \left[2\xi_a + 3\beta \xi_a + 10\beta^2 \xi_a + \epsilon (4\xi_a + \xi_d) + 16\xi_a^2 \xi_1 \right]$$
(4)a)

$$\lambda_{2} = \frac{\pi G_{ee}}{\omega_{\mu} D} \left[2\xi_{\mu} + \beta \xi_{\mu} + \epsilon \xi_{2} + 8\xi_{\mu} \xi_{1}^{2} \right]$$
(41b)

where the denominator D is

$$D = 16 \left[\xi_1 \xi_2 (4\xi_0^2 + \beta^2) + \epsilon \xi_0^2 \right]$$
(41c)

Note that second order approximations are derived for the numerators in the above expressions. These are necessary to derive further results in this section.

For the first moment λ_{\pm} , the integrand in Eq.39 is an odd-powered polynomial, therefore Eqs.40a-c cannot be used. The obvious alternative is the method of residues; using the frequencies derived in the preceding section with some mathematical manipulation, the secondorder approximation for λ_{\pm} can be found [23]

$$\lambda_{1} = \frac{\pi G_{00}}{\omega_{d}^{2} D} \left[2\xi_{a} + 2\beta \xi_{d} + \frac{3}{4} \epsilon \left(3\xi_{a} + \xi_{d} \right) + 8\xi_{1} \xi_{a} \left(5\xi_{1} + 3\xi_{2} \right) \right]$$
(41d)

It is noted that the lowest order expressions for the three moments are very similar:

$$\lambda_m \approx \frac{2\pi\xi_a G_{ac}\omega_a^{m-3}}{D} \quad \text{for } m = 0, 1, 2 \tag{41e}$$

Graphs comparing the expressions for λ_{μ} in Eqs.41a-d with exact results obtained by numerical integration are presented in Figs.3.5a-c.

As stated in the Introduction, the first three spectral moments can be used to derive several important response quantities. The first of these quantities to be examined are the mean squares of the displacement response σ_1^2 and the velocity response σ_1^2 which are given by λ_0 and λ_2 , respectively

$$\sigma_{\mu}^{2} \approx \frac{\pi \xi_{\mu} G_{\nu\nu}}{8\omega_{\mu}^{2} [\xi_{\mu} \xi_{\mu}^{2} + \beta^{2}] + \epsilon \xi_{\mu}^{2}]}$$
(42a)

$$\sigma^{2} \approx \frac{\pi \xi_{\alpha} G_{\alpha\alpha}}{8\omega_{\alpha} [\xi_{1} \xi_{2} (4\xi_{\alpha}^{2} + \beta^{2}) + \epsilon \xi_{\alpha}^{2}]}$$
(42b)

Note the presence of the parameter ϵ : The response of the equipment to ground motion is not independent of the mass ratio as is commonly assumed in the current design practice. In fact, as the value for ϵ increases, the mean square of the response decreases, as shown in Fig.3.5d. This behavior is a consequence of the interaction between the equipment and the structure, as was described by Newmark [33] and others [31,37].

For perfectly tuned systems, Eq.42a reduces to

$$\sigma_{i}^{2} \approx \frac{\pi G_{vv}}{8\omega_{i}^{3}\xi_{v}(4\xi_{1}\xi_{2}+\epsilon)}$$
(43)

This can be compared with the expression derived by Der Kiureghian, et. al. [19]

$$\sigma_{i} \approx \frac{\pi G_{ev}}{8\omega_{a}^{3}\xi_{a}(4\xi_{a}^{2} + \epsilon)}$$
(44)

which was based on the assumption that the system was classically damped. The two expressions are in agreement only when $\xi_i = \xi_{ij}$, which, as noted in the previous chapter, corresponds to the classical damping condition. As the difference between the damping ratios increases, the system becomes non-classically damped, and Eq.44 tends to underestimate the true value for σ_{ij}^2 , particularly for small values of ϵ . The plots of σ_{ij}^2 in Fig.3.5c using the above formulae demonstrate this phenomena. This was also shown in Chapter 2 using modal decomposition methods (see Fig.2.5). Note that for small differences of damping ratios, the formula based on classical damping will yield reasonable results.

For systems that are not perfectly tuned, the square of the detuning parameter β appears in the denominator in Eq.42a. Thus, as the system becomes increasingly detuned, the mean square of the response will decrease, as is expected. This behavior was exhibited in Figs.3.5a-e; note that σ_i^2 is symmetric with respect to β . Other useful quantities that can be derived from the spectral moments are the mean zero crossing rate $\nu = \sqrt{\lambda_0/\lambda_0}/\pi$ and the shape factor $\delta = \sqrt{1 - \lambda_0/(\lambda_0\lambda_0)}$. Using the second order expressions for λ_{∞} in Eqs.41a-d, the following are derived

$$\nu = \frac{\omega_a}{\pi} \left[1 - \frac{\beta \xi_a}{2\xi_a} \right] \tag{45a}$$

$$\delta = \left[\frac{\beta^2}{4}\left(1 - \frac{\xi_0^2}{4\xi_0^2}\right) + \frac{1}{4}\epsilon + \xi_0\xi_0\right]$$
(45b)

The accuracies of the above formulae are shown in Figs.3.6a-c. The factor ν is physically interpreted as the average frequency of zero crossings of the response process [44] and the above expression is in agreement with this interpretation. The shape factor δ is a measure of the band-width of the response process [44]. For a SDOF oscillator with frequency ω and damping ratio ξ ,

$$\delta \approx 2 \sqrt{\frac{\xi}{\pi}} \tag{46}$$

For small values of ξ , δ is small reflecting the fact that the response process is very narrowly banded. For the 2-DOF ES system, δ is a more complex function of the parameters β , ξ , and ϵ . The general behavior of δ is as follows:

- 1. For perfectly tuned systems and very small mass ratios, $\delta \approx \sqrt{\xi_1 \xi_2}$ which is an order of magnitude smaller than the shape factor for a SDOF oscillator in Eq.46.
- 2. As ϵ increases, δ increases (see Fig.3.6b). This reflects the fact that the frequencies of the system are moving apart which increases the band-width of the response.
- 3. As β increases, δ increases for the same reason as above (see Fig.3.6c).

All of the results derived in this section are based on white-noise input. For a general input specified by an arbitrary PSD $G_{cc}(\omega)$, two alternative methods can be used to find the response. The most straightforward method is numerical integration: the spectral moments can be computed using Eq.39 for any input PSD. However, if $G_{cc}(\omega)$ is a smoothly varying and wide-banded function such as the PSD for the Kanai-Tajimi filtered white-noise input [24], the highly peaked property of the transfer function $T_{ij}(\omega)$ permits the approximate use of the same

expressions for λ_{in} as those for white-noise input. The only difference is that the term G_{in} in Eqs.41a-e is replaced by the value of the PSDF at the peak of $T_{in}(\omega)$, i.e. $G_{in}(\omega_a)$.

3.4 Modal Decomposition Approach

3.4.1 Introduction

An alternative method of deriving the spectral moments of response of general 2-DOF systems is through the modal decomposition approach for non-classically damped systems, as developed in Chapter 2. The analysis will begin with tuned systems and the results compared with those of the previous section. It will be shown that the two sets of results are equivalent, the primary difference being the order of the approximation: Only lowest order expressions can be readily calculated in the modal approach. The analysis of the equipment-structure system is extended to detuned systems with results similar to those of previous investigators [18, 19, 35].

To conclude the modal analysis, general expressions which apply for both tuned and detuned systems are formulated using matching techniques from perturbation theory.

3.4.2 Tuned Systems

3.4.2.1 Mode shapes

The basic relationship between the mode shapes ϕ , and frequencies ω , of a general structural system is given by the eigenvalue problem

$$\left[-\omega\right]^{2}\mathbf{M} + i\omega\left[\mathbf{C} + \mathbf{K}\right]\boldsymbol{\phi}_{1} = 0 \tag{47}$$

For the 2-DOF system, the frequencies ω_i^* have been derived earlier by solving the characteristic equation. Consequently, the mode shapes can be computed readily by substituting the expressions in Eq.12 for the mass, damping, and stiffness matrices and the expressions in Eq.28 for the frequencies. Using the notation $\phi_i^* = \{\alpha_i, 1\}^T$, the equations for the eigenvectors are

$$\begin{bmatrix} G_1(\boldsymbol{\omega}_i) & f_1(\boldsymbol{\omega}_i) \\ f_1(\boldsymbol{\omega}_i) & g_1(\boldsymbol{\omega}_i) \end{bmatrix} \begin{bmatrix} \alpha_i \\ 1 \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \end{bmatrix} \quad i=1,2$$
(48)

Solving for α_i and taking lowest order terms yields

$$\alpha = -\frac{g_{\mu}(\omega)}{f_{\mu}(\omega)} \simeq \frac{2[\omega_{\mu}(1+i\xi_{\mu})-\omega]}{\omega_{\mu}}$$
(49)

Note that the numerator is simply the difference between the frequency of the fixed base secondary subsystem and the frequency of the combined system. Substituting Eq.28 for ω^2 , the eigenvalue, eigenvector solution pairs for Eq.47 are

$$\omega_1 = \omega_1 \left[1 + i\xi_0 + \frac{1}{2}\sqrt{\epsilon + (i\xi_0 + \beta)^2} \right], \quad \alpha_1 = -\beta - i\xi_0 - \sqrt{\epsilon + (i\xi_0 + \beta)^2}$$
(50a)

$$\omega_1^2 = \omega_1 \left[1 + i\xi_0 - \frac{1}{2}\sqrt{\epsilon + (i\xi_0 + \beta)^2} \right], \quad \alpha_2 = -\beta - i\xi_0 + \sqrt{\epsilon + (i\xi_0 + \beta)^2}$$
(50b)

The component α_{i} is compared with exact values, computed by numerically solving the eigenvalue problem, in Table 2. The exact and approximate values of α_{i} are also plotted in Figs.3.7a-c using the same values for the parameters used in Figs.3.3a-c. The key relationships between the mode shapes ϕ_{i} and the system parameters are:

- 1. For very small values of the mass ratio ϵ , α_2 is very nearly zero, and the mode shape ϕ_2^{-1} is dominated by equipment motion. Similarly, mode 1 is associated with the structure subsystem and α_1 is approximately $-2(\beta + i\xi_{\alpha})$. This is indicated in Figs.3.7a,b, where α are shown to converge to the limiting values 0 and $-2(\beta + i\xi_{\alpha})$ for small values of ϵ . This corresponds to the convergence of the frequencies of the combined system to those of the subsystems for increasingly smaller values of the mass ratio, ϵ .
- 2. If the system is perfectly tuned and $\xi_d^2 < \epsilon$, then α_l and α_2 would be given by

$$\alpha_1 = -i\xi_d - \sqrt{\epsilon - \xi_d^2}$$
(51a)
$$\alpha_2 = -i\xi_d + \sqrt{\epsilon - \xi_d^2}$$
(51b)

which have the same absolute value. Thus, the characteristics of the two subsystems would be distributed equally to both modes; it is no longer possible to associate a mode with either subsystem. This demonstrates some of the symmetry of the system. This is shown graphically in Fig.3.7b for points A and B, where the coordinates α lie to each side of the imaginary axis at a distance of approximately $\sqrt{\epsilon - \xi_{0}^{2}}$.

3. The mode shapes will be real-valued only when the damping ratios of the equipment and structure are equal. This follows since the damping matrix would be proportional to the
stiffness matrix to lowest order terms, and the system, by definition, attains classical damping. This can be observed in Fig.3.7c, where the imaginary components of α are relatively small.

4 Finally, the product of α_1 and α_2 is

 $\alpha_1 \alpha_2 \approx -\epsilon$ (52) which is constant with respect to $\beta_1 \xi_1$, and ξ_2 . This relation will be used in the next section.

3.4.2.2 Spectral Moments

The modal decomposition method can be applied directly to the results in Eqs.50a,b to obtain the spectral moments of response.

First, new notation will be introduced. Following the convention in Chapter 2, the frequencies ω' can be rewritten

$$\omega' = \tilde{\omega} \sqrt{1 - \tilde{\xi}^2} + i \tilde{\xi} \, \tilde{\omega} \tag{53a}$$

where $\hat{\omega}$ and $\hat{\xi}$ are the undamped frequencies and the damping ratios, respectively, which are associated with ω^2 and are given by

$$\tilde{\omega} = |\omega|^2 \approx \operatorname{Re}\omega^2$$
 and $\xi = \frac{\operatorname{Im}\omega^2}{\tilde{\omega}}$ (53b)

In general, the system parameters $\tilde{\omega}_{\perp}$ and $\tilde{\xi}_{\perp}$ are not related to the original parameters ω_{\perp} and ξ_{\perp} , however, the averages are the same. This fact is verified by referring to the expressions in Eq.50a,b

$$\frac{\tilde{\omega}_1 + \tilde{\omega}_2}{2} \approx \frac{\operatorname{Re}(\tilde{\omega}_1^* + \omega_2)}{2} \approx \omega_{ii}$$
(54a)

$$\frac{\tilde{\xi}_1 + \tilde{\xi}_2}{2} = \frac{1}{2} \left[\frac{|\mathbf{m}\omega_1|}{\tilde{\omega}_1} + \frac{|\mathbf{m}\omega_2|}{\tilde{\omega}_2} \right] \approx \frac{1}{2} \left[\frac{|\mathbf{m}(\omega_1 + \omega_2)|}{\omega_0} \right] \approx \xi_0$$
(54b)

It is useful to define the differences

$$\tilde{\boldsymbol{\beta}} = \frac{(\tilde{\omega}_1 - \tilde{\omega}_2)}{\omega_a} \text{ and } \tilde{\boldsymbol{\xi}}_d = \tilde{\boldsymbol{\xi}}_1 - \tilde{\boldsymbol{\xi}}_2 \tag{55}$$

From Eqs.50a,b and Eq.53b, it can be shown by direct substitution that

$$|\hat{\boldsymbol{\beta}}^{2} + \hat{\boldsymbol{\xi}}_{\beta}| \approx |\boldsymbol{\epsilon} + (\boldsymbol{\eta} \boldsymbol{\xi}_{\beta} + \boldsymbol{\beta})^{2}|$$
(56)

This relation will be useful in computing the spectral moments

€

The first step in the modal decomposition method is the evaluation of the three sets of constants C_{\pm} , D_{\pm} , and E_{\pm} defined in Eqs.2.26 and 2.34. In order to obtain these quantities, some preliminary expressions must be computed as outlined in Chapter 2.

The quantities A and b defined by Eqs.2.17a and 2.20 can be obtained in terms of α and

$$A = \hat{\phi}^{T} A \hat{\phi} \approx 2i \phi^{-T} M \phi^{T} \omega_{T} + 2i (\alpha^{T} + \epsilon) m_{T} \omega_{T}$$
(57a)

$$b = -\left[\frac{\phi^{\prime} Mr}{4}\right] q^{\prime} \phi \approx -\frac{i(\alpha + \epsilon)}{2(\alpha + \epsilon)\omega_{\alpha}} \approx -\frac{i}{2(\alpha + \alpha - \epsilon)\omega_{\alpha}}$$
(57b)

The formula for b can be expressed in terms of α_1 and α_2 by using Fq.52.

$$b = \frac{\ell(-1)}{2\omega_{\mu}(\alpha_{\mu} - \alpha_{\nu})}$$
(58)

Note that $b = -b_{\infty}$. The constants a and c defined by Eq.2.26 can be evaluated in terms of b using the relation $s = i\omega$.

$$C \approx 4\omega_{\rm c}({\rm lin}b_{\rm c})({\rm lm}b_{\rm c}) = D_{\rm c} \approx 0 - E_{\rm c} \approx 4({\rm Re}b_{\rm c})({\rm Re}b_{\rm c})$$
 (60)

$$c = 2 \operatorname{Re} b \tag{59b}$$

The lowest order terms for C_i , D_j , and E_j are calculated using Eq.2.34

$$C_{\perp} \approx 4\omega_{0}^{2}(\mathrm{Im}b_{1})^{2} \quad D_{\perp} \approx 0 \quad E_{\perp} \approx 4(\mathrm{Re}b_{1})^{2} \tag{60}$$

It was noted in Chapter 2 that the coefficient D_{ij} was always nearly equal to zero in the numerical studies of the tuned 2-DOF equipment-structure system. The expressions above confirm that result.

The next step in the modal decomposition method is the evaluation of the cross spectral moments λ_{μ} , for white-noise input. By using the parameters $\tilde{\omega}_{-}$ and $\tilde{\xi}_{+}$, the computation is straightforward using Eqs.2.49a-h. The imaginary parts of $\lambda_{\mu-1}$ are nearly zero due to the fact that the frequencies $\tilde{\omega}_{+}$ are closely specified for tuned systems. Therefore, only the real parts of the cross moments are needed. The zeroeth cross moments are

$$\operatorname{Re\lambda}_{0,\omega} \approx \frac{\pi G_{cc}}{4\omega_0^2 \tilde{\xi}_0} \quad i = 1,2$$
(61a)

$$\operatorname{Re}_{\lambda_{0+2}} = \frac{\operatorname{Pr}(G_{cv}(\hat{\xi}_{1}\hat{\omega}_{1}+\hat{\xi}_{2}\hat{\omega}_{2}))}{(\hat{\omega}_{1}-\hat{\omega}_{2})^{2}+4(\hat{\xi}_{1}\hat{\omega}_{1}+\hat{\xi}_{2}\hat{\omega}_{2})(\hat{\xi}_{1}\hat{\omega}_{2}+\hat{\xi}_{2}\hat{\omega}_{1})\hat{\omega}_{1}\hat{\omega}_{2}} \approx \frac{\pi G_{cv}}{\omega_{a}} \frac{\xi_{a}}{\hat{\beta}^{2}+4\xi_{a}^{2}}$$
(61b)

As an initial low-order approximation, higher cross moments can be obtained by using the fol-

lowing

$$\mathbf{Re}\boldsymbol{\lambda}_{0} \equiv \boldsymbol{\omega}_{0}^{T}\mathbf{Re}\boldsymbol{\lambda}_{0} \qquad (62)$$

which will yield low order expressions for λ_{μ}

The final step in the modal decomposition method is the evaluation of λ_{12} by substituting the preceding results into Eq.2.36. The lowest order expression for λ_{12} is

$$\lambda_{\mu} \approx \sum_{i} \sum_{j} \left\{ 4\omega_{i}^{2} \mathrm{Im} h \mathrm{Im} h \mathrm{Re} \lambda_{\mu} |_{i} + 4 \mathrm{Re} h \mathrm{Re} b_{i} \omega_{i}^{2} \mathrm{Re} \lambda_{\mu} |_{i} \right\}$$
$$= 4\omega_{i}^{2} \left\{ h \left[\left[\mathrm{Re} \lambda_{\mu} \right] + \mathrm{Re} \lambda_{\mu} |_{i} \right] - 2 \mathrm{Re} \lambda_{\mu} |_{i} \right\}$$
(63)

The term $|b_1|^2$ is found by substituting Eqs.50a,b into Eq.58 and utilizing Eq.56

$$|b_{i}|^{2} = \frac{1}{16} \left| \omega_{a} \sqrt{\epsilon + (i\xi_{a} + \beta)^{2}} \right|^{2} = \frac{1}{16\omega_{a}} \left[\hat{\beta}^{2} + \hat{\xi}_{a}^{2} \right]^{-1}$$
(64)

After substituting Eqs.61a,b and 64 into Eq.63, the expression for the moment λ_{μ} reduces to

$$\lambda_{\mu} = \frac{\pi G_{\mu\nu}}{8\omega_{\mu}^{3/\mu}} \frac{\xi_{\mu}}{\xi_{1}\xi_{2}(\tilde{\mu}^{2} + 4\xi_{\mu}^{2})}$$
(65a)

This can be compared with the formula for λ_{\pm} in the preceding section which, to lowest order is

$$\lambda_{\mu} = \frac{\pi G_{\alpha}}{8\omega_{\mu}^{3/\mu}} \frac{\xi_{\mu}}{[\xi_{\mu}\xi_{\mu}(\beta^{2} + 4\xi_{\mu}^{2}) + \epsilon\xi_{\mu}^{2}]}$$
(65b)

The latter expression is entirely in terms of the original parameters $\omega_{i,j}$, $\xi_{i,j}$, and ϵ while the former is in terms of the derived parameters $\tilde{\omega}_{i,j}$ and $\tilde{\xi}_{i,j}$. The mass ratio $\epsilon_{i,j}$ which is present in Eu.65b is implicitly included in Eq.65a in the derived parameters. It can be shown that the two expressions are equivalent by rewriting the derived parameters in terms of the original parameters.

Higher order refinements can be made to the expression in Eq.65a for m > 0 by using the results of Section 3.3.4. From the definitions of the factors ν and δ , it is possible to derive the following relationships between the first three moments:

$$\lambda_{1} = \pi \nu \sqrt{1 - \delta} \lambda_{0} \quad \lambda_{2} = \pi (\nu) \lambda_{0} \tag{66}$$

Substituting the expressions for ν and δ from Eqs.45a,b into the above, higher order relations are obtained which are equivalent to the expressions in Eqs.41.

3.4.3 Detuned Systems

3.4.3.1 Introduction

Thus far, the analysis of the 2-DOE equipment-structure system has been restricted to tuned systems. For detuned systems, the parameter β is large and can not be treated as a perturbation variable, consequently a new set of expressions must be developed. The analysis is straightforward through the application of the modal decomposition method.

3.4.3.2 Frequencies and Mode Shapes

The eigenvalue problem for the 2-DOF system can be rewritten as

$$\Gamma(\omega^{2})\phi^{2} = 0 \tag{67a}$$

where

$$\Gamma(\omega^{-1}) = \begin{bmatrix} G_{1}(\omega^{-1}) & f_{2}(\omega^{-1}) \\ f_{1}(\omega^{-1}) & g_{1}(\omega^{-1}) \end{bmatrix}$$
(67b)

To begin the analysis, the following approximations for the frequencies and mode shapes are used.

$$\boldsymbol{\omega}^{\text{train}} = \boldsymbol{\omega} \left(\sqrt{1 - \boldsymbol{\xi}^{\text{train}} + i\boldsymbol{\xi}} \right)$$
(68a)
$$\boldsymbol{\phi}^{\text{train}}_{\text{train}} = \begin{bmatrix} 1 \\ 0 \end{bmatrix} \quad \boldsymbol{\phi}^{\text{train}}_{\text{train}} = \begin{bmatrix} 0 \\ 1 \end{bmatrix}$$
(68b)

The 0 superscript is used to indicate that the above are initial approximations. Low-order refinements of these rough estimates are possible using perturbation techniques. Details are presented for mode 1 which is associated with the structure mode.

The above approximations have errors, which are found by direct substitution into the left-hand-side of Eq.67a

$$\boldsymbol{\Gamma}(\boldsymbol{\omega}_{1}^{(0)})\boldsymbol{\phi}_{1}^{(0)} = \begin{bmatrix} G_{1}(\boldsymbol{\omega}_{1}^{(0)}) & f_{1}(\boldsymbol{\omega}_{1}^{(0)}) \\ f_{1}(\boldsymbol{\omega}_{1}^{(0)}) & g_{1}(\boldsymbol{\omega}_{1}^{(0)}) \end{bmatrix} \begin{bmatrix} 1 \\ 0 \end{bmatrix} \approx \begin{bmatrix} G_{1}(\boldsymbol{\omega}_{1}^{(0)}) \\ f_{1}(\boldsymbol{\omega}_{1}^{(0)}) \end{bmatrix}$$
(69)

.

Using Eqs.17a-c to evaluate the above polynomials, the order of magnitude of the above errors can be found

$$\Gamma(\omega_1^{(0)})\boldsymbol{\phi}_1^{(0)} = \begin{bmatrix} O(\boldsymbol{\epsilon}) \\ O(\boldsymbol{\epsilon}) \end{bmatrix} m_1 \omega_1^2$$
(70)

To reduce the above errors, a refined mode shape and frequency, which will be denoted $\phi^{(1)}$ and $\omega^{(1)}$, will be derived.

The first step in reducing the errors is finding a suitable value for the equipment component of ϕ^{\pm} . By examining Eq.69, it is clear that if the following value is given to the refined mode shape

$$\boldsymbol{\Delta}_{1}^{(1)} = \begin{bmatrix} 1 \\ -\frac{f_{1}(\boldsymbol{\omega}_{1}^{(0)})}{g_{1}(\boldsymbol{\omega}_{1}^{(0)})} \end{bmatrix}$$
(71a)

the error in the second coordinate in Eq.69 would be zero and the new error terms would be

$$\mathbf{\Gamma}(\boldsymbol{\omega}_{1}^{(0)})\boldsymbol{\phi}_{1}^{(1)} = \begin{bmatrix} G_{1}(\boldsymbol{\omega}_{1}^{(0)}) - \frac{f_{1}^{(2)}(\boldsymbol{\omega}_{1}^{(0)})}{g_{1}(\boldsymbol{\omega}_{1}^{(0)})} \\ 0 \end{bmatrix}$$
(71b)

The error in the first coordinate can be reduced by finding the refined frequency $\omega_1^{(1)}$ that satisfies

$$G_{\lambda}(\boldsymbol{\omega}_{\lambda}^{(i)}) = \frac{f_{\lambda}^{(i)}(\boldsymbol{\omega}_{\lambda}^{(0)})}{g_{\lambda}(\boldsymbol{\omega}_{\lambda}^{(0)})}$$
(72)

Let $\Delta \omega_1 = \omega_1^{(1)} = \omega_1^{(0)}$. Then $G_1(\omega)$ can be reduced to a linear function in $\Delta \omega_1$ by using the derivative approximation

$$G_{1}(\omega_{1}^{(1)}) - G_{1}(\omega_{1}^{(0)}) \approx \frac{d}{d\omega} G_{1}(\omega_{1}^{(0)}) \Delta \omega_{1}$$

$$\approx 2m_{1}[-\omega_{1}^{(0)} + i\xi_{1}\omega_{1}] \Delta \omega_{1}$$

$$\approx -2m_{1}\omega_{1}\Delta \omega_{1}$$
(73)

The solution to $\Delta \omega_{\pm}$ is found by combining Eqs.72 and 73

$$\Delta \omega_{\pm} = -\frac{1}{2m_{\pm}\omega_{\pm}} \left\{ \frac{f_{\pm}^{(2)}(\omega_{\pm}^{(0)})}{g_{\pm}(\omega_{\pm}^{(0)})} - G_{\pm}(\omega_{\pm}^{(0)}) \right\}$$
(74)

It is clear that $\Delta \omega_1 = O(\epsilon) \omega_1$ from the definitions of f_1, g_1 , and G_1 . By comparing each ele-

م د معود الد ال

ment of the matrix $\Gamma(\omega_i^{(0)})$ with the corresponding element in $\Gamma(\omega_i^{(0)})$, it can be shown that

$$\Gamma(\omega^{(i)})\phi^{(i)} = \begin{bmatrix} O(\epsilon^2) \\ O(\epsilon^2) \end{bmatrix} m_1 \omega_1^2$$
(75)

Thus, the refined mode shape and frequency have errors an order of magnitude smaller than those of the initial solutions. Further refinements are possible, however the small second order improvements in accuracy generally does not warrant the increased analytical and computation effort, particularly in engineering applications. Using the * notation for the final low-order approximations, the above expressions reduce to

$$\boldsymbol{\phi}^{\dagger} \approx \left\{ \frac{1}{\frac{\omega_{\gamma}}{\omega_{\gamma} - \omega_{\gamma}}} \right\} \quad \omega_{1}^{\dagger} \approx \omega_{1}(1 + i\xi_{\gamma}) \tag{76}$$

Although the expression for $\omega_1^{(1)}$ is not used in the final result, its derivation is necessary to prove that the solutions $\phi_1^{(1)}$ and $\omega_1^{(1)}$ reduce the error in the eigenvalue problem. This proof justifies that the solutions in Eq.76 are valid low-order approximations.

In a very similar procedure, refined solutions can be obtained for the mode shape and the frequency of mode 2, which is associated with the equipment mode. The high-order expressions are:

$$\boldsymbol{\phi}_{2}^{(1)} = \begin{bmatrix} \frac{f_{1}(\boldsymbol{\omega}_{2}^{(0)})}{G_{1}(\boldsymbol{\omega}_{2}^{(0)})} \\ 1 \end{bmatrix} \quad \boldsymbol{\omega}_{2}^{(1)} = \boldsymbol{\omega}_{2}^{(0)} + \frac{f_{1}^{2}(\boldsymbol{\omega}_{2}^{(0)})}{2m_{1}\epsilon\omega_{2}G_{1}(\boldsymbol{\omega}_{2}^{(0)})}$$
(77)

which reduce to the following, final low-order approximations

$$\boldsymbol{\phi}_{1}^{*} \approx \begin{bmatrix} \frac{\boldsymbol{\epsilon}\omega^{2}}{\omega_{1}^{2} - \omega_{1}^{2}} \\ 1 \end{bmatrix} \quad \boldsymbol{\omega}_{2}^{*} \approx \boldsymbol{\omega}_{2}(1 + i\xi_{2})$$
(78)

Note the similarity between the above results for mode 2 and those for mode 1.

The mode shapes and frequencies for detuned systems have an entirely different character than those of tuned systems. The mode shapes are real, indicating that detuned systems are classically damped. Also, the frequencies of the combined system are close to the fixed hase frequencies of each subsystem indicating that interaction plays a negligible role in detuned systems.

3.4.3.3 Spectral Moments of Response

As in Section 3.4.2.2, the modal decomposition approach can be applied directly to the above mode shapes and frequencies to obtain the spectral moments of response. Since detuned systems are classically damped, the method for classically damped systems, as developed by Der Kureghian [16], can be used. The effective participation factor Ψ is given by

$$\Psi = \frac{(q^{\dagger} \phi^{\dagger})(\phi^{\dagger} \mathbf{M}\mathbf{r})}{M}$$
(79)

where $M = \phi' M \phi$ is the modal mass. By substituting the preceding expressions for ϕ' into Eq.79 and taking lowest-order terms, the result is

$$\psi_{1} \approx -\frac{\omega_{1}^{2}}{\omega_{1}^{2} - \omega_{2}^{2}} \quad \psi_{1} \approx -\psi_{1}$$
(80)

The expressions for the moments λ_{μ} , are easily found in terms of ω_{μ}

$$\lambda_{\mu} \approx \frac{\pi G_{\mu} \omega^{\mu}}{4\ell} \tag{21}$$

As for the cross-spectral moments λ_{m-1} , it has been established that the cross spectral moments $\lambda_{m-2} \approx 0$ for well-spaced modes [15]. It will be shown in Section 3.4.4.3 that detuned modes are well-spaced; therefore this approximation for λ_{m-1} is applicable here.

The spectral moments are calculated using the formula Eq.2.38

$$\lambda_{m} = \sum_{i=1}^{2} \sum_{j=1}^{2} \psi_{i} \psi_{j} \lambda_{m-i} \approx \left[\frac{\omega_{1}^{2}}{\omega_{1}^{2} - \omega_{2}^{2}} \right]^{2} \frac{\pi G_{\mathrm{ec}}}{4} \left[\frac{\omega_{1}^{m-3}}{\xi_{1}} + \frac{\omega_{2}^{m-3}}{\xi_{2}} \right]$$
(82)

The most notable characteristic of the above expression is that λ_m decreases as the system becomes increasingly defuned. This is illustrated in Fig.3.8.

3.4.4 Comparison between Tuned and Detuned Systems

3.4.4.1 Introduction

The 2-DOF system has been considered as being either tuned or detuned with different sets of expressions derived for each case. However, for practical applications a general expression applicable for all systems would be more useful, particularly for those systems which the between the tuned and detuned categories. This is accomplished by matching, which is a standard procedure from perturbation methods [32]. Also, guidelines are established to determine if a system is characterized by tuning or detuning.

3.4.4.2 A General Expression for A .

The idea behind matching is to combine two expressions into one general expression that would apply for all values of a perturbation parameter. In this case, the expression that will be evaluated is the spectral moment λ_{\pm} , and the perturbation parameter is the detuning, β_{\pm}

For systems with large values of β , the detuned moment in Eq.82 would be used, and for systems with small values of β , the tuned moment in Eq.65b would be used. However, for systems with values of β which are large enough to be considered detuned yet small enough to be considered tuned, both expressions for λ , would be applicable. The detuned version can be rewritten in terms of β

$$\mathbf{\lambda}^{\mu} = \frac{\omega^{\mu} \pi G}{16\beta^{\mu} \omega_{\mu}^{\mu}} \frac{\omega^{\mu} \varepsilon^{\mu} \xi_{\mu} + \omega_{\mu}^{\mu} \varepsilon^{\mu} \xi_{\mu}}{\omega^{\mu} \varepsilon^{\mu} \varepsilon^{\mu} \xi_{\mu} \xi_{\mu}}$$
(83)

where the *d* superscript denotes detuning. The tuned version can be rewritten to account for the relatively large value for β

$$\lambda_{\perp} = \frac{\pi G_{c_1}(\xi + \xi_2)}{16\beta' \omega_{a}^{-1} \xi_1 \xi_2}$$
(84)

where the *t* superscript denotes tuning. Since the two expressions above are derived using different assumptions on the parameter β , they are slightly different. However, it is clear that small modifications can be made to λ_{\perp}^{\perp} so that the expressions would agree, or match. If these modifications are applied to the original expression for λ_{\perp}^{\perp} in Eq.65b, the result would be

$$\lambda_{\pm} = \frac{\pi G_{ee}}{16\omega_{\pm}^{4}} \frac{\omega_{\pm}^{4}}{\omega_{\pm}^{4}} \frac{\omega_{\pm}^{3}}{\omega_{\pm}^{2}} \frac{\omega_{\pm}^{3}}{|\xi_{\pm}\xi_{\pm}(4\xi_{\pm}^{2}+\beta^{2})+\epsilon\xi_{\pm}^{3}|}$$
(85)

Note that when β is small, $\omega_1 \approx \omega_2$ and the above expression reduces to the original expression in Eq.65b. For larger values of β , i... above will closely approximate the defuned moment λ_{\perp}^{μ} . This can be clearly seen in Fig.3.9.

3.4.4.3 Detuning Criteria

For systems with large values for the parameter β , the detuned moment λ_m^d in Eq.83 would be in close agreement with the general expression for λ_m in Eq.85. However, systems with sufficiently small values for β would be characterized by tuning and λ_m^d and λ_m would no longer be in agreement.

If the relative error tolerance is e_{λ} then a system will be defined to be detuned if the difference between λ_{a}^{d} and λ_{a} is less than e_{λ} i.e.

$$\frac{\text{relative}}{\text{error}} = \frac{\lambda_{w}^{d} - \lambda_{w}}{\lambda_{w}} < e$$
(86)

Substituting Eqs.83 and 85 into the above yields

$$\frac{\text{relative}}{\text{error}} = \frac{\xi_{i}^{2}(4\xi_{1}\xi_{2} + \epsilon)}{\xi_{i}\xi_{j}\beta_{i}^{2}} < e$$
(87)

This error is plotted in Fig.3.10; clearly in the vicinity of perfect tuning $(\beta - 0)$ the error becomes very large. Rewriting the above in terms of β yields

$$\beta^{2} > \frac{4\xi_{1}\xi_{2} + \epsilon}{\xi_{1}\xi_{2}e} \xi_{a}^{2} = \frac{1}{e} \left[4 + \frac{\epsilon}{\xi_{1}\xi_{2}} \right] \xi_{a}^{2}$$

$$(88)$$

which will be used hereafter to define dejuning.

It was noted earlier that detuned systems have widely-spaced modes. This hypothesis will be proven, presently. In the context of the present study, two modes are defined to be widely spaced if the correlation coefficients for white noise input $p_{m,n} <<1$ for m=0,1,2 and $i\neq j$. This relation will be shown to be true for the 2-DOF detuned system for m=0; similar proofs hold for m=1 and 2.

Using Eq.2.56a, the expression for $\rho_{0,12}$ can be written in terms of the parameters of the PS system:

$$P_{0,12} = \frac{\sqrt{\xi_1 \xi_2} (4\xi_d + \beta \xi_d)}{4\xi_d^2 + \beta^2}$$
(89)

The relation in Eq.88 can be simplified to

$$\beta > \frac{2\xi_a}{\sqrt{e}} \tag{90}$$

which, when substituted for the first summand in the numerator of Eq.89, yields

$$\rho_{0,12} < \frac{\sqrt{\xi_1 \xi_2} (2\sqrt{e_1} + \xi_n) \beta}{4\xi_n^2 + \beta^2}$$
(91)

Dividing through by β , dropping the first summand in the denominator, and substituting Eq.90 to the remaining summand in the denominator, the above simplifies to

$$p_{0+2} < \frac{\sqrt{\xi_1 \xi_2} (2\sqrt{e_1} + \xi_d)}{\frac{2\xi_d}{\sqrt{e_1}}}$$
(92)

Finally, using the fact that the geometric mean is less than or equal to the arithmetic mean, the above reduces to the following inequality

$$|\mathbf{p}_{0,1}| \le |\mathbf{e}| + \frac{4}{2}\sqrt{\mathbf{e}}|\mathbf{\xi}_{d}| << 1$$
(93)

which proves that detuned modes are widely spaced.

3.5 Non-Interaction Results

All of the results derived thus far in this chapter correctly account for the effect of interaction between the structure and the equipment. A question with practical implications is: What is the difference between these results and the results which would be obtained if interaction was neglected?

In the derivation of non-interaction results the response is first found for the structure alone without accounting for the equipment. Then the structural motion is used as the base input to the equipment. In mathematical terms, the equations for the system response which were coupled in Eq.13 for the interaction study are decoupled in the non-interaction analysis.

The structural response $x_1^{(max)}$ relative to the ground is given by

$$m_1 \ddot{x}_1^{(max)} + c_1 \dot{x}_1^{(max)} + k_1 x_1^{(max)} = -m_1 \ddot{x}_c(t)$$
(94a)

where the superscript (non) indicates that the variables are the results of the non-interaction analysis. The motion at the base of the equipment is $x_1^{(non)} + x_n$, therefore the equipment response $y^{(non)}$ relative to the attachment point is

$$m_y \hat{y}^{(non)} + c_y \hat{y}^{(non)} + k_y y^{(non)} = -m_y (\hat{x}_1^{(non)} + \hat{x}_y)$$
 (94b)

The Fourier transform $X_1^{(mon)}(\omega)$ and $Y^{(mon)}(\omega)$ of the displacements $x_1^{(mon)}(t)$ and $y^{(mon)}(t)$ are

3.4.4.3 Detuning Criteria

For systems with large values for the parameter β , the defuned moment λ_{m}^{d} in Fq.83 would be in close agreement with the general expression for λ_{m} in Eq.85. However, systems with sufficiently small values for β would be characterized by tuning and λ_{m}^{d} and λ_{m} would no longer be in agreement.

If the relative error tolerance is e, then a system will be defined to be detuned if the difference between λ_{ee}^d and λ_{ee} is less than e, i.e.

$$\frac{\text{relative}}{\text{error}} = \frac{\lambda_m^d - \lambda_m}{\lambda_m} < e$$
(86)

Substituting Eqs.83 and 85 into the above yields

$$\frac{\text{relative}}{\text{error}} = \frac{\xi_a^2(4\xi_1\xi_2 + \epsilon)}{\xi_1\xi_2\beta^2} < \epsilon$$
(87)

This error is plotted in Fig.3.10; clearly in the vicinity of perfect tuning ($\beta=0$) the error becomes very large. Rewriting the above in terms of β yields

$$\beta^{2} > \frac{4\xi_{1}\xi_{2} + \epsilon}{\xi_{1}\xi_{2}e} \xi_{a}^{2} = \frac{1}{e} \left[4 + \frac{\epsilon}{\xi_{1}\xi_{2}} \right] \xi_{a}^{2}$$
(88)

which will be used hereafter to define detuning.

It was noted earlier that detuned systems have widely-spaced modes. This hypothesis will be proven, presently. In the context of the present study, two modes are defined to be widely spaced if the correlation coefficients for white noise input $p_{m,q} \ll 1$ for m=0,1,2 and $i \neq j$. This relation will be shown to be true for the 2-DOF detuned system for m=0; similar proofs hold for m=1 and 2.

Using Eq.2.56a, the expression for ρ_{0,j_2} can be written in terms of the parameters of the PS system:

$$\rho_{0,12} = \frac{\sqrt{\xi_1 \xi_2} (4\xi_a + \beta \xi_a)}{4\xi_a^2 + \beta^2}$$
(89)

The relation in Eq.88 can be simplified to

$$\beta > \frac{2\xi_u}{\sqrt{e}} \tag{90}$$

which, when substituted for the first summand in the numerator of Eq.89, yields

$$\rho_{0,12} < \frac{\sqrt{\xi_1 \xi_2} (2\sqrt{c_1 + \xi_0})\beta}{4\xi_1^2 + \beta^2}$$
(91)

Divide β through by β , dropping the first summand in the denominator, and substituting Eq.90 to the remaining summand in the denominator, the above simplifies to

$$\rho_{0,12} < \frac{\sqrt{\xi_1 \xi_2} (2\sqrt{e_1} + \xi_n)}{\frac{2\xi_n}{\sqrt{e_1}}}$$
(92)

Finally, using the fact that the geometric mean is less than or equal to the arithmetic mean, the above reduces to the following inequality

$$p_{net} < e + \sqrt[n]{e} \xi_d \ll 1$$
(93)

which proves that detuned modes are widely spaced.

3.5 Non-Interaction Results

All of the results derived thus far in this chapter correctly account for the effect of interaction between the structure and the equipment. A question with practical implications is: What is the difference between these results and the results which would be obtained if interaction was neglected?

In the derivation of non-interaction results the response is first found for the structure alone without accounting for the equipment. Then the structural motion is used as the base input to the equipment. In mathematical terms, the equations for the system response which were coupled in Eq.13 for the interaction study are decoupled in the non-interaction analysis.

The structural response $x_i^{(max)}$ relative to the ground is given by

$$m_1 \tilde{x}_1^{(mon)} + c_1 \tilde{x}_1^{(mon)} + k_1 c_1^{(mon)} = -m_1 \tilde{x}_0(t)$$
(94a)

where the superscript (*non*) indicates that the variables are the results of the non-interaction analysis. The motion at the base of the equipment is $x_1^{(non)} + x_c$, therefore the equipment response $p^{(non)}$ relative to the attachment point is

$$m_{3} \dot{y}^{(non)} + c_{3} \dot{y}^{(non)} + k_{3} \dot{y}^{(non)} = -m_{3} (\ddot{x}_{1}^{(non)} + \ddot{x}_{c})$$
(94b)

The Fourier transform $X_1^{(mon)}(\omega)$ and $Y^{(mon)}(\omega)$ of the displacements $x_1^{(mon)}(t)$ and $y^{(mon)}(t)$ are

$$K_1^{(m,n)}(\omega) = -h_1(\omega)m_1X_{\nu}(\omega)$$
(95a)

$$Y^{(nan)}(\omega) = -h_2(\omega)m_2[-\omega^2 X_1^{(nan)}(\omega) + X_c(\omega)]$$

= $-h_2(\omega)m_2[\omega^2 h_1(\omega)m_1 + 1]X_c(\omega)$ (95b)

where $h_1(\omega)$ and $h_2(\omega)$ are the complex frequency response functions of the structure and equipment subsystems, respectively, given by

$$h_{i}(\omega) = (-\omega^{2}m + i\omega c_{i} + k_{i})^{-1}$$
(96)

Using this definition, the expression for $Y^{(max)}(\omega)$ can be reduced

$$Y^{(max)}(\omega) = -h_1(\omega)h_2(\omega)m_2[\omega^2 m_1 + h_1^{-1}(\omega)]X_{\nu}(\omega)$$

= $-h_1(\omega)h_2(\omega)m_2[i\omega c_1 + h_1]X_{\nu}(\omega)$
= $-h_1(\omega)h_2(\omega)m_1m_2[2i\omega_1\xi_1\omega + \omega_1^2]X_{\nu}(\omega)$ (97)

It follows from Eq.32 and Eq.34b that the transfer function $T_{11}^{(max)}(\omega)$ for the relative displacement response is

$$T_{ij}^{(mat)}(\omega) = \frac{(\omega_1^4 + 4\xi_j^2 \omega_j^2 \omega^2) m_j^2 m_j^2}{d^{(nat)}(\omega) d^{(nat)}(-\omega)}$$
(98a)

where

$$d^{(aoa)}(\omega) = h_1^{-1}(\omega)h_2^{-1}(\omega)$$
(98b)

Note the similarity between Eq.98a and the corresponding expression which includes interaction, Eq.35. The difference is in the denominator: in the non-interaction analysis $d(\omega)$ is replaced by $d(\omega)^{(\alpha\alpha\alpha)}$. Comparing Eq.17d with Eq.98b, it can be seen that

$$d(\omega) = d(\omega)^{(mm)} + \epsilon (additional terms)$$
(99)

where the additional terms account for the interaction between the equipment and the structure. As the mass ratio becomes very small, the additional interaction term in the above equation becomes negligible and $d(\omega)$ approaches $d(\omega)^{(num)}$, i.e.

$$\lim_{n \to 0} d(c - d(\omega)^{(n(\alpha))}$$
(100a)

It follows that a similar relation holds for the transfer function

$$\lim_{\epsilon \to 0^+} T_{ii}(\omega) = T_{ii}^{(mn)}(\omega)$$
(100b)

Using the above fact, the rederivation of the spectral moments and other related quantities for the non-interaction study becomes trivial. Most of the results found earlier can be adjusted to yield results that do not account for interaction by setting ϵ to zero, i.e., modeling the equipment as a subsystem with negligible mass. This makes sense intuitively - such equipment is not expected to affect the motion of the structure.

As an example, the formula for the moments given in Eq.85 would simplify to

$$\lambda_{w}^{(100)} = \frac{\pi G_{w}}{16\omega_{d}^{4}} - \frac{\omega_{1}^{4}}{\omega_{1}^{4} - \omega_{2}^{2} - \omega_{1}^{2}} - \frac{(\omega_{1}^{3} - \omega_{2}^{3} - \omega_{2}^{3} - \omega_{2}^{3})}{\xi_{1}\xi_{2}(4\xi_{d}^{2} + \beta^{2})}$$
(101)

Note that the non-interaction expression tends to overestimate the true value for λ , particularly for luned systems due to the absence of the c term in the denominator of Eq.101 (see Fig.3.11).

It is instructive to investigate how small ϵ must be in order to insure that the noninteraction expressions will yield reasonable approximations. The simplest measure of the difference between the non-interaction and exact approaches is the ratio

$$r_{0} = \frac{\lambda_{0}^{(uot)}}{\lambda_{0}}$$
(102)

which, for white-noise input, can be calculated from Eqs.85 and 101

$$r_{0} \approx \frac{(4\xi_{a}^{2} + \beta^{2})\xi_{1}\xi_{2} + \epsilon\xi_{a}^{2}}{(4\xi_{a}^{2} + \beta^{2})\xi_{1}\xi_{2}}$$
(103)

If the error tolerance is e, then the non-interaction results can be used if

$$v_a - 1 < c \tag{104}$$

or equivalently,

$$\epsilon < (4\xi_u^2 + \beta^2) \frac{\xi_1 \xi_2}{\xi_u^2} e$$
(105)

For detuned systems where β is large, the above condition will usually be satisfied and $\lambda_{ij}^{(1-\epsilon)}$ will be a good approximation for λ_{ij} as was stated earlier. However, for tened systems, β is small, and Eq.105 will be satisfied only if ϵ is sufficiently small. A conservative upper bound for ϵ is made by setting $\beta=0$, in which case Eq.105 simplifies to

$$\epsilon < 4\xi_{|\xi|}$$

The relationship between the error in non-interaction analysis and the size of the mass ratio ϵ is illustrated Fig.3.12.

3.6 Summary

3.6.1 Discussion

A thorough study of the 2-DOF ES system was made. The mode shapes and frequencies were derived for tuned and detuned systems and physical interpretations of these quantities were given to elucidate the dynamic characteristics of these systems. Two different methods of analysis were used to find the response of the EC system to stationary input. In the first approach, the frequency response function method was used for funed systems, and second order expressions were derived for the spectral moments λ_{μ} for m=0,1,2. These expressions were subsequently used to find the shape factors ν and δ . In the second approach, the modal decomposition method was used in analyzing both tuned and detuned systems. The results for tuned systems were compared with those from the frequency response function analysis and the two expressions were found to be equivalent. Then the results for defuned systems were combined with those for tuned systems to form a general expression applicable for all systems and to develop a criteria for detuning. Finally, the commonly used decoupled equations of motion for the ES systems were analyzed. It was shown that the results of this analysis can be obtained directly from the results which included interaction by setting the mass ratio ϵ to zero. Also. a criterion for the use of the decoupled approximation was established.

The emphasis of this chapter was to understand the underlying mathematical relationships that exis s in the 2-DOF system in preparation of the study of more general and complex PS systems.

3.6.2 Results

For future reference, the important results derived in this chapter are listed below:

Detuning Criteria

$$\beta' > \frac{1}{e} \left[4 + \frac{\epsilon}{\xi_i \xi_j} \right] \xi_i^2$$
(88)

Non-interaction Criteria

. 73 .

()

Detuned Systems

$$\epsilon < (4\xi_{i}^{2} + \beta^{2}) \frac{\xi_{i}(\xi_{j})}{\xi_{i}^{2}} \epsilon$$
(05)

Tuned Systems

$$\epsilon < \xi \cdot \xi \cdot c \tag{106}$$

Frequencies and Mode Shapes

Tuned Systems

$$\boldsymbol{\phi}^{\prime} = \begin{bmatrix} \alpha \\ 1 \end{bmatrix}$$
$$\boldsymbol{\omega}^{\prime} = \boldsymbol{\omega}_{\alpha} \left[1 + i\boldsymbol{\xi}_{\alpha} + \frac{1}{2}\sqrt{\boldsymbol{\epsilon} + (i\boldsymbol{\xi}_{\alpha} + \boldsymbol{\beta})^{2}} \right], \quad \boldsymbol{\alpha}_{1} = \left[-\boldsymbol{\beta} - i\boldsymbol{\xi}_{\alpha} - \sqrt{\boldsymbol{\epsilon} + (i\boldsymbol{\xi}_{\alpha} + \boldsymbol{\beta})^{2}} \right]$$
(50a)

$$\omega_2 = \omega_0 \left[1 + i\xi_0 - \frac{1}{2} \sqrt{\epsilon + (i\xi_0 - \beta)^2} \right], \quad \alpha_2 = \left[-\beta - i\xi_0 + \sqrt{\epsilon + (i\xi_0 + \beta)^2} \right]$$
(50b)

Detuned Systems

$$\boldsymbol{\phi}_{i} \approx \begin{bmatrix} 1 \\ \omega_{i} \\ \overline{\omega_{i}^{2} - \omega_{i}^{2}} \end{bmatrix} \qquad \omega_{i} \approx \omega_{i} (1 + i\xi_{i})$$

$$(76)$$

$$\begin{bmatrix} \boldsymbol{\epsilon} \omega_{i}^{2} \\ \overline{\omega_{i}^{2} - \omega_{i}^{2}} \end{bmatrix}$$

$$\boldsymbol{\phi}_{1}^{\prime} \approx \begin{bmatrix} \boldsymbol{\omega}_{1}^{\prime} - \boldsymbol{\omega}_{2}^{\prime} \\ 1 \end{bmatrix} \qquad \boldsymbol{\omega}_{2}^{\prime} \approx \boldsymbol{\omega}_{2}(1 + i\boldsymbol{\xi}_{2})$$
(78)

Spectral Moments

Tuned Systems

$$\lambda_{\mu} \approx \frac{\pi \xi_{\mu} G_{ee}}{8\omega_{\mu}^{3-\mu} [\xi_{1}\xi_{2}(4\xi_{\mu}^{2}+\beta^{2})+\epsilon\xi_{\mu}^{2}]}$$
(41e)

Detuned Systems

$$\lambda_{m} = \sum_{i=1}^{2} \sum_{j=1}^{2} \psi_{j} \psi_{j} \lambda_{m_{i}n} \approx \left[\frac{\omega_{1}^{2}}{\omega_{1}^{2} - \omega_{2}^{2}} \right]^{2} \frac{\pi G_{gg}}{4} \left[\frac{\omega_{1}^{m-1}}{\xi_{1}} + \frac{\omega_{2}^{m-1}}{\xi_{2}} \right]$$
(82)

All Systems

$$\lambda_{\mu} = \frac{\pi G_{\nu\nu}}{16\omega_{\mu}^{4}} \frac{\omega_{1}^{4}}{\omega_{1}^{4-\nu}\omega_{2}^{2-\nu}} \frac{\omega_{1}^{5} - \xi_{1} + \omega_{2}^{3-\nu} \xi_{2}}{\xi_{1}\xi_{2}(4\xi_{\mu}^{2} + \beta^{2}) + \epsilon\xi_{\mu}^{2}}$$
(85)

Mean zero crossing rate and shape factors for tuned systems

$$\nu = \frac{\omega_a}{\pi} \left[1 - \frac{\beta \xi_a}{2 \xi_a} \right]$$
(45a)

$$\delta = \left[\frac{\beta^2}{4} \left(1 - \frac{\xi^2}{4\xi^2_0}\right) + \frac{1}{4}\epsilon + \xi_1\xi_2\right]$$

(45b)

| E | Mode | Exact Frequency | | Computed Frequency | | Error |
|-------|------|-----------------|------------|--------------------|------------|--------|
| | | Real Part | Imag. Part | Real Part | Imag. Part | 96 |
| 0.01 | 1 | 0.949 | 0.022 | 0.951 | 0.021 | 0.3 |
| | 2 | 1.050 | 0.037 | 1.050 | 0.038 | 0.1 |
| 0.005 | 1 | 0.963 | 0.019 | 0.964 | 0.018 | 0.2 |
| | 2 | 1.036 | 0.040 | 0.036 | 0.041 | - C. L |
| 0.001 | 1 | 0.976 | 0.012 | 0.976 | 0.012 | 0.0 |
| | 2 | 1.023 | 0.047 | 1.023 | 0.047 | 00 |

Table 3.1 Frequencies for Example System $\omega_n = 1.0$, $\beta = 0.04$. $\xi_n = 0.03$, $\xi_d = 0.04$

Table 3.2 Mode Shape Component α_i of Example System $\omega_a = 1.0$, $\beta = 0.04$, $\xi_a = 0.03$, $\xi_d = 0.04$

| ŧ | Mode | Exact α , | | Computed a | | Error |
|-------|------|------------------|------------|------------|------------|-------|
| | | Real Part | Imag. Part | Reai Part | Imag. Part | %n |
| 0.01 | 1 | 0.061 | -0.024 | 0.058 | -0.022 | 8.9 |
| | 2 | -0.141 | -0.055 | -0.148 | -0.063 | 7.0 |
| 0.005 | 1 | 0.033 | -0.918 | 0.032 | -0.017 | 3.8 |
| | 2 | -0.113 | -0.061 | -0.118 | -0.068 | 6.7 |
| 0.001 | 1 | 0.006 | -0.005 | 0.006 | -0.005 | 0.0 |
| | 2 | -0.086 | -0.074 | -0.088 | -0.080 | 5.6 |



Fig.3.1. Two Degree-of-Freedom Equipment-Structure System







Fig.3.3a. Tuned Complex Frequencies: General Configuration B=0.04, &_=0.04, &_=0.03, w_=1 rad/sec

İ



Fig.3.3b. Tuned Complex Frequencies: Perfect Tuning **B=0.00**, **£**₄=0.04 **£**₄=0.03, **w**₄=1 rad/sec

- 80 -









Fig.3.4. Transfer Function $T_{y}(\omega)$: Tuned System ω_1^{-1} rad/sec. $\omega_2 = 1.04$ rad/sec. $\epsilon = .001$. $\xi_2 = .02$





























Fig.3.6b. Shape Factor b versus Mass Ratio c and/sec. 6a-0.03, 8-0.0







Fig.3.7a. Mode Shape Component α_i : General Configuration $\beta = 0.04$, $\xi_a = 0.04$, $\xi_a = 0.03$, $\omega_a = 1$ rad/sec







Fig.3.7c. Mode Shape Component α_i: Equal Damping β=0.02, ξ₀=0.00, ξ_a=0.03, ω_a=1 rad/sec





- 94 -








ì





ı



Fig.3.12. Non-interaction Error in Spectral Moment λ_0 $\omega_a = 1$ rad/sec, $\xi_1 = 0.05$, $\xi_2 = 0.02$, $\beta = 0$

CHAPTER 4

ANALYSIS OF SINGLE-DEGREE-OF-FREEDOM EQUIPMENT ATTACHED TO MULTI-DEGREE-OF-FREEDOM STPUCTURES

4.1 Introduction

In this chapter, MDOF/SDOF PS systems will be analyzed, where the structure studied in the previous chapter is generalized to one with multiple degrees of freedom and an arbitrary configuration. The equipment remains to have a single degree of freedom however, it can be attached to the structure at more than one node; it may also be attached to the base.

The methods of analysis of the MEOF/SDOF PS system will follow the same approach developed in the previous chapter. It will be shown that the basic enaracteristics and properties of equipment-structure systems, such as tuning, interaction, non-classical damping, and closely-spaced, correlated modes that were found earlier are also present in the MDOF/SDOF system. However, the results in Chapter 3 will be generalized to account for the more complex inter-relationships that are possible between the multiple structure modes and the equipment mode.

It has been recognized that the analysis of certain MDOF/SDOF PS systems can be reduced to the analysis of the 2-DOF equipment-structure system [37, 36]. However, w is shown that this reduction can not be utilized in many MDOF/SDOF systems, therefore more general methods of analysis will be developed.

The analysis will begin with a discussion of tuning and multiple tuning. Then, the frequency response function will be derived for the response of the secondary subsystem. Next, expressions for the mode shapes and frequencies will be derived which are suitable for use in many dynamic analysis techniques such as those described in the Introduction. For further insight to the behavior of the MDOF/SDOF system, these expressions are used directly in the

. 99 .

modal decomposition method developed earlier for finding algebraic formulations for the spectral moments of response to white-noise input. In the final part of this chapter, non-interaction is considered. The results derived earler in the chapter reduce to simple, closed form expressions for all arbitrary configurations of the two subsystems.

4.2 Definitions

4.2.1 Parameters

In this section a set of matrices will be developed to describe the MDOF/SDOF system which will be used in the remainder of this study.

First, the parameters of the primary subsystem will be defined. Let N be the number of degrees of freedom of this subsystem and x_1, \ldots, x_N be the displacement coordinates relative to the base. Then define the primary subsystem matrices

$$\mathbf{K}_{pp} = \text{stiffness matrix } (N \times N) \tag{1a}$$

$$C_{\mu\nu} = \text{damping matrix } (N \times N)$$
 (1b)

$$\mathbf{M}_{\mu\nu} = \text{mass matrix } (N \times N) \tag{1c}$$

which are associated with the displacements x_{i} .

It is assumed that the primary subsystem in itself is classically damped. Therefore by standard eigenvalue analysis the following properties can be derived:

$$\omega_{\mu}$$
 = natural frequencies (2a)

$$\xi_p \simeq \text{damping ratios}$$
 (2b)
 $\Phi = \text{mude shapes}$ (2c)

. . . .

$$\boldsymbol{\phi}_{p_{0}} = \text{mode shapes} \tag{2c}$$

In general, it is not required to obtain all N mode shapes and frequencies; it is assumed that the above parameters are obtained for $i = 1, ..., n \leq N$. From the orthogonal properties of the mode shapes, the mass, stiffness, and damping matrices can be diagonalized using the $(N \times n)$ transformation matrix

$$\boldsymbol{\Phi}_{p} = \left[\boldsymbol{\phi}_{p1} \cdots \boldsymbol{\phi}_{pn} \right] \tag{3}$$

with the following result

$$\mathbf{\Phi}_{p}^{T}\mathbf{M}_{pp}\mathbf{\Phi}_{p} = \operatorname{diag}\{m_{p1} \cdots m_{pn}\}$$
(4a)

$$\mathbf{\Phi}_{\nu}^{T} \mathbf{C}_{\nu\nu} \mathbf{\Phi}_{\nu} = \operatorname{diag} \left[2m_{\nu} (\boldsymbol{\omega}_{\nu}) \xi_{\nu} + \cdots + 2m_{\nu} (\boldsymbol{\omega}_{\nu}) \xi_{\nu} \right]$$

$$(4b)$$

$$\Phi_{p}^{T} \mathbf{K}_{pp} \Phi_{p} = \operatorname{diag} \{ m_{p} | \omega_{p} \rangle + (m_{p} | \omega_{p}) \}$$

$$(4c)$$

For the ringle DOF secondary subsystem, let m_{β} , ξ_{β} , and ω_{β} be the mass, damping ratio, and natural frequency of this subsystem, respectively. Then the 1×1 mass, damping, and stiffness matrices are simply.

$$\mathbf{M}_{ij} = [m_{ij}]_{ij} \mathbf{C}_{ij} = [2m_{ij}\omega_{ij}\xi_{ij}]_{ij} \text{ and } \mathbf{K}_{ij} = [m_{ij}\omega_{ij}]_{ij},$$
(5)

The parameters of the two subsystems will now be used to describe the properties of the combined N+1 DOF PS system. The first N DOF will correspond to the primary subsystem and the N+1 DOF will correspond to the secondary subsystem. Thus, the displacements are defined by the N+1 vector.

$$\mathbf{x}_{0} = \{\mathbf{x}_{1} \mid \dots \mid \mathbf{x}_{N} \mid \mathbf{x}_{N+1}\}^{T}$$
(6)

where x_{N+1} is the displacement of the secondary subsystem relative to the base of the combined system. The $N+1 \times N+1$ mass, damping, and stiffness matrices are

$$\mathbf{M}_{sys} = \begin{bmatrix} \mathbf{M}_{ys} & \mathbf{M}_{ys} \\ \mathbf{M}_{sys} & \mathbf{M} \end{bmatrix} \quad \mathbf{C}_{sys} = \begin{bmatrix} \mathbf{C}_{yy} & \mathbf{C}_{sys} \\ \mathbf{C}_{yz} & \mathbf{C}_{sys} \end{bmatrix} \quad \mathbf{K}_{sys} = \begin{bmatrix} \mathbf{K}_{yy} & \mathbf{K}_{yss} \\ \mathbf{K}_{sys} & \mathbf{K}_{sys} \end{bmatrix}$$
(7)

where $\mathbf{C}_{qr} = \mathbf{C}_{qr}^{T}$, $\mathbf{M}_{qr}^{T} = \mathbf{M}_{pr}^{T}$, and $\mathbf{K}_{pr} = \mathbf{K}_{pr}^{T}$ are $1 \times N$ matrices representing the physical coupling between the primary and secondary subsystems. The N+1 vector \mathbf{r}_{0} is defined to be the influence vector coupling the base input to the displacements \mathbf{x}_{0r} . From these matrices, the equations of motion for the combined system is given by

$$\mathbf{M}_{i_{1}} \, \ddot{\mathbf{x}}_{0} + \mathbf{C}_{s_{1}} \, \dot{\mathbf{x}}_{0} + \mathbf{K}_{s_{1}} \, \mathbf{x}_{0} = -\mathbf{M}_{s_{1}s_{1}} \mathbf{r}_{0} \ddot{\mathbf{y}}_{s_{1}}(\mathbf{y})$$
(8)

where $\tilde{x}_i(t)$ is the base input acceleration.

To be precise, the primary subsystem matrices \mathbf{M}_{pp} , \mathbf{C}_{ppr} and \mathbf{K}_{pp} should be replaced by more complicated matrices to account for the small added mass, damping, and stiffness terms arising from the influence of the secondary subsystem. However, these terms are of secondorder magnitude in relation to the original primary subsystem matrix elements. Therefore, the use of the original primary matrices in the above equations is consistent with the lowest and first-order perturbation analysis that will be performed. The damping and stiffness matrices C_{yr} and K_{yp} have a simple physical interpretation: For secondary subsystems with a single attachment point at, say, the *i*-th DOF of the primary subsystem, these matrices are given by

$$\mathbf{C}_{\mathbf{y}_{1}} = \begin{bmatrix} 0 & \cdots & 0 - 2m_{\mathbf{y}_{1}}\boldsymbol{\omega}_{\mathbf{y}_{1}}\boldsymbol{\xi}_{\mathbf{y}_{1}} & 0 & \cdots & 0 \end{bmatrix}$$
(10a)

$$\mathbf{K}_{st} = \begin{bmatrix} \mathbf{0} & \cdots & \mathbf{0} - \mathbf{m}_{st} \mathbf{\omega}_{st}^T & \mathbf{0} & \cdots & \mathbf{0} \end{bmatrix}$$
(10b)

where the non-zero terms are at the *I*-th coordinate. For multiply attached secondary subsystems, stiffness and damping terms would be found in the coordinates corresponding to each attachment point. On the other hand, the mass matrix \mathbf{M}_{sp} is assumed to be zero since PS systems are usually modeled with lumped masses.

In general, $\mathbb{C}_{v,v}$ and $\mathbb{K}_{v,v}$ are full matrices and are difficult to utilize for analysis. For this reason dynamic analysis of the combined PS system has been avoided in the past. However, from the above discussion it is clear that these matrices can be reduced using the transformation matrix

$$\boldsymbol{\Phi} = \begin{bmatrix} \boldsymbol{\Phi}_{\mu} & \mathbf{0} \\ \mathbf{0} & \boldsymbol{\Phi}_{\lambda} \end{bmatrix}$$
(11a)

where Φ_{n} =[1] is the 1×1 identity matrix. Using this transformation a new coordinate vector z_{n} given by the n+1 modal coordinates

$$\mathbf{z} = [z_{p1}, \dots, z_{pn}, z_{s1}]$$
(11b)

is obtained through the relation $\Phi z - x_0$. The mass, damping, and stiffness matrices in terms of these modal coordinates are, to lowest order,

$$\mathbf{M} = \mathbf{\Phi}^T \mathbf{M}_{xx} \mathbf{\Phi} = \text{diag} \begin{bmatrix} m_{p1} \cdots m_{pn} & m_{xl} \end{bmatrix}$$
(12a)
$$\begin{bmatrix} 2m_{n1} \mathbf{\omega}_{n1} \mathbf{\varepsilon}_{n2} & 0 & c_{xx} \end{bmatrix}$$

$$\mathbf{C} = \mathbf{\Phi}^{T} \mathbf{C}_{xx} \mathbf{\Phi} = \begin{bmatrix} 0 & p_{1} e_{p_{1}} e_{p_{1}} e_{p_{1}} & \cdots & \cdots & \cdots \\ 0 & 2m_{p_{m}} \omega_{p_{m}} \xi_{p_{m}} & c_{n1} \\ c_{11} & c_{n1} & 2m_{y_{1}} \omega_{y_{1}} \xi_{y_{1}} \end{bmatrix}$$
(12b)

$$\mathbf{K} = \mathbf{\Phi}^T \mathbf{K}_{\text{MN}} \mathbf{\Phi} = \begin{bmatrix} m_{p1} \boldsymbol{\omega}_{p1}^2 & \mathbf{U} & \boldsymbol{k}_{11} \\ \vdots & \vdots & \vdots \\ \mathbf{U} & \vdots & \vdots \\ \mathbf{0} & m_{pn} \boldsymbol{\omega}_{pn}^2 & \boldsymbol{k}_{n1} \\ \boldsymbol{k}_{11} & \boldsymbol{k}_{n1} & m_{N} \boldsymbol{\omega}_{N1}^2 \end{bmatrix}$$
(12c)

where

$$|\psi_1 = \phi_1' C_{\mu_1} - h_1 = \phi_2' K_{\mu_2}$$
(12d)

Note the simplicity of the form of the above matrices as compared with the original system matrices. Eqs 7. The original equations of motion that were given in Eq.8 for the nodal coordinates \mathbf{x}_{i} has the following form for the modal coordinates \mathbf{z}_{i}

$$\mathbf{M}\mathbf{\ddot{z}} + \mathbf{C}\mathbf{\ddot{z}} + \mathbf{K}\mathbf{z} = -\mathbf{M}\mathbf{r}\mathbf{\dot{x}}_{1}(t) \tag{13}$$

where the vector of modal participation factors \mathbf{r} are given in terms of \mathbf{M} — and \mathbf{r}_0 by the standard formulation [10].

$$\mathbf{r} = \mathbf{M} - \mathbf{\Phi}^T \mathbf{M}_{\rm ext} \mathbf{r}_{\rm ext} \tag{14}$$

It is worthwhile to examine the terms in Eqs.12d more closely. For secondary subsystems with only a single attachment point at the 1-th DOF, these equations reduce to

$$k = -m \omega \{ \mathbf{a}_{n} \} \tag{15a}$$

$$c_{\rm c} = 2m_{\rm c} \alpha_{\rm c} \left\{ \xi_{\rm c} \left[\phi_{\rm c} \right] \right\}$$
(15b)

where $|\phi_{ij}|$ denotes the *t*-th coordinate of the vector ϕ_{ij} . For more general support configurations, similar expressions can be obtained. Let ζ be the ratio

$$\zeta_{\pm} = -\frac{k_{\pm}}{m_{\pm}\omega_{\pm}} \tag{16}$$

Then k and c_{\pm} can be written as

$$k = -m_{\chi}\omega_{\chi}\zeta \quad \psi_{\chi} \approx -2m_{\chi}\omega_{\chi}\xi_{\chi}\zeta_{\chi}$$
(17)

where the first equation is exact, by the definition of ζ_{\pm} , and the second is an approximation. The approximation is justified for obtaining tow-order results for two reasons: (1) it is exact in the important case of single attachment point, as shown in Eq.15b, (2) this relatively small term has been neglected altogether in previous research work without much loss of accuracy.

Before going further, a physical interpretation of the term ζ will be investigated. For systems with a single attachment point, $\zeta_{\pm} = [\phi_{\pm}]_{i}$ is the displacement that the mass of the secondary subsystem experiences when the primary subsystem is statically displaced into its mode shape $\phi_{\mu\nu}$. For multiply supported secondary subsystems, the meaning for ζ_{\pm} remains the same. This is clear from the static equilibrium equation $K_{z=0}$, which can be rewritten in terms of the equipment DOF

$$\sum_{i=1}^{n} k_{ii} z^{-iz} = \omega^2 m_{ii} z_{i+1}$$
(18)

Given a unit displacement of the *i*-th primary mode, i.e., z=1 and z=0 for $i \neq i$, the corresponding displacement for the equipment is

$$z_{m} = -\frac{k}{\omega m} = \zeta_{m} \tag{19}$$

As an example, consider the 3-DOF system in Fig.4.1. By following the matrix analysis outlined in this section, it can be shown that

$$\boldsymbol{\phi}_{p1} = \begin{bmatrix} 1\\ .5 \end{bmatrix} \quad \boldsymbol{\phi}_{p2} = \begin{bmatrix} 1\\ .-5 \end{bmatrix} \quad \mathbf{K}_{sp} = \begin{bmatrix} -\frac{1}{2}\boldsymbol{\omega}_{s1}^2 \boldsymbol{m}_{s1} & -\frac{1}{2}\boldsymbol{\omega}_{s2}^2 \boldsymbol{m}_{s1} \end{bmatrix}$$
(20a)

from which it follows, by definition, that

$$\zeta_{41} = -.75$$
 and $\zeta_{21} = -.25$ (20b)

The deflections of the primary subsystem into its mode shapes are shown, and it is observed that the corresponding static displacement of the equipment are given exactly by ζ_{ij} .

Before continuing, it is useful to introduce notation for mass ratios. Let ϵ be the ratio

$$\boldsymbol{\epsilon}_{\perp} = \frac{m_{\perp}}{m_{p}} \tag{21}$$

Also, let γ be the effective mass ratio

$$\mathbf{y}_{1} = \zeta \sum_{n=1}^{\infty} \frac{m_{n}}{m_{n}} \tag{22}$$

which is a generalization of the effective mass ratio defined by Der Kiuregnian, et al. [18] for the case of a single attachment point, i.e., where $\zeta_{i,i} = \{\phi_{i,i}\}_{i=1}^{n}$

4.2.2 Classifications for MDOF/SDOF Systems

The 2-DOF PS system has only two simple classifications: tuned and detuned. However, the MDOF/SDOF PS system has a larger number of possible relationships between the modes of the structure and the mode of the equipment, therefore more classifications are necessary for an organized, well-defined analysis of this system.

First, a definition of tuning for MDOF/SDOF systems will be presented. For the 2-DOF

system the structure mode was found to be tuned to the equipment mode if

$$\beta^{2} = \left[\frac{\omega_{j,1} - \omega_{s,1}}{\omega_{a}}\right]^{2} \leq \frac{\xi_{a}^{2}}{e} \left[4 + \frac{\epsilon}{\xi_{s,1}\xi_{j+1}}\right]$$
(23)

where ω_{α} is the average of $\omega_{r,i}$ and $\omega_{s,i}$. This definition can be extended to MDOF/SDOF systems: mode *i* of the structure is said to be tuned to the equipment mode if

$$\left(\frac{\omega_{p}-\omega_{s,1}}{\omega_{a}}\right)^{2} < \frac{\xi_{a}^{2}}{c} \left(4 + \frac{\gamma_{\perp}}{\xi_{s,1}\xi_{p}}\right)$$
(24)

where ω_{μ} is the average of ω_{x1} and ω_{μ} and γ_{\pm} is the effective mass ratio defined earlier.

The above definition is used to define several types of tuning classifications. Let I_i and I_{ii} denote the sets of structure modes tuned and defuned from the equipment, respectively, i.e.

$$I_{i} = \left\{ i : \boldsymbol{\omega}_{p} \text{ is tuned to } \boldsymbol{\omega}_{s,1} \right\}$$
(25a)

$$I_{\mu} = \left\{ i : \omega_{\mu} \text{ is defuned from } \omega_{\mu\nu} \right\}$$
(25b)

For the 2-DOF system, n=1 and only two classification are possible:

$$I_{t} = \{1\}, I_{a} = \{\} \text{ fo: tuned systems}$$
(26a)

$$I_{i} = \{ \{ I_{i} = \{ I_{i} \} \text{ for detuned systems}$$
 (26b)

For MDOF structures where n > 1, several relationships must be considered. If all primary modes are defuned from the equipment, the system is totally defuned. Otherwise the system is singly or multiply tuned, according to the number of primary modes tuned to the secondary subsystem. Using the above notation,

$$I_i = \{\}, I_d = \{1, \cdots, n\} \text{ for defuned systems}$$
(27a)

$$l_i = \{l_i\}$$
 for singly tuned systems (27b)

$$I_i = \{k, k+1, \cdots, k+l-1\} \text{ for multiply tuned systems}$$
(27c)

In the last case, l primary modes, beginning with the k-th mode, are tuned to the equipment.

4.2.3 Example System

To illustrate the major characteristics of MDOF/SDOF systems and check the accuracy of the formulations derived in this chapter, the example system shown in Fig.4.2 is used. The primary subsystem is composed of two parts, a SDOF oscillator and a 2-DOF subsystem, both attached to a common base. The SDOF secondary subsystem is attached to both parts of the primary subsystem as well as the base.

The dynamic properties of the subsystems are described in Tables 1 and 2 and are chosen so that the combined system would exhibit important characteristics found in general MDOF/SDOF systems. For instance, the frequency of the equipment is a variable parameter, which allows for an investigation of tuning. For $\omega = .38$ rad/sec, the system is singly tuned as shown in Fig 4 3a, and for $\omega = 1.0$ rad/sec, the system becomes multiply tuned as shown in Fig 4 3b. The mass ratio, ϵ_i is also chosen to be a variable parameter in order to study the influence of interaction. Finally, the damping ratio of the equipment is unequal to the damping ratios of the primary subsystem, thus the combined system is, in general, non-proportionally damped.

The response quantity that will be investigated is the relative displacement between the mass of the secondary subsystem and the upper mass of the primary subsystem 2.

4.3 Frequency Response Function Approach

4.3.1 Introduction

The complex frequency response matrix will be derived for general MDOF/SDOF PS systems. The transfer function is obtained which is used to find the power spectral density function and its moments for the response of the system to random excitation. The exact form of the transfer function is complicated; however, perturbation methods are used to reduce the expressions to a relatively simple closed form rational polynomial.

4.3.2 The Complex Frequency Response Matrix

The complex frequency response matrix for the system described by Eq.3.15 is found by substituting the expressions Eqs.12a-c for the matrices M, C, and K into the definition Eq.3.15. The result is

$$\mathbf{H}(\boldsymbol{\omega}) = \begin{bmatrix} G_{1}(\boldsymbol{\omega}) & \cdots & 0 & f_{11}(\boldsymbol{\omega}) \\ \cdots & \cdots & \cdots & \cdots \\ 0 & \cdots & G_{n}(\boldsymbol{\omega}) & f_{n1}(\boldsymbol{\omega}) \\ f_{11}(\boldsymbol{\omega}) & \cdots & f_{n1}(\boldsymbol{\omega}) & g_{1}(\boldsymbol{\omega}) \end{bmatrix}$$
(28a)

where

1

$$g_1(\omega) = m_{11}(-\omega^2 + 2i\omega_{11}\xi_{11}\omega + \omega_{11}^2)$$
(28b)

$$G(\omega) = m_{\nu} \left(-\omega^2 + 2i\omega_{\nu}\zeta_{\mu}\omega + \omega_{\nu}^2 \right)$$
(28c)

$$f_{1}(\omega) = -\zeta_{1}m_{s1}(2i\omega_{s1}\xi_{s1}\omega + \omega_{s1}^{2})$$
(28d)

for $i=1, \dots, n$. As in the analysis of the 2-DOF system, $G_i(\omega)$ and $g_1(\omega)$ are the reciprocals of the complex frequency response functions for the *i*-th mode of the primary subsystem and the secondary subsystem, respectively. The function $f_{-1}(\omega)$ represents the coupling between these two modes. Due to the simple form of the matrix in Eq.2ⁿa, a closed form expression can be found for the inverse

$$\mathbf{H}(\boldsymbol{\omega}) = \frac{\prod_{i=1}^{n} G_{i}}{d(\boldsymbol{\omega})} \begin{bmatrix} \frac{1}{G_{1}} \left[g_{1} - \sum_{i \neq i} \frac{f_{i}^{2}}{G_{i}} \right] & \text{(sym)} \\ \frac{f_{11}f_{11}}{G_{1}G_{2}} & \frac{1}{G_{2}} \left[g_{1} - \sum_{i \neq i} \frac{f_{i}^{2}}{G_{i}} \right] & \text{(sym)} \\ \frac{f_{11}}{G_{1}G_{2}} & \frac{1}{G_{2}} \left[g_{1} - \sum_{i \neq i} \frac{f_{i}^{2}}{G_{i}} \right] & \text{(28e)} \\ \frac{f_{11}f_{n1}}{G_{1}G_{n}} & \frac{f_{21}f_{n1}}{G_{2}G_{n}} & \cdots & \frac{1}{G_{n}} \left[g_{1} - \sum_{i \neq i} \frac{f_{i}^{2}}{G_{i}} \right] \\ - \frac{f_{11}}{G_{1}} & - \frac{f_{21}}{G_{1}} & \cdots & - \frac{f_{n1}}{G_{n}} \end{bmatrix} 1 \end{bmatrix}$$

where $d(\omega)$ is the characteristic polynomial of the system

$$d(\omega) = \iint_{i=1}^{n} G_i \left[g_1 \cdots \sum_{i=1}^{n} \frac{f_{i+1}^{(i)}}{G_i} \right]$$
(28f)

The frequencies of the system are the roots of the 2(n+1) order equation

$$d(\omega) = 0 \tag{29}$$

Unlike the study of 2-DOF systems, general closed form solutions for the above equation de not exist.

Most response variables can be expressed as linear combinations of the original DOF $\mathbf{x}(t)$. Following the notation of chapter 3, a response variable y(t) can be written

$$\mathbf{y}(t) = \mathbf{q}_t^T \mathbf{x}(t) \tag{30}$$

For example, if the response of interest is the displacement of the equipment relative to the attachment point, the vector \mathbf{q}_0 is given by

$$\mathbf{q}_0 = [0 \cdots 0 - 1 0 \cdots 0 1]^T$$

where the position of the -1 corresponds to the attachment DOF. Transforming to the modal coordinates z(t), the expressions for y(t) becomes

$$y = \mathbf{q}_0' \mathbf{x} = \mathbf{q}_0' \mathbf{\Phi} \mathbf{z} = \mathbf{q}' \mathbf{z}$$
(3))

where $\mathbf{q} = \mathbf{\Phi}^{T} \mathbf{q}_{0}$

The formula for the frequency response function $H_i(\omega)$ for the response y(i) is (see Eq.3.33)

$$H_{\alpha}(\omega) = -q^{T} H(\omega) Mr$$
(32)

By substituting Eq.28e into the above, the exact solution for $H_i(\omega)$ can be obtained; however, this expression is rather cumbersome for frequency response analysis. By using perturbation methods, it will be shown that the dominant terms of the frequency response matrix are along the diagonal and the last row.

The order relationships between the elements of the complex frequency response matrix $\mathbf{H}(\omega)$ are examined for various values of the frequency ω . For values of ω in a neighborhood of each of the detuned primary frequencies $\omega_{\mu\nu}$ for $j \in I_d$, the order of magnitude of the polynomials $f_{\pm}(\omega)$, $g_{\pm}(\omega)$, and $C_{\pm}(\omega)$ are estimated by setting $\omega = \omega_{\mu\nu}$:

$$g_1(\omega_{p_1}) = O(e^2) - f_{c1}(\omega_{p_2}) = O(e^2)$$
 (33a)

$$G_i(\omega_{\mu i}) = \begin{cases} O(e) & \text{for } i = j \\ O(1) & \text{for } i \neq j \end{cases}$$
(*3b)

where e represents the order of the problem. Substituting the above into Eq.28e,

$$H(\omega) = \frac{1}{O(e^2)} \begin{vmatrix} O(e^2) & (sym) \\ O(e^3) & O(e^2) \\ \cdots & \cdots \\ O(e^3) & O(e^3) \\ O(e^2) & O(e^2) \\$$

where only the *j*-th terms of the diagonal and of the last row are of order O(e).

For values of ω that are not in the neighborhood of any of the primary or secondary frequencies,

$$g_1(\omega) = O(e^2) \quad f_{11}(\omega) = O(e^2) \quad G_1(\omega) = O(1)$$
 (34a)

and $H(\omega)$ is of the same form as Eq.33c except that all of the terms of the diagonal and the last row are of order $O(e^2)$.

Next, consider ω near the secondary subsystem frequency $\omega_{i,i}$ for defined systems. In this case, $g_{\perp}(\omega) = O(e^{i})$, $f_{\perp}(\omega)$ and $G_{\perp}(\omega)$ are as in Eq.34a, ind the order relationships for $H(\omega)$ are

$$\mathbf{H}(\boldsymbol{\omega}) = \frac{1}{O(e^3)} \begin{vmatrix} O(e^3) & & (sym) \\ O(e^4) & O(e^3) & & \\ \cdots & \cdots & \cdots & \\ O(e^4) & O(e^4) & \cdots & O(e^3) \\ O(e^2) & O(e^2) & \cdots & O(e^2) & O(1) \end{vmatrix}$$
(34b)

Finally, consider tuned systems, where for the sake of notational convenience it is assumed that the first *l* primary subsystem modes (not necessarily having the lowest *l* frequencies) are the tuned modes, i.e. $I_l = \{1, \ldots, l\}$ and $I_d = \{l+1, \ldots, n\}$. For values of ω in a neighborhood of the equipment frequency ω_{s1}

$$g_{1}(\omega_{s1}) = O(e^{s}) \quad f_{i}(\omega_{s1}) = O(e^{2})$$
(35a)
$$G_{i}(\omega_{s1}) = \begin{cases} O(e) \text{ for } 1 \le i \le l \\ O(1, \text{ for } l \le i \le n \end{cases}$$
(35b)

and

$$H(\omega) = \frac{1}{O(e^3)} \begin{vmatrix} O(e^2) & (sym) \\ O(e^2) & O(e^3) \\ \cdots & \cdots \\ O(e^2) & O(e^2) & \cdots \\ O(e^2) & O(e^2) & \cdots \\ O(e^2) & O(e^2) & 0(1) \end{vmatrix}$$
(35c)

where, in the last row, the first *l* terms are of order O(e) and the *l*+1-st through the *n*-th terms are of order $O(e^2)$.

In the calculation of $H_i(\omega)$, the matrix $H(\omega)$ is post multiplied by Mr. In this process, the last column of the matrix is multiplied by a term of order $O(v^2)$ smaller than the multiplier for the first *i* columns. It follows from the order relations above that the reduced expression for $H_i(\omega)$ is obtained from the diagonal and last row of $H(\omega)$

$$H_{i}(\omega) \approx \frac{g_{1}(\omega) \prod_{i=1}^{n} G_{i}(\omega)}{d(\omega)} \left[\sum_{i=1}^{n} \frac{q_{n+1}m_{p_{i}}r_{i}f_{i+1}(\omega)}{g_{1}(\omega)G_{i}(\omega)} - \frac{q_{n+1}m_{i+1}r_{n+1}}{g_{1}(\omega)} - \sum_{i=1}^{n} \frac{q_{i}m_{p_{i}}r_{i}}{G_{i}(\omega)} \right]$$
(36)

where r, is the r-th component of the influence vector r and q_i is the r-th component of q.

By giving similar considerations to the characteristic polynomial $d(\omega)$, it can be shown that only the terms corresponding to the tuned frequencies are needed for the summation in the expression in Eq.28f. Thus, for the multiply tuned case,

$$a(\omega) = \prod_{i=1}^{n} G_{i}(\omega) \left[g_{1}(\omega) - \sum_{i,j \in G} \frac{f_{ij}^{(1)}(\omega)}{G_{j}(\omega)} \right]$$
(37)

The above constitutes a considerable simplification for the expression for the frequency response function $H_{c}(\omega)$ and by observing the order relationships and keeping all dominant terms, first-order accuracy is maintained. The transfer function $T_{c_{1}}(\omega)$ and spectral density function $G_{c_{1}}(\omega)$ are obtained using Eqs.3.34a,b. From the order relationships — above, it can be shown that for tuned systems the transfer function has a peak of order $O(e^{-4})$ at $\omega = \omega_{c_{1}}$ as for the 2-DOF system. There are also smaller peaks of order $O(e^{-1})$ at the defuned frequencies $\omega_{p_{1}}$ for $i \in I_{d_{1}}$.

A plot of $T_{11}(\omega)$ for the example system in Fig.4.2 is shown in Figs.4.4a b. In Fig.4.4a, the equipment frequency, $\omega_{x1} = .38$ rad/sec, is tuned to the first primary frequency, $\omega_{p1} = .374$ rad/sec, thus a sharp peak is found for $\omega = \omega_{x1}$. There is also a small peak corresponding to the detuned primary modes near $\omega = 1.00$ rad/sec, however it is considerably smaller than the first peak and is not visible on the plot. In Fig.4.4b, the equipment frequency, $\omega_{v1} = 1.00$ rad/sec, is multiply tuned to two primary frequencies, $\omega_{p2} = .98$ rad/sec and $\omega_{p3} = 1.02$ rad/sec, and a peak is found for $\omega = \omega_{x1}$. For this case, a smaller peak is also found for the detuned primary mode at $\omega = \omega_{p1}$.

4.3.3 Spectral Moments of Response

The spectral moments λ_{μ} of the response variable y(t) are given by the integral Eq.2.1

$$\lambda_{\mu\nu} = \int_{0}^{0} \omega^{\mu\nu} G_{\nu\nu} d\omega$$
 (38)

For general forms of the input power spectral density this integral can be obtained by numerical integration. This method of finding the moments was used to find the response of the example

system in Fig.4.2 to white-noise base excitation. The results obtained from the approximate and exact forms of the complex frequency response function $H_{1}(\omega)$ derived above are plotted as a function of the equipment frequency, ω_{x1x} and compared in Figs.4.5a-c. The peaks correspond to the responses for singly tuned systems ($\omega_{x1} \approx \omega_{p1}$) and multiply tuned systems ($\omega_{x1} \approx \omega_{p2}, \omega_{p1}$). The difference between approximate and exact results are slight, illustrating the accuracy of the perturbation methods.

Also, an integration formula which is a generalization of Eq.3.40a-c can be used to find λ_{μ} for m=0 and 2 for a white noise input. However, this formula does not yield simple closed form solutions as in the previous chapter and the method is difficult to implement into a computer.

4.4 Modal Decomposition Approach

4.4 1 Mode Shapes and Frequencies

First, mode shapes and frequencies will be derived for each set of modes classified earlier in this chapter. An analysis of the system requires the solutions to an $n+1 \times n+1$ order complex eigenvalue problem, however, this problem is reduced significantly using perturbation principles. For most modes, simple closed form solutions exist for the eigenvalues and eigenvectors; only the multiply tuned case requires the solution of an eigenvalue problem. In the latter case, it *t* is the number of modes tuned to the equipment, the size of the problem is $l+1 \times l+1$ which, in general, is far smaller than the original $n+1 \times n+1$ problem. Thus, given any **MDOF/SDOF PS** system, the expressions for the mode shapes and frequencies can be evaluated numerically, and the results used directly in the modal decomposition method developed in Chapter 2, or in any other suitable dynamic analysis method.

In this chapter, the algebraic form of the mode shapes and frequencies are used in the modal decomposition method to form general approximate expressions for the spectral moments for response to white noise input. Closed form solutions are obtained for all but the multiply tuned case. The results are less accurate than the numerical application of the method, and is not recommended for computer applications. However, the closed form expressions are useful in gaining further insight into the behavior of PS systems. For computer applications, the direct modal decomposition approach mentioned above is more appropriate.

4.4.1.1 Primary Detuned Modes

The modes shapes and frequencies detuned from the secondary subsystem are found by solving the eigenvalue problem

$$\mathbf{\Gamma}(\boldsymbol{\omega})\boldsymbol{\phi} = 0 \tag{39a}$$

where

$$\mathbf{\Gamma}(\boldsymbol{\omega}^{*}) = \begin{bmatrix} G_{\pm}(\boldsymbol{\omega}) & \cdots & 0 & f_{\pm\pm}(\boldsymbol{\omega}) \\ \cdots & \cdots & \cdots & \cdots \\ 0 & \cdots & G_{\pm}(\boldsymbol{\omega}) & f_{\pm\pm}(\boldsymbol{\omega}) \\ f_{\pm\pm}(\boldsymbol{\omega}) & \cdots & f_{\pm\pm}(\boldsymbol{\omega}) & g_{\pm}(\boldsymbol{\omega}) \end{bmatrix}$$
(39b)

This problem is similar to the eigenvalue problem in Section 3.4.3.2 and the method of analysis developed in detail in that section is used here.

For notational convenience, assume that primary mode 1 is detuned, i.e. $1 \in I_{cl}$. Also, assume that mode 1 is widely spaced from the other primary subsystem modes. (This analysis will be repeated without this assumption in a later section.) The initial approximation for the mode shape and frequency are similar to those given in Eqs.3.68

$$\phi_{1}^{(0)} = [1 \ 0 \ \cdots \ 0]'$$
(40a)
$$\omega_{1}^{(0)} = \omega_{e1}(\sqrt{1 - \xi_{e1}^{2}} + i\xi_{e1})$$
(40b)

By applying the same error analysis developed in Eqs.3.69-71a and using the more general form for the matrix $\Gamma(\omega)$, a higher order approximation for ϕ_1 can be found. The resulting mode shape corresponding to Eq.3.71a is

$$\boldsymbol{\phi}_{1}^{(1)} = \left[1 \frac{f_{11}f_{21}}{g_{1}G_{2}} \cdots \frac{f_{1}(f_{n})}{g_{1}G_{n}} - \frac{f_{11}}{g_{1}}\right]^{T}$$
(41)

where the polynomials $f_{(1)}$, g_{1} , and G_{i} are evaluated at $\omega = \omega_{1}^{(1)}$. Note that if the *i*-th primary mode is closely spaced to mode 1, $G_{i}(\omega_{1}^{(1)})$ would be relatively small and the corresponding coordinate in $\phi_{1}^{(1)}$ would be relatively large, invalidating the perturbation analysis used to derive

 $\phi_1^{(1)}$. Therefore, the widely spaced assumption for the defined mode is necessary in this analysis.

As for the 2-DOF system, the low-order approximations for the mode shape and frequency are found from $\omega_1^{(0)}$ and $\phi_1^{(1)}$:

$$\boldsymbol{\phi}_{1} \approx \left[1 \ 0 \ \cdots \ 0 \ \frac{\boldsymbol{\zeta}_{11} \boldsymbol{\omega}_{11}^{2}}{\boldsymbol{\omega}_{11}^{2} - \boldsymbol{\omega}_{p1}^{2}}\right]^{T}$$
(42a)

$$\boldsymbol{\omega}_{1} = \boldsymbol{\omega}_{p1} (1 + i\boldsymbol{\xi}_{p1}) \tag{42b}$$

Note the similarity between the above results and the results for the 2-DOF system in Eq.3.76. The only significant difference is in the appearance of the coefficient ζ_{c1} . This coefficient reflects the more general form of the physical coupling between the primary and secondary subsystems in MDOF/SDOF systems.

4.4.1.2 Secondary Detuned Mode

If the secondary subsystem mode is defined from the primary subsystem modes, i.e. if I_{n-1} , then the mode shape and frequency for mode n+1 which is associated with the secondary subsystem can be derived in a manner parallel to the preceding analysis. The initial approximations for ϕ_{n+1} and ω_{n+1} are

$$\boldsymbol{\phi}_{n+1}^{(0)} = [0 \ 0 \ \cdots \ 1]^{T}$$
(43a)

$$\omega_{n+1}^{(0)} = \omega_{s1}(\sqrt{1 - \xi_{s1}^{s}} + i\xi_{s1})$$
(43b)

The higher order expression for ϕ_{n+1} corresponding to Eq.41 is

$$\boldsymbol{\phi}_{n+1}^{(1)} = \left[-\frac{f_{11}}{G_1} \cdots - \frac{f_{n1}}{G_n} \right]^l$$
(44)

where the polynomials are evaluated at $\omega = \omega_{n+1}^{(0)}$. The final, low-order results are

$$\boldsymbol{\phi}_{n+1} = \left[\zeta_{11} \boldsymbol{\epsilon}_{11} \frac{\boldsymbol{\omega}_{11}^2}{\boldsymbol{\omega}_{n1}^2 - \boldsymbol{\omega}_{n1}^2} \cdots \zeta_{n1} \boldsymbol{\epsilon}_{n1} \frac{\boldsymbol{\omega}_{n1}^2}{\boldsymbol{\omega}_{nn}^2 - \boldsymbol{\omega}_{n1}^2} \mathbf{1} \right]^T$$
(45a)

$$\omega_{n+1} = \omega_{x1}(1 + i\xi_{x1})$$
(45b)

where ϵ_{c1} are the mass ratios defined in Eq.21. Note the similarities between the above results and the results for the 2-DOF system in Eq.3.78. Again, the only difference is in the appearance of the coupling coefficient ζ_{c1} .

4.4.1.3 Singly Tuned Modes

The eigenvalue analysis of singly tuned systems is similar to that of tuned 2-DOF systems presented in section 3.4.2.1. For notational convenience, assume that primary subsystem mode 1 is tuned to the secondary subsystem, i.e. $I_i = \{1, 1\}, I_{ij} = \{2, \dots, n\}$.

The initial approximations for the tuned modes are obtained by neglecting the effect of the detuned modes and considering only the first and n+1 coordinates of the eigenvalue problem in Eqs.39a,b. The resulting problem is essentially the same as that solved for the 2-DOF system in Eq.3.48. The parameters $\beta_{\lambda} \xi_{\alpha\lambda} \xi_{\beta\lambda}$ and ω_{α} which were defined in Eqs.3.1a,b,d,e for the 2-DOF system are redefined here by setting $\xi_1 = \xi_{\mu 1}$, $\xi_2 = \xi_{\lambda 1}$, $\omega_1 = \omega_{\mu 1}$, and $\omega_2 = \omega_{\lambda 1}$. Using this notation, the initial approximations for the mode shapes and frequencies are nearly identical to the expressions in Eq.3.50a,b. The only difference is that the mass ratio ϵ is replaced by the more general effective mass ratio γ_{\perp}

$$\boldsymbol{\omega}_{1}^{(0)} = \boldsymbol{\omega}_{a} \left\{ \mathbf{i} + i\boldsymbol{\xi}_{a} + \frac{1}{2} \left[\boldsymbol{\gamma}_{11} + (i\boldsymbol{\xi}_{a} + \boldsymbol{\beta})^{2} \right] \right\}$$
(46a)

$$\boldsymbol{\omega}_{2}^{(0)} = \boldsymbol{\omega}_{a} \left\{ 1 + i\xi_{a} - \frac{1}{2} \left[\boldsymbol{\gamma}_{1} + (i\xi_{a} + \beta)^{2} \right] \right\}$$
(45b)
$$\boldsymbol{\psi}_{2}^{(0)} = i_{a} \left[(0) \boldsymbol{\varphi}_{a} - (0) \boldsymbol{\varphi}_{a} + (i\xi_{a} + \beta)^{2} \right] \left\{ (1 + i\xi_{a} - \frac{1}{2} \left[(1 + i\xi_{a} + \beta)^{2} \right] \right\}$$
(45b)

$$\phi_1^{(0)} = [\alpha_1^{(0)} 0 \cdots 01]^T$$
(46c)
$$\phi_2^{(0)} = [\alpha_2^{(0)} 0 \cdots 01]^T$$
(46d)

$$\alpha_{1}^{(0)} = \frac{1}{\zeta_{11}} \left\{ -\beta - i\xi_{il} - [\gamma_{11} + (i\xi_{il} + \beta)^{2}] \right\}$$
(46e)

$$\alpha_{2}^{(0)} = \frac{1}{\zeta_{11}} \left\{ -\beta - i\xi_{d} + [\gamma_{12} + (i\xi_{d} + \beta)^{2}] \right\}$$
(46f)

Higher order approximations for the mode shapes are obtained using the error analysis developed in Eqs.3.69-71a. The first and n+1 coordinates remain unchanged and the coordinates corresponding to the detuned modes are similar to those in Eq.44

$$\boldsymbol{\phi}_{i}^{(1)} = \left[\alpha_{i}^{(0)} - \frac{f_{21}}{G_{2}} \cdots - \frac{f_{n1}}{G_{n}} \right]^{\prime}$$
(47)

From the above solutions, the final low-order approximations are

$$\boldsymbol{\omega}_{i}^{(4)} = \boldsymbol{\omega}_{i}^{(0)} \tag{48a}$$

$$\boldsymbol{\phi}_{1}^{T} = \left[\boldsymbol{\alpha}_{1}^{(0)} \boldsymbol{\epsilon}_{21} \boldsymbol{\zeta}_{21} \frac{\boldsymbol{\omega}_{11}^{2}}{\boldsymbol{\omega}_{22}^{2} - \boldsymbol{\omega}_{11}^{2}} \cdots \boldsymbol{\epsilon}_{n1} \boldsymbol{\zeta}_{n1} \frac{\boldsymbol{\omega}_{11}^{2}}{\boldsymbol{\omega}_{nn}^{2} - \boldsymbol{\omega}_{11}^{2}} \mathbf{1} \right]^{T}$$
(48b)

والمراجع ومعاري مسر بالمركبة ومنهو والمريج فالمراجع المحمد والمراجع والمحموم والمراج

where i=1.2 denotes the two modes arising from the single tuning.

4.4.1.4 Multiply Tuned Modes

The eigenvalue analysis of a multiply tuned MDOFZSDOF system is a generalization of the analysis of singly tuned systems. Although simple closed form expressions can not be obtained for this system, it is possible to reduce the analysis to a small eigenvalue problem.

Let *I* modes of the primary subsystem be tuned to the secondary subsystem; for notational convenience assume that the first *I* primary subsystem modes (not necessarily basing the lowest *I* frequencies) are the tuned modes, i.e. $I_i = \{1, ..., I\}$ and $I_a = \{l+1, ..., n\}$. Initial approximations are obtained for the mode shapes by neglecting the effect of the detuned modes, as in the previous section. When the detuned coordinates $l \neq 1, ..., n$ are eliminated from the original $n+1 \times n+1$ order eigenvalue problem given in Eq.39a, the result is a relatively small $l+1 \times l+1$ problem

$$\mathbf{\Gamma}_{\boldsymbol{\beta}}(\boldsymbol{\omega}^{(0)}) \begin{bmatrix} \boldsymbol{\phi}_{\boldsymbol{\beta}}^{(0)} \\ \mathbf{i} \end{bmatrix} = 0$$
(49)

where $\Gamma_{l}(\omega)$ is the submatrix of $\Gamma(\omega)$ corresponding to the coordinates $\{1, 2, ..., l, n+1\}$. The *l*-vector $\phi_{l}^{(m)}$ corresponds to the first *l* coordinates of the *i*-th mode shape $\phi_{l}^{(m)}$ and is used in the initial approximation

$$\boldsymbol{\phi}_{\perp}^{(0)} = \begin{bmatrix} \boldsymbol{\phi}_{\perp}^{(0)} \\ 0 \\ 1 \end{bmatrix}$$
(50)

A higher order appreximation is found through an error analysis, as before. The result is

$$\boldsymbol{\phi}_{i}^{(1)} = \begin{bmatrix} \boldsymbol{\phi}_{i}^{(0)} \\ \boldsymbol{\phi}_{i}^{(0)} \\ \boldsymbol{\phi}_{i}^{(0)} \\ 1 \end{bmatrix}$$
(51a)

where $\phi_{d_1}^{(1)}$ is the n-1 vector

$$\boldsymbol{\phi}_{di}^{(0)} = \left[-\frac{f_{i+1,1}}{G_{i+1}} \cdots - \frac{f_{n_1}}{G_n} \right]^T$$
(51b)

and the polynomials are evaluated at the frequencies $\omega_{c}^{(0)}$ found from Eq.49. Finally, the above can be reduced to the following low-order expressions

$$\boldsymbol{\phi}^{\top} \coloneqq \left[\begin{array}{c} \boldsymbol{\phi}_{\perp}^{(0)} \\ \boldsymbol{\phi}_{\perp}^{(b)} \\ 1 \end{array} \right]$$
(52a)

$$\boldsymbol{\omega}^{*} \approx \boldsymbol{\omega}^{(0)} \tag{52b}$$

where

$$\boldsymbol{\phi}_{ij} = \left[\boldsymbol{\epsilon}_{i+1,1} \boldsymbol{\zeta}_{i+1,j} \frac{\boldsymbol{\omega}_{ij}^2}{\boldsymbol{\omega}_{ij}^2 + 1 - \boldsymbol{\omega}_{ij}^2} \cdots \boldsymbol{\epsilon}_{i+1} \boldsymbol{\zeta}_{ij} \frac{\boldsymbol{\omega}_{ij}^2}{\boldsymbol{\omega}_{ij}^2 - \boldsymbol{\omega}_{ij}^2} \right]$$
(52c)

Note the similarity between the above expressions and Eqs.48a.b.

4.4.1.5 Closely Spaced Primary Detuned Modes

As noted in Section 4.4.1.1, the analysis of closely spaced detuned modes may require special attention. The theoretical method for deriving these mode shapes which is consistent with the results of the previous sections would require the solution of a small eigenvalue probtem and is outlined below. However, it is shown in Section 4.4.2.5 that this derivation is unnecessary if the quantity of interest is the response of the secondary subsystem. If the expressions for the detuned mode shapes and frequencies in Eqs.42a,b are used directly, the results for the system response are shown to be the same as those obtained from the theoretical mode shapes. Therefore, the mode shapes and frequencies derived in Section 4.4.1.1 can be used for all detuned modes, regardless of the spacing of the frequencies. If the mode shapes of closely spaced detuned modes are themselves efficiency, the derivation for the theoretical low-order expressions given below can be used.

The analysis of systems with closely spaced detuned modes is similar to the analysis in the preceding section. Assume the first l primary modes are closely spaced and detuned from the secondary subsystem. The initial approximations for these l modes are obtained by neglecting the effect of those primary modes that are not among the closely spaced modes. Thus, the original $n+1 \times n+1$ order eigenvalue problem reduces to the same $l+1 \times l+1$ problem in Eq.49. There are l+1 solutions to this equation, however only the l solutions associated with the primary subsystem are used. The subsequent analysis is essentially the same as in the previous section and the final low-order expression for the mode shapes and frequencies are given by

Eqs.52a-c.

4.4.1.6 Example Study

The complex modal properties of the example system in Fig.4.2 were computed using the formulations developed in this section and were compared with exact results obtained by using a complex eigenvalue solver from the IMSL library [1]. The frequencies are shown in Table 3 and are plotted in Figs.4.6a-c for various values of the equipment frequency ω_{11} and the mode shapes for the multiply tuned case (i.e., $\omega_{x1} = 1.00 \text{ rad/sec}$) are shown in Table 4. The nonclassical damping character of the multiply tuned system is apparent in the mode shapes, which have imaginary components. Good agreement between approximate and exact values is found in all cases.

The effect of the equipment mass is illustrated in Fig.4.6b, where the frequencies corresponding to $\epsilon = .01, .005$, and .001 are represented by points .4, *B*, and *C*, respectively (which can be compared with Fig.3.3a-c from the 2-DOF system study). Mode 1 is not affected by the mass ratio because it is detuned. The frequencies corresponding to the other modes converge to the subsystem natural frequencies as was observed in the study of the 2-DOF system.

4.4.2 Spectral Moments

As stated earlier, the most direct method for finding the moments is through numerical computation. The mode shapes and frequencies of any MDOF/SDOF system can be calculated using the formulae developed above and the results substituted into the modal decomposition method developed in Chapter 2. It is straightforward to implement this procedure into a computer; the complex eigenvalue solution, which is of small order and is only required for multiply tuned systems, can be solved by routines found in standard libraries such as the IMSL [1]

For the example system in Fig.4.2, the results of this numerical computation are compared with exact results obtained by integrating the complex frequency response function in Figs.4.7a-c. Also, the mean zero-crossing rate, ν_{s} and the shape factor, δ_{s} are computed and plotted in Figs.4.8a,b. All plots show good comparison between exact and the proposed approximate results.

At tuning, the response of the equipment is dominated by oscillatory motion at the equipment frequency, thus the mean zero-crossing rate is close to $\nu = \omega_{eff}\pi$ [19]. The line representing this equation is plotted in Fig.4.8a and it can be seen that the actual values for ν approach this line at tuning. For other values of ω_{eff} , the contribution from the detuned modes of the primary system to the response motion becomes more significant, and the contribution from those modes with frequencies which are less than ω_{eff} tend to lower the values for the mean zero-crossing rate.

The shape factor, δ , which is a measure of the band-width of the response process, becomes small at tuning due to the predominance of the response of the tuned modes in the response motion. This is reflected by the marked decreased values of δ in Fig.4.8b at the two tuning frequencies, $\omega_{x1} \approx .38$ and 1.00 rad/sec. For $\omega_{x1} = .38$ rad/sec, it was noted that the complex frequency response function $H_1(\omega_1)$ has only a single peak, thus the power spectral density function is narrow-banded and δ has a very small value. For $\omega_{x1} = 1.00$ rad/sec, the complex frequency response function had two significant peaks, one at the tuned frequencies and coother at the detuned frequencies, thus the power spectral density function is not as narrowly-banded and δ is not as small as in the previous case.

Although the numerical approach to finding the spectral moments is useful in provided applications, it is instructive to utilize the algebraic forms for the mode shapes and frequencies derived in the previous section in order to obtain expressions for the response. Closed form solutions will be derived for response to white noise input for all but the multiply tuned systems. Although these expressions are not as accurate as the numerical method outlined above, they provide important information about PS systems which would be hidden in a numerical or parametric analysis.

4.4.2.1 Totally Defuned Systems with Well-Spaced Modes

Totally detuned systems with well-spaced modes are the simplest and most basic to analyze. Since totally detuned systems are classically damped and the correlations between well-spaced modes are negligible, the modal decomposition method simplifies to the squareroot-of-sum-ot-squares (SRSS) method [16]. The expressions for the mode shapes and frequencies which were derived in Section 4.4.1.2 can be substituted directly into the formula for the effective participation factors given in Eq.3.79. The factors associated with the primary subsystem modes are

$$\Psi = r \left[\frac{q_{i,i}, \omega_i; \zeta}{\omega_i - \omega_i} + q \right] \qquad i = 1, \cdots, n$$
(53a)

and that associated with the secondary subsystem mode is

$$\boldsymbol{\psi}_{i,ij} = \boldsymbol{q}_{i,ij} \left[\sum_{i=1}^{l} \frac{c_i c_i^{\dagger} \tilde{\boldsymbol{\zeta}}}{\boldsymbol{\omega}_{ij}^{\dagger} - \boldsymbol{\omega}_{ij}} + c_{i,ij} \right]$$
(53b)

The moments λ are given by the expressions in Eq.2.48a-c. Substituting into the formula Eq.2.38 the final expressions for the spectral moments are

$$\lambda_{n,n} = \frac{\pi G_{n,n}}{4} \left\{ \sum_{j=1}^{n} r_{ij}^{(j)} \left[\frac{q_{n+1} \omega_{n,1}^{2} \zeta_{j-1}}{\omega_{n,1}^{2} - \omega_{n,2}^{2}} + q \right]^{2} \frac{\omega_{n,n}^{(m-3)}}{\xi_{n,n}} + q_{n,n}^{(m-3)} \left[\sum_{j=1}^{n} \frac{r_{j} \omega_{n,1}^{2} \zeta_{j-1}}{\omega_{n,2}^{2} - \omega_{n,2}^{2}} + r_{n+1} \right]^{2} \frac{\omega_{n,1}^{(m-3)}}{\xi_{n,1}} \right\}$$
(54)

Note that for n = 1 this reduces to a result similar to the expression for the 2-DOF system in Eq.3.82.

4.4.2.2 Singly Tuned Systems

In this section, singly tuned systems with well spaced primary subsystem modes are considered. The well-spaced assumption allows a separate analysis of the detuned modes and the tuned modes in the modal decomposition method. Thus, the spectral moment can be expressed as a sum of two additive components,

$$\lambda_m = \lambda_{dm} + \lambda_m \tag{55}$$

where λ_{du} and λ_{du} constitute the contributions of the defuned and funed modes to the spectral moments, respectively. The defuned moment λ_{du} arises from the n-1 defuned primary mode shapes and is similar to the expression in Eq.54.

$$\lambda_{dn} = \frac{\pi G_{vv}}{4} \sum_{i \in J_{d}} e^{i \left[\frac{q_{n+i} \omega_{i1}^{2} \xi_{i1}}{\omega_{i1}^{2} - \omega_{i1}^{2}} + q \right]^{2} \frac{\omega_{\mu}^{(i)}}{\xi_{i}}}{\xi_{i}}$$
(56)

where the summation above is taken over the detuned modes I_{ab} . Note that the second summation in Eq.54 arises from the contribution of the secondary subsystem mode in the totally detuned system; therefore it is not included here.

The tuned moment λ_{10} is obtained from the two singly tuned mode shapes of the system and its derivation follows the analysis of tuned 2-DOF systems in Section 3.4.2.2. Taking lowest order terms for the moments, the result is identical to the low-order expression Eq.3.65a,b

$$\lambda_{m} = \frac{\pi G_{m}}{8} \frac{q_{m+1} r_{1} \zeta_{11}^{2} \xi_{n} \omega_{n}^{2}}{\xi_{m} \xi_{m} \xi_{m} \xi_{m}^{2} + \beta^{2}} + \frac{\gamma_{11} \xi_{n}^{2}}{\gamma_{11} \xi_{n}^{2}}$$
(57)

The detuned moment $\lambda_{\mu\nu}$ is small in comparison with the tuned moment $\lambda_{\mu\nu}$. However, this term is not ignored since it is important for the calculation of the factors ν and δ .

4.4.2.3 Multiply Tuned Systems

The analysis of multiply tuned systems follows the same line of reasoning as used for singly tuned systems. The expression for the detuned moment λ_{abc} is identical to Eq.56, however, there is no simple closed form algebraic solution for the tuned moment λ_{abc} . This quantity must be computed numerically by solving for the tuned mode shapes using the reduced eigenvalue problem in Section 4.4.1.4 and substituting the results directly into the modal decomposition method.

4.4.2.4 Systems with Closely Spaced Primary Modes

In the above, it was assumed that the defuned primary modes are widely spaced from each other, therefore the relatively small correlations between these modes were ignored. However, if these modes are closely spaced, the correlations become significant and another term λ_{en} must be included in the expressions for the moments λ_n :

$$\lambda_m = \lambda_{dm} + \lambda_{m} + \lambda_{cm} \tag{58}$$

This term would have the general form

$$\boldsymbol{\lambda}_{i,m} = \sum_{i=1,i,j} \left[C_{i} \boldsymbol{\lambda}_{m-1} - \mathcal{D}_{i} \boldsymbol{\lambda}_{m+1,i} + E_{ij} \boldsymbol{\lambda}_{m+2,j} \right]$$
(59)

where

1

$$I = \begin{cases} \text{pairs of closely spaced modes } (i,j) \\ \text{excluding pairs of tuned modes} \end{cases}$$
(60)

and the sum includes correlations between detuned primary modes and singly or multiply tuned modes.

Closed form expressions for $\lambda_{i,i}$ can be obtained only for closely spaced detuned modes. Such modes are classically damped, therefore, $D_{i,i} = E_i = 0$ and $C_i = a a_i$ and the cross term between modes *i* and *j* reduces to

$$C|\boldsymbol{\lambda}_{n+1}| = \psi|\psi_{1}\boldsymbol{\rho}_{n+1}\sqrt{\boldsymbol{\lambda}_{n+1}}\boldsymbol{\lambda}_{n+1}$$
(61)

where ψ_{i} are the effective participation factors given in Eq.53a and ρ_{abc} are the correlation coefficients defined in Chapter 2 (e.g., see Eqs.2.56 for wide-band inputs). For defuned modes closely spaced to a tuned mode, such simple forms for C_{abc} , D_{cl} , and E_{abc} do not exist, and the corresponding cross terms in Eq.59 must be evaluated numerically

4.4.2.5 Systems with Closely Spaced Primary Modes

Assume that the first *f* primary subsystem modes are defuned from the secondary subsystem and are very closely spaced and the damping ratio. If these modes are approximately equal. From physical considerations, it is expected that modes with nearly identical frequencies and damping ratios respond to a common input as a collective unit. This can be shown rigorously using an argument based on frequency response methods. Thus, these modes can be replaced by a single equivalent mode containing their essential properties. The frequency ω_i and damping ratio ξ_i of the equivalent mode are given by the averages

$$\omega_{I} = \frac{1}{I} \sum_{i=1}^{I} \omega_{ii} \quad \xi_{I} = \frac{1}{I} \sum_{i=1}^{I} \xi_{ii}$$
(62)

The effective participation factor ψ_1 is given by the sum of the modal participation factors ψ_2 , given in Eq.53a

$$\Psi_{I} = \sum_{i=1}^{I} \Psi_{i} = \sum_{i=1}^{I} r_{i} \left[\frac{q_{n+1} \omega_{z_{1}}^{2} \zeta_{z_{1}}}{\omega_{z_{1}}^{2} - \omega_{p_{n}}^{2}} + q_{i} \right]$$
(63)

It follows that the detuned moment λ_{dot} is simply

 $\lambda_{dm} = |\Psi_T|^2 \lambda_{m+T}$ + contribution from other defined modes (64) where $\lambda_{m,TT}$ is the spectral moment corresponding to the frequency ω_T and damping ratio ξ_T .

The same solutions for the response is obtained if the mode shapes and frequencies for widely spaced detuned modes are used. Using the participation factors Ψ obtained in Section 4.4.2.1, the expression for the moment λ_{abc} is

$$\lambda_{abc} = \sum_{i=1}^{L} \sum_{j=1}^{L} \psi_{i} \psi_{j} \lambda_{ac,ij} + \text{other contributions}$$
(65)

Since the modes are closely spaced and the damping ratios are assumed to be approximately equal,

$$\lambda_{m,j} \approx \lambda_{m+l} \quad \text{for all } i, j \leq l \tag{66}$$

Then the above expression for λ_{div} simplifies to

$$\lambda_{duv} \approx \left(\sum_{i=1}^{J} \sum_{j=1}^{J} \psi_{ij} \psi_{j} \right) \lambda_{m,II} + \text{other contributions}$$

= $|\psi_{I}|^{2} \lambda_{m,II} + \text{other contributions}$ (67)

which is the same result as Eq.64. The latter approach is more suitable for computer implementation since the same solutions for the mode shapes, frequencies, and participation factors are used for closely spaced and widely spaced detuned modes.

4.5 Non-Interaction Results

4.5.1 Introduction

The effect of interaction between the primary and secondary subsystems has been included in both the frequency response function analysis in Section 4.3 and the modal decomposition method in Section 4.4. However, for many PS systems, the mass of the secondary subsystem is considerable smaller than the masses of the primary subsystem and the results from Section 3.5 for 2-DOF systems suggest that the interaction effect would be negligible for such systems. Although the results of the preceding sections would remain valid for this case, it is worthwhile to genalyze the system without accounting for interaction for several reasons:

- 1. The results are simpler than the results which include interaction.
- 2. Closed form expressions will be derived for results that were not obtainable in that form in the previous sections.
- 3. Comparisons can be made between interaction and non-interaction results.

The frequency response analysis is presented first. Then, after deriving closed form solutions for the mode shapes, the modal decomposition method is presented.

4.5.2 Frequency Response Function Approach

The procedure in Section 3.5 is used to obtain the complex frequency response function $H_1(\omega)$. The equations of motion are decoupled into two sets of equations. The first set corresponds to the response of the primary subsystem to the base input and is given by

$$\hat{z}_{p_{e}}^{(nor)} + 2\omega_{p_{e}}\xi_{p_{e}}\hat{z}_{p_{e}}^{(nor)} + \omega_{p_{e}}^{2}\hat{z}_{p_{e}}^{(nor)} = -r_{e}\hat{x}_{e} \qquad i=1, \cdots, n$$
(68)

and the second set is an equation or the response of the secondary subsystem to the motions at the support points,

$$\vec{z}_{s1}^{(non)} + 2\omega_{s1}\xi_{s1}\vec{z}_{s1}^{(non)} + \omega_{s1}^{2}z_{s1}^{(non)} = \sum_{j=1}^{n} [\omega_{s1}^{2}z_{j} + 2\xi_{s1}\omega_{s1}\vec{z}_{j}]\zeta_{s1} - r_{n+1}\vec{x}_{n}$$
(69)

where the secondary DOF z_{s1} is identical to the nodal coordinate x_{n+1} . The Fourier transforms $Z_{\mu}^{(non)}(\omega)$ and $Z_{s1}^{(non)}(\omega)$ of the subsystem responses $z_{\mu}^{(non)}(t)$ and $z_{s1}^{(non)}(t)$ are

$$Z_{p_{i}}^{(+)ar}(\boldsymbol{\omega}) = -\frac{X_{g}(\boldsymbol{\omega}) m_{p_{i}} r_{i}}{G_{i}(\boldsymbol{\omega})} \qquad i=1,\cdots,n$$
(70a)

$$Z_{n+1}^{(non)}(\omega) = \frac{X_{c}(\omega)}{g_{1}(\omega)} \left[\sum_{i=1}^{n} \frac{f_{i1}(\omega) m_{p_{i}} r_{i}}{G_{i}(\omega)} - m_{s1} r_{n+1} \right]$$
(70b)

From these expressions, the Fourier transform $Y^{(non)}(\omega)$ of the response variable $y(t) = \mathbf{q}^t \mathbf{z}$ is obtained

$$Y^{(non)}(\omega) = \sum_{i=1}^{n} q_i Z_{p_i}^{(non)}(\omega) + q_{n+1} Z_{s1}^{(non)}(\omega)$$

= $X_s(\omega) \left\{ \sum_{i=1}^{n} \frac{q_{n+1} f_{i1}(\omega) m_{p_i} r_i}{g_1(\omega) G_i(\omega)} - \sum_{i=1}^{n} \frac{q_i m_{p_i} r_i}{G_i(\omega)} - \frac{q_{n+1} m_{s1} r_{n+1}}{g_1(\omega)} \right\}$ (71)

It follows that the frequency response function is

$$H_{i}^{(non)}(\omega) \approx \frac{g_{1}\prod_{i=1}^{n}G_{i}}{d^{(mn)}(\omega)} \left\{ \sum_{i=1}^{n} \frac{q_{n+1}f_{i1}(\omega)m_{pi}r_{i}}{g_{1}(\omega)G_{i}(\omega)} - \sum_{i=1}^{n} \frac{q_{i}m_{pi}r_{i}}{G_{i}(\omega)} - \frac{q_{n+1}m_{s1}r_{n+1}}{g_{1}(\omega)} \right\}$$
(72a)

$$= \sum_{i=1}^{n} \frac{q_{n+1} f_{i1}(\omega) m_{in} r_{i}}{g_{1}(\omega) G_{i}(\omega)} - \sum_{i=1}^{n} \frac{q_{i} m_{pi} r_{i}}{G_{i}(\omega)} - \frac{q_{n+1} m_{n+1} r_{n+1}}{g_{1}(\omega)}$$
(72b)

where

$$d^{(\omega_{\alpha})}(\omega) = g_1(\omega) \prod_{i} G_i(\omega)$$
(72c)

Note the similarity between Eq.72a with Eq.36, in the latter case $\gamma_{i,1}$ order terms are included in the polynomial $d(\omega)$ to account for interaction. This important fact is expressed mathematically as $z_{i,1}$.

$$\lim_{\alpha \in U^{-n}} H_{i}(\omega) = H_{i}^{(i_{0}\alpha)}(\omega)$$
(73)

which indicates that all of the closed form results previously obtained in this chapter can be applied to the non-interaction study by taking the limit $m_{s1} \rightarrow 0$, as was done for the 2-DOF system.

For the example system in Fig.4.2, a comparison between the transfer function T_{i} . (ω) = $|H_i(\omega)|^2$ for interaction and non-interaction analysis is shown in Fig.4.9 for various values of the secondary subsystem mass. The differences are most notable for values of ω near the tuned mode particularly for larger values of the secondary mass where interaction is more prominent. For other values of ω , the transfer function is insensitive to interaction. This can be shown analytically simply by examining the order relationships in the expressions for the functions $H_i(\omega)$ and $H_i^{(man)}(\omega)$.

The spectral moments of the response of the system is found by integrating the transfer function $T_{i1}^{(4000)}(\omega)$ directly with the input power spectral density function as in Section 4.3.3. In Fig.4.10, plots of the non-interaction and interaction moments are given for the system in Fig.4.2 with varying values for the mass and frequency of the secondary subsystem. The result is similar to the findings in Chapter 3: the difference between λ_0 and $\lambda_0^{(4000)}$ are greatest at tuning and diminish at detuning.

4.5.3 Modal Decomposition Method

4.5.3.1 Introduction

The approach taken here is essentially the same as in Section 4.4: the expression for the mode shapes and frequencies are derived for the combined system and the results substituted into the modal decomposition method of Chapter 2. However, by neglecting interaction, it is possible to obtain closed form expressions for all mode shapes and frequencies, including multiply tuned modes due to the simplicity of decoupled PS systems. Furthermore, closed form expressions are derived for the factors a_i and c_j defined in Eq.2.26, which are key factors for the modal decomposition method. The final results for the spectral moments are easily obtainable from the expressions for a_i and c_j and the original parameters of the two subsystems.

4.5.3.2 Closed Form Expressions for the Mode Shapes

The original eigenvalue problem

$$\boldsymbol{\Gamma}(\boldsymbol{\omega})\boldsymbol{\phi} = 0 \tag{74}$$

from Eqs.39a,b is reinvestigated. First, the modes associated with the primary subsystem are analyzed. It is intuitively clear that the frequencies associated with these n modes are given by the original primary subsystem frequencies

$$\omega_{i} = \omega_{pi} \left(\sqrt{1 - \boldsymbol{\xi}_{pi}^{2}} + i \boldsymbol{\xi}_{pi} \right) \quad i = 1, \dots, n$$
(75)

The corresponding mode shapes are derived by substituting ω_{\perp}^{-1} into Eq.74 and solving the eigenvalue problem. The solution is

$$\boldsymbol{\phi}_{1}^{*} = \begin{bmatrix} 0 & \cdots & 0 + 0 & \cdots & 0 - \frac{f_{-1}(\omega_{1})}{g_{1}(\omega_{1})} \end{bmatrix}^{T} \\ \approx \begin{bmatrix} 0 & \cdots & 0 + 0 & \cdots & 0 - \frac{\zeta_{+1}\omega_{+1}^{2}}{2(\beta_{+1} + i\xi_{d_{1}+1})\omega_{u_{-1}}^{2}} \end{bmatrix}^{T}$$
(76)

where the unit term is at the *i*th coordinate and the parameters ω_{a-1} , β_{-1} , and ξ_{d-1} are generalizations of the average frequency, detuning, and damping difference parameters used in the analysis of the 2-DOF system

$$\boldsymbol{\omega}_{a,i} = \frac{\boldsymbol{\omega}_{pi} + \boldsymbol{\omega}_{x1}}{2} \quad \boldsymbol{\beta}_{i} = \frac{\boldsymbol{\omega}_{pi} - \boldsymbol{\omega}_{x1}}{\boldsymbol{\omega}_{a,i}} \quad \text{and} \quad \boldsymbol{\xi}_{d,i} = \boldsymbol{\xi}_{pi} - \boldsymbol{\xi}_{x1} \tag{77}$$

The derivation of the modal properties associated with the equipment is similar to the

above analysis. The frequency is given by the equipment subsystem frequency

$$\omega_{11} = \omega_{11} (\sqrt{1 - \xi_{11}} + i\xi_{11})$$
(78)

which, when substituted into Eq.74, yields the following expression for the mode shapes

$$\boldsymbol{\phi}_{n+1}^{*} = \left[-\frac{f_{(1}(\boldsymbol{\omega}_{n+1}^{*}))}{G_{1}(\boldsymbol{\omega}_{n+1}^{*})} \cdots - \frac{f_{n}(\boldsymbol{\omega}_{n+1}^{*})}{G_{n}(\boldsymbol{\omega}_{n+1}^{*})} 1 \right]^{t}$$

$$\approx \left[\frac{\epsilon_{11}\zeta_{11}\boldsymbol{\omega}_{n1}^{2}}{2(\beta_{11}+i\xi_{n}+1)\boldsymbol{\omega}_{n,1}^{2}} \cdots \frac{\epsilon_{n}\zeta_{n}\omega_{n}^{2}}{2(\beta_{n1}+i\xi_{n})\boldsymbol{\omega}_{n,1}^{2}} 1 \right]^{t}$$
(79)

These expressions reduce to the results in Sections 4.4.1.1 and 4.4.1.2 for the defuned cases.

It appears that the expression for the mode shape ϕ_{m+1}^{-1} is indeterminate since it involves an arbitrarily small parameter ϵ_{\pm} , and the terms β_{\pm} and ξ_{m+1} in the denominator may also be small or zero. However, the limit $m_{\pm} \rightarrow 0$ is not taken until after the coefficients a and c of the modal decomposition rule are derived. Also, the problem arising from the condition $\beta_{\pm} \rightarrow \xi_{m+1} = 0$ is resolved after the modal quantities are combined, as will be shown subsequently.

4.5.3.3 Spectral Moments

As stated earlier, closed form expressions will be obtained for the factors a and c, defined in Eq.2.26, which are the key factors of the modal decomposition method. Due to the simplicity of the expressions for the mode shapes and frequencies, the derivation is straightforward.

By following the matrix multiplications in Eqs.2.17a,c and 2.20 and taking the limit $m_{v1} \rightarrow 0$, the following expressions for the factors b_i are obtained, which are independent of the mass ratio ϵ_{v1} :

$$b_{i} = \frac{ir_{i}}{2\omega_{m}} \left[q_{i} - \frac{q_{n+1}\zeta_{i+1}\omega_{\chi_{1}}^{2}}{2(\beta_{i+1} + i\zeta_{d,i+1})\omega_{d,i+1}^{2}} \right] \quad \text{for } i = 1, \dots, n$$
(80a)

$$b_{n+1} = \frac{iq_{n+1}}{2a_{s,1}} \left[r_{n+1} + \sum_{i=1}^{n} \frac{r_i \zeta_{i,1} \omega_{s,1}^2}{2(\beta_{i,1} + (\xi_{d_i,1}) \omega_{u,i,1}^2)} \right]$$
(80b)

The factor a_i and c_i are found from Eq.2.26

$$a_{i} \approx 2\omega_{p_{i}} \operatorname{Im} b_{i} = -\frac{q_{n+1}\zeta_{i1}r_{i}\beta_{i1}\omega_{i1}^{2}}{2(\beta_{i1}^{2} + \xi_{i1}^{2})\omega_{i1}^{2}} + r_{i}q_{i}$$
(81a)

$$a_{n+1} \approx 2\omega_{n+1} = \sum_{i=1}^{n} \frac{q_{n+1}\zeta_{i,1}r_i\beta_{i,1}\omega_{i,1}^2}{2(\beta_{i,1}^2 + \xi_{i,1}^2)\omega_{n,1}^2} + r_{n+1}q_{n+1}$$
(81b)

$$c \approx 2\operatorname{Re}b = -\frac{q_{n+1}\xi_{-1}r_{*}\xi_{-1,-1}\omega_{+1}^{2}}{2\omega_{p_{*}}(\beta_{+1}^{2}+\xi_{-1,-1}^{2})\omega_{-1,-1}^{2}}$$
(81c)

$$c_{n+1} \approx 2\operatorname{Re} b_{n+1} \neq -\sum_{i=1}^{n} \frac{\omega_{p_i}}{\omega_{s_i}} c_i$$
(8)d)

From these expressions, the spectral moments are easily obtained from Eqs.2.34 and 2.36, which are repeated here:

$$C_{i} = a_{i}a_{i} \quad D_{i} = a_{i}c_{i} - a_{i}c \quad E_{i} = c_{i}c_{i}$$
(82a)

$$\lambda_{m} = \sum_{i=1}^{n} \sum_{j=1}^{n} (C_{i} \lambda_{m-j} - D_{ij} \lambda_{m+1,j} + E_{ij} \lambda_{m+2,j})$$
(82b)

The frequencies and damping ratios needed to calculate the cross-spectral moments are given by the original subsystem parameters as indicated in Eqs.75 and 78. Equations 81a-d and the above combination rule are in a form suitable for computer implementation. The moments calculated by the above expressions are compared with exact results for various values of the secondary masses in Fig.4.11. The results are similar to those observed in Fig.4.10.

For the case where $\xi_{d,i}$ and $\beta_{i,1}$ are small or zero, it suffices to give ξ_{d-1} an arbitrary residual number (for computer applications, this number would depend on the precision of the hardware). After combining modal responses, the end result would be consistent with previous results and would be independent of the parameter $\xi_{d,i}$; this will be shown presently with an example for the important case of perfect tuning.

4.5.3.4 Multiply Tuned Example

Consider a primary subsystem with all modes perfectly tuned to the secondary system and sharing the same damping ratio, i.e. $\xi_{d,i,1} = \beta_{i,1} = 0$ for all *i*. As mentioned earlier, a residual value $\Delta \ll 1$ is given to ξ_{d-1} to keep all terms well-defined. Then, the factors *a*, and *c* become

$$a_i \approx 0 \quad i = 1, \ldots, n \tag{83a}$$

$$c_i = -\frac{r_i \zeta_{i1}}{2\omega_a \Delta} \quad i = 1, \dots, n$$
(83b)

$$c_{n+1} = \frac{\sum_{i=1}^{n} r_i \zeta_{i,1}}{2\omega_n \Delta}$$
(83c)

where ω_{α} is the common frequency of the system modes. Substituting into Eqs.82a,b, the

expression for λ_{in} is obtained

$$\lambda_{n} = \sum_{i=1}^{n} \sum_{j=1}^{n} E_{i} \lambda_{n-j} + 2 \sum_{i=1}^{n} E_{i-n+1} \lambda_{n-j+1} + E_{i-1-n+1} \lambda_{i-n+1-j+1} \\ = \frac{\pi G_{ee} \left(\sum_{i=1}^{n} r_{i} \zeta_{-1}\right)^{2}}{16\omega_{e}^{\frac{1}{3} - m} \Delta^{2}} \left[\frac{1}{\xi_{p1}} - \frac{4}{\xi_{p1} + \xi_{s1}} + \frac{1}{\xi_{s1}} \right] = \frac{\pi G_{ee} \left(\sum_{i=1}^{n} r_{i} \zeta_{-1}\right)^{2}}{16\omega_{e}^{\frac{1}{3} - m} \xi_{p1} + \xi_{s1}}$$
(84)

The final result does not include the residual term Δ and is very similar to the result for the 2-DOF system in Eq.3.65b. The main difference is that the factor $\sum_{i=1}^{r} r \xi_{i-1}$ which represents the general physical coupling between the primary and secondary subsystems is included in the above expression for λ_{in} .

4.6 Floor Spectra

The floor spectrum for a prinary subsystem and a given ground input, is the response spectrum associated with the motion of the system at a selected attachment point. More precisely, it is defined as the mean of the peak displacements of a set of oscillators with variable values for damping ratios and frequencies subjected to the motion of the primary subsystem at the selected attachment point.

Clearly, floor spectra are special cases of the results developed in this chapter. Since the secondary subsystem is attached to the primary subsystem at a single attachment point, the vector \mathbf{q}_0 is of the form

$$\mathbf{q}_0 = [\mathbf{0} \cdots \mathbf{0} - 1 \mathbf{0} \cdots \mathbf{0} \mathbf{1}]^T$$
(85)

where the -1 is located at the coordinate associated with the attachment point.

The remainder of the analysis is based on the results derived in this chapter. Frequency response analysis or the modal decomposition method can be employed, and for the latter approach, input specified by its response spectrum can be used. Finally, interaction can be included or ignored, according to the particular application of the problem.

| Table 4.1. | Physical | Proper | ties of | the |
|------------|----------|--------|---------|-----|
| MDOF/ | SDOF E | kample | Systen | a |

| Subsystem | Parameter Relationships | | | |
|-----------|--|--|--|--|
| Primary | $k_1 = (1.02)^2 m_1$ (radians/sec) ² $k_2 = (.6057)^2 m_1$ (radians/sec) ² $m_2 = m_1$ | | | |
| Secondary | $k_{\perp} = \omega_{\perp}^{2} m_{\perp} - (\omega_{\perp} \text{ variable})$ $m_{\perp} = \epsilon m_{\perp} - (\epsilon \text{ variable})$ | | | |

Table 4.2. Modal Properties of the Fixed Base Subsystems

| Subsystem | Modal DOF | Frequency (rad/s) | Damping Ratio |
|-----------|--------------|----------------------|------------------|
| Primary 1 | 1 | 1.02 | 0.02 |
| Primary 2 | 2 | 0.374 | 0.023 |
| - | 3 | 0.98 | 0.06 |
| Secondary | 4 | ω | 0.01 |

Table 4.3 Frequencies for Example System ($\epsilon_{11}=0.01$)

| <u>س, ۱</u> | Mada | Exact Frequency | | Computed Frequency | | Error |
|-------------|------|------------------|------------|--------------------|------------|-------|
| (rad/sec) | MOUC | Real Part | Imag. Part | Real Par | Imag. Part | 1% |
| 0.38 | 1 | 0.371 | 0.007 | 0.371 | 0.007 | 0.0 |
| l | 2 | 0.382 | 0.005 | 0.382 | 0.005 | 0.0 |
| | 3 | 0.978 | 0.058 | 0.980 | 0.058 | 02 |
| | 4 | 1.019 | 0.020 | 1.020 | 0.020 | 01 |
| 1.00 | 1 | 0.376 | 0.008 | 0.374 | 0.008 | 0.5 |
| | 2 | 0.991 | 0.013 | 0.991 | 0.013 | 0.0 |
| | 3 | 0.978 | 0.057 | 0.977 | 0.057 | 0.1 |
| | 4 | 1.029 | 0.018 | 1.028 | 0.018 | 0.1 |
| 1.40 | 1 | 0.378 | 0.008 | 0.374 | 0.008 | 0.2 |
| | 2 | 0.978 | 0.058 | 0.980 | 0.058 | 0.2 |
| | 3 | 1.020 | 0.020 | 1.020 | 0.020 | 0.0 |
| | 4 | 1.402 | 0.014 | 1.400 | 0.014 | 0.2 |

| Mada DOF | | Exact Mode Shape | | Computed Mode Shape | | Error |
|----------|-----|------------------|------------|---------------------|------------|-------|
| Mode | DOF | Real Part | Imag. Part | Real Part | Imag. Part | 1/4. |
| 1 1 | 1 | 1,000 | 0,000 | 1.000 | 0.000 | 0.0 |
| | 2 | 0.000 | 0.000 | 0.000 | 0,000 | 0.0 |
| | 3 | 0.000 | 0.000 | 0.000 | 0.000 | 0.0 |
| 1 | 4 | 0.388 | 0.002 | 0.387 | 0.000 | 0.3 |
| 2 | 1 | -0.002 | 0.000 | -0.002 | 0.000 | 0.0 |
| | 2 | 0.110 | -0.300 | 0.110 | -0.300 | 8 0 0 |
| { | 3 | 0.022 | 0.019 | 0.022 | 0.019 | 0.0 |
| 1 | 4 | 1.000 | 0.900 | 1.000 | 0.000 | 0.0 |
| 3 | 1 | -0.062 | 0,000 | -0.002 | 0.000 | 0.0 |
| | 2 | -0.002 | -0.009 | -0,002 | -0.009 | 0.0 |
| 1 | 3 | 0.053 | -0.011 | 0.054 | -0.011 | 0.1 |
| | 4 | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |
| 4 | 1 | -0.002 | 0.000 | -0.002 | 0.000 | 0.0 |
| | 2 | -0.005 | -0.004 | -0.005 | -0.004 | 0.0 |
| } | 3 | -0.175 | -0.043 | -0.171 | -0.043 | 0.4 |
| 1 | 4 | 1,000 | 0.000 | 1.000 | 0.000 | 0.0 |

Table 4.4 Mode Shapes of Example System ($\epsilon_{11}=0.01$, $\omega_{s1}=1.0$)


Fig. 4.1. Example of ζ_{ij} Calculation

Frimary Mode 2 \$21-0.25

Primary Mode 1 ₅₁₁=0.75

Example System



ì

Figure 4.2. Example MDOF/SDOF System



Multiply Tuned System, wsi=1.0 rad/sec



Singly Tuned System, ω_{s1} =0.38 rad/sec

Fig.4.3. Distribution of Subsystem Free Vibration Frequencies







Fig.4.4b. Transfer Function $T_{\mu\nu}(\omega)$: Multiply Tuned System $\omega_{\nu i} = 1.0$ rad/sec. $\epsilon_{11} = 0.01$





:

- 136 -







Fig4.5c. Spectral Moment λ_2 derived from $H_s(\omega)$, $\epsilon_{11}=0.01$







Fig.4.6b. Complex Frequencies: Multiply Tuned System $\omega_{x_1}=1.0 \text{ rad/sec}, \epsilon_{11}=0.01, 0.005, 0.001$



- 141 -









.











Fig4.8b. Shape Factor 8, e₁₁=0.01



Fig4.9. Transfer Function $\mathcal{T}_{ix}^{(n,n)}(\omega)$ without Interaction





•





CHAPTER 5

ANALYSIS OF MULTI-DEGREE-OF-FREEDOM SECONDARY SYSTEMS ATTACHED TO SINGLE-DEGREE-OF-FREEDOM PRIMARY SYSTEMS

5.1 Introduction

In this chapter systems consisting of a SDOF primary subsystem supporting a MDOF secondary subsystem as shown in Fig.5.1 will be studied. Although this system appears to be entirely different from the MDOF/SDOF system studied in the previous chapter, the two systems are, in fact, strongly related. In mathematics, this relationship is called duality: every formula or theorem derived for one system has a nearly identical dual counterpart for the other system. Once this dual relationship between the two systems has been established, all of the results obtained for the MDOF/SDOF system in the previous chapter can be directly reformulated for the SDOF/MDOF system without further analytical derivation.

5.2 Definitions

5.2.1 Parameters

The first step in defining a duality relationship is to establish a one-to-one correspondence between the parameters of the two systems. In this case, the parameters defined for the n primary modes and the single secondary mode of the MDOF/SDOF system are associated with the parameters defined for the n secondary modes and the single primary mode of the SDOF/MDOF system, respectively. For instance, the natural frequencies for the subsystems of the MDOF/SDOF system, which were

$$\boldsymbol{\omega}_{p1}, \, \boldsymbol{\omega}_{p2}, \, \cdots, \, \boldsymbol{\omega}_{pn}, \, \boldsymbol{\omega}_{N}$$
 (1a)
are associated with the frequencies

 $\boldsymbol{\omega}_{x1}, \boldsymbol{\omega}_{x2}, \cdots, \boldsymbol{\omega}_{xn}, \boldsymbol{\omega}_{xn}$ (1b)

for the SDOF/MDOF system. To maintain the above correspondence, the numbering of the

modal degrees-of-freedom for the SDOF/MDOF system is as follows: the first n degrees of freedom correspond to the modes of the secondary subsystem and the n+1-th degree of freedom corresponds with the primary subsystem.

Most of the definitions of the previous chapter are applied in this chapter with obvious modifications. For instance, L is the set of all secondary subsystem modes tuned to the primary subsystem mode.

There were several expressions used in the previous chapter with two indices, the first associated with the primary subsystem and the second with the secondary subsystem. This convention is maintained in this chapter. Thus, the mass ratios are defined as

$$\boldsymbol{\epsilon}_{ij} = \frac{m_{sj}}{m_{p1}} \qquad i = 1, \cdots, n.$$
(2a)

Also, the terms ζ_{1i} are defined as

$$\zeta_{1i} = \frac{k_{1i}}{m_0 \omega_0^2} \quad \text{for } i = 1, \dots, n.$$
(2b)

and are interpreted as the displacement of the *i*th mode of the secondary subsystem produced by a static unit displacement of the primary subsystem. Similar definitions apply for the constants c_{12} , k_{12} , and the polynomial $f_{12}(\omega)$. It follows that the complex frequency response matrix $H(\omega)$ is

$$\mathbf{H}(\boldsymbol{\omega}) = \begin{bmatrix} g_{1}(\boldsymbol{\omega}) & \cdots & 0 & f_{11}(\boldsymbol{\omega}) \\ \cdots & \cdots & \cdots & \cdots \\ 0 & \cdots & g_{n}(\boldsymbol{\omega}) & f_{1n}(\boldsymbol{\omega}) \\ f_{11}(\boldsymbol{\omega}) & \cdots & f_{1n}(\boldsymbol{\omega}) & G_{1}(\boldsymbol{\omega}) \end{bmatrix}^{-1}$$
(3a)

where

$$g_{\nu}(\omega) = m_{\nu}(-\omega^2 + 2i\omega_{\nu}\xi_{\nu}\omega + \omega_{\nu}^2)$$
(3b)

$$G_{1}(\omega) = m_{\mu 1}(-\omega^{2} + 2i\omega_{\mu 1}\xi_{\mu 1}\omega + \omega_{\mu 1}^{2})$$
(3c)

$$f_{1\nu}(\boldsymbol{\omega}) = -k_{1\nu} - ic_{1\nu} \approx -\zeta_{1\nu} m_{\nu} (\boldsymbol{\omega}_{\nu}^2 + 2i\omega_{\nu}\xi_{\nu}\boldsymbol{\omega})$$
(3d)

for $i=1, \dots, n$.

5.2.2 Duality Relationships

The duality between the MDOF/SDOF and SDOF/MDOF systems is established by com-

paring their complex frequency response matrices $H(\omega)$ in Eqs.4.28a and 3a. The matrix in Eq.3a has an identical form to that in Eq.4.28a with the polynomials

$$G(\omega)$$
 replaced by $g(\omega)$, (4a)

$$g(\omega)$$
 replaced by $G_1(\omega)$, (4b)

 $f_{\beta}(\omega)$ replaced by $f_{\beta}(\omega)$. (4c)

For the MDOF/SDOF system, the matrix $H(\omega)$ and its inverse were used to derive results in the frequency response and modal analysis, respectively. Consequently, the derivations for the SDOF/MDOF system would be essentially the same as those in the previous chapter with the replacements in Eqs.4a-c. Thus, a repetition of the analysis is unnecessary and only the final results will be presented through the use of the dual relationships defined above.

5.2.3 Example System

To illustrate the major characteristics of SDOF/MDOF systems and check the accuracy of the formulations derived in this chapter, the example system shown in Fig.5.1 is used. A 2-DOF secondary system is supported by the SDOF primary subsystem and the base of the combined system. The dynamic properties of the subsystems are described in Tables 1 and 2 and are chosen so that the combined system would exhibit important characteristics found in general MDOF/SDOF systems, as was done for the study of MDOF/SDOF systems. By varying the stiffnesses k_2 and k_3 , it is possible to vary the frequencies of the secondary subsystem. One frequency of the secondary subsystem was chosen to be fixed at $\omega_{x2} = 1.0$ rad/sec, thus the secondary subsystem was always tuned to the primary system, which had a frequency of $\omega_{p3} =$ 1.02 rad/sec. The other frequency of the secondary subsystem was variable. For $\omega_{x1} = 0.7$ rad/sec, the system is singly tuned as shown in Fig.5.2a, and for $\omega_{x3} = 1.0$ rad/sec, the system becomes multiply tuned as shown in Fig.5.2b. The mass ratio, ϵ_3 is also chosen to be a variable parameter, as in the MDOF/SDOF example system. Finally, the damping ratio of the equipment is unequal to the damping ratios of the orimary subsystem, thus the combined system is, in general, non-proportionally damped.

The response quantity that will be investigated is the relative displacement between the

mass of the primary subsystem and the adjacent mass of the secondary subsystem.

5.3 Frequency Response Results

As a first application of the duality the expression for the complex frequency response function $H_{\ell}(\omega)$ of the response variable $y(\ell) = \mathbf{q}^{\ell} \mathbf{z}(\ell)$ is given. For the MDOE/SDOE system, $H_{\ell}(\omega)$ is given in Eq.4.36 and is repeated here

$$H_{s}(\boldsymbol{\omega}) \approx \frac{g_{1}(\boldsymbol{\omega}) \prod_{i} G_{i}(\boldsymbol{\omega})}{d(\boldsymbol{\omega})} \left[\sum_{j=1}^{n} \frac{q_{n+1}m_{j} r_{j} f_{j}(\boldsymbol{\omega})}{g_{1}(\boldsymbol{\omega}) G_{i}(\boldsymbol{\omega})} - \frac{q_{-j+1}m_{s+1}r_{j+1}}{g_{1}(\boldsymbol{\omega})} - \sum_{j=1}^{n} \frac{q_{-m_{j}} r_{j}}{G_{i}(\boldsymbol{\omega})} \right]$$
(5)

The dual form of the above is found through the relations in Eqs.1a,b and Eqs.4a-c

$$H_{1}(\omega) \approx \frac{G_{1}\prod_{i=1}^{n}g_{i}}{d(\omega)} \left[\sum_{i=1}^{n} \frac{q_{n+1}m_{n}r_{n}f_{1}(\omega)}{G_{1}(\omega)g_{i}(\omega)} - \frac{q_{n+1}m_{n}r_{n+1}}{G_{1}(\omega)} - \sum_{i=1}^{n} \frac{q_{i}m_{i}r_{i}}{g_{i}(\omega)} \right]$$
(6a)

where $d(\omega)$ is the characteristic polynomial

$$d(\omega) = \prod_{i=1}^{n} g_i(\omega) \left[G_1(\omega) - \sum_{\alpha \in I_i} \frac{f_{i_i}^2(\omega)}{g_i(\omega)} \right]$$
(6b)

These expressions are considerably simpler than the exact form of $H_1(\omega)$ which requires the inversion of an $n+1 \times n+1$ matrix.

A plot of $T_{i+}(\omega)$ for the example system in Fig.5.1 is shown in Fig.5.3a,b. In Fig.5.3a, the secondary subsystem frequency $\omega_{s1} = 0.7$ rad/sec is detuned, and a peak is found only for the tuned modes at $\omega = \omega_{\mu 1}$. There is also a small peak corresponding to the detuned secondary mode near $\omega = 0.7$ rad/sec, however it is considerably smaller than the first peak and is not visible on the plot. In Fig.5.3b, both secondary subsystem frequencies are multiply tuned to the primary subsystem frequency, and a single peak is found at $\omega = \omega_{s1}$ which is slightly larger and broader than the corresponding peak of the singly tuned system, as expected.

As in Section 4.3.3, the approximate and exact forms of the complex frequency response function $H_i(\omega)$ derived above were used to calculate the power spectral density function for the response of the example system in Fig.5.1 to white-noise base excitation. From this result, the spectral moments of the response were obtained by numerical integration and plotted as a function of the equipment frequency, ω_{x1} , in Figs.5.4a-c. For values of ω_{11} detuned from the primary subsystem frequency, the response of the system is dominated by the singly tuned modes which are relatively independent of the detuned secondary mode. Thus, the spectral moment maintains a large and nearly constant value. However, when the system is multiply tuned, an increase in the spectral moment arises at $\omega_{12} = 1.0$ rad/sec from the additional tuned secondary mode. The values for λ_{11} for m = 0, 1, and 2 are nearly identical, due to the fact that the frequencies of the dominant modes in the response are all nearly equal to 1.0 rad/sec. Also, the difference between approximate and exact results are slight, illustrating the accuracy of the perturbation methods

5.4 Modal Decomposition results

5.4.1 Mode Shapes and Frequencies

In this section, expressions for the mode shapes and frequencies of the SDOF/MDOF system are developed which are suitable for numerical evaluation and subsequent use in the modal decomposition analysis to provide accurate measures of the response. Approximate expressions for the spectral moments of response to white noise input are derived from the algebraic form of the mode shapes to provide further insight to the dynamic behavior of the system.

Due to the duality between the SDOF/MDOF and MDOF/SDOF systems, a derivation of the mode shapes and frequencies is unnecessary, and final expressions for these quantities are found directly from the results of Section 4.4.1 through the use of the relationships in Section 5.2.

5.4.1.1 Detuned Modes

For notational convenience, assume that the first secondary subsystem mode is defuned from the primary subsystem. Then, the first-order expressions for the frequency and mode shape of the combined system which are associated with mode 1 of the secondary subsystem are given by the dual forms of Eq.4.40b and 4.41, respectively:

$$\boldsymbol{\omega}_{1}^{(0)} = \boldsymbol{\omega}_{11}(\sqrt{1-\boldsymbol{\xi}_{11}} + \boldsymbol{i}\boldsymbol{\xi}_{11})$$
(7a)

$$\boldsymbol{\phi}_{1}^{(1)} = \left[1 \frac{f_{1}(f_{12})}{G_{1}g_{2}} \cdots \frac{f_{1}(f_{12})}{G_{1}g_{2}} - \frac{f_{2}}{G_{1}}\right]^{T}$$
(7b)

where all of the above polynomial are evaluated at $\boldsymbol{\omega} = \boldsymbol{\omega}_1^{(0)}$. The low-order approximations of the above are

$$\boldsymbol{\omega}' = \boldsymbol{\omega}_{s,i} (\mathbf{i} + i\boldsymbol{\xi}_{s,i}) \tag{8a}$$

$$\boldsymbol{\phi} \approx \left[1 \ 0 \ \cdots \ 0 \ \frac{\boldsymbol{\zeta}_{\pm 1} \boldsymbol{\epsilon}_{\pm 1} \boldsymbol{\omega}_{\pm}}{\boldsymbol{\omega}_{p}^{2} - \boldsymbol{\omega}^{2}} \right]$$
(8b)

Similarly, if the primary subsystem mode is detuned from all of the secondary subsystem modes, the first-order expressions associated with the primary subsystem mode are found from Eqs.4.43b and 4.44

$$\boldsymbol{\phi}_{i+1}^{(1)} = \begin{bmatrix} -\frac{f_{i+1}}{g_1} & \cdots & -\frac{f_{i+1}}{g_i} \end{bmatrix}^T$$
(9a)

$$\boldsymbol{\omega}_{1}^{(\text{in})} = \boldsymbol{\omega}_{11} \left(\sqrt{1 - \boldsymbol{\xi}_{1}^{2}} + i \boldsymbol{\xi}_{1} \right)$$
(9b)

where the above polynomials are evaluated at $\omega = \omega_{i+1}^{(0)}$. To low-order, the above reduces to

$$\boldsymbol{\phi}_{n+1} = \left[\zeta_{11} \frac{\boldsymbol{\omega}_{11}^2}{\boldsymbol{\omega}_{11}^2 - \boldsymbol{\omega}_{p1}^2} \cdots \zeta_{1n} \frac{\boldsymbol{\omega}_{p1}^2}{\boldsymbol{\omega}_{p1}^2 - \boldsymbol{\omega}_{p1}^2} \mathbf{1} \right]^T$$
(10a)

$$\boldsymbol{\omega}_{r+1}^{*} = \boldsymbol{\omega}_{r} \left(1 + i\boldsymbol{\xi}_{r} \right) \tag{10b}$$

The mode shapes ϕ and ϕ_{n+1} can be viewed as a MDOF generalization of Eq.2.78 and 2.76. respectively, which are for a SDOF secondary subsystem.

5.4.1.2 Singly tuned modes

Assume that the first secondary subsystem mode is tuned to the primary subsystem. The expressions for the frequencies $\omega_i^{(0)}$ given in Eqs.4.46a,b are not changed by the duality relationships due to the symmetry of the singly tuned modes.

$$\omega_1^{(0)} = \omega_a \left\{ 1 + i\xi_a - \frac{1}{2} [\gamma_{11} + (i\xi_a + \beta)^2] \right\}$$
(11a)

$$\omega_{2}^{(0)} = \omega_{a} \left\{ 1 + i\xi_{a} + \frac{1}{2} [\gamma_{11} + (i\xi_{d} + \beta)^{2}] \right\}$$
(11b)

The first-order solutions to the mode shapes are found from Eq.4.47 using the results of section 4.4.1.3

$$\boldsymbol{\phi}_{1}^{(4)} = \left[\alpha_{1}^{(m-4)} - \frac{f_{42}}{g_{2}} + \cdots - \frac{f_{4n}}{g_{n}} \mathbf{1} \right]^{T}$$
(12)

where $\alpha^{(0)}$ is given by Eqs.4.46c, f and the above polynomials are evaluated at the vequency $\omega^{(0)}$. Note that the reciprocal of $\alpha^{(0)}$ is used here since the roles of the secondary and primary degrees-of-freedom are reversed. From these solutions, the following low-order approximations are obtained

$$\boldsymbol{\omega} = \boldsymbol{\omega}^{(1)} \tag{13a}$$

$$\boldsymbol{\phi}^{*} = \left[\alpha^{(0)-1} \frac{\boldsymbol{\zeta}_{12} \boldsymbol{\omega}_{22}^{*}}{\boldsymbol{\omega}_{22}^{*} - \boldsymbol{\omega}_{p1}^{*}} \cdots \frac{\boldsymbol{\zeta}_{1n} \boldsymbol{\omega}_{2n}^{*}}{\boldsymbol{\omega}_{2n}^{*} - \boldsymbol{\omega}_{p1}^{*}} \mathbf{1} \right]^{T}$$
(13b)

5.4.1.3 Multiply Tuned Modes

Assume the first *l* secondary subsystem modes are funed to the primary subsystem. The first-order approximations for the mode shapes for a multiply funed system are written in the same form as in Eq.4.51a

$$\boldsymbol{\phi}^{(1)} = \begin{bmatrix} \boldsymbol{\phi}_{d}^{(0)} \\ \boldsymbol{\phi}_{d}^{(0)} \\ 1 \end{bmatrix}$$
(14)

The tuned components $\phi_{i,i}^{(0)}$ and the frequency $\omega_{i}^{(0)}$ are found by solving the $l+1\times l+1$ eigenvalue problem in Eq.4.49. The detuned component $\phi_{i,i}^{(0)}$ is found from the dual of Eq.4.51b

$$\boldsymbol{\phi}_{ij}^{(0)} = \left\{ -\frac{f_{1,l+1}}{g_{l+1}} \cdots - \frac{f_{1n}}{g_n} \right\}^{T}$$
(15)

which, to low-order, reduces to

$$\boldsymbol{\phi}_{ik}^{*} = \begin{bmatrix} \boldsymbol{\zeta}_{i,k+1} \boldsymbol{\omega}_{i+1}^{*} & \cdots & \boldsymbol{\zeta}_{ik} \boldsymbol{\omega}_{ik}^{*} \\ \boldsymbol{\omega}_{i}^{*} & (i+1) \boldsymbol{\omega}_{ik}^{*} & \cdots & \boldsymbol{\omega}_{ik}^{*} \\ \boldsymbol{\omega}_{ik}^{*} & \boldsymbol{\omega}_{ik}^{*} \\ \boldsymbol{\omega}_{ik}^{*} & \boldsymbol{\omega}_{ik}^{*} \end{bmatrix}$$
(16)

Note that the detuned components of the modes shapes are identical for multiply tuned, singly tuned, and detuned systems.

5.4.1.4 Very Closely Spaced Detuned Modes

It was shown in section 4.4.2.5 that the expressions for the mode shapes for widely spaced detuned modes can be used in obtaining results for the system response even for very clos is spaced detuned modes. This fact continues to hold for SDOF/MDOF systems. If accurate

expressions for the mode shapes are required, then the same set of equations that were used in the multir – tuning problem are solved, as explained in Section 4.4.1.5.

5.4.1.5 Examples

1

The complex modal properties of the example system in Fig.5.1 sere computed using the formulations developed in this section and was compared with exact results. The frequencies are shown in Table 3 and are plotted in Figs.5.5a,b for various values of the equipment frequency $\omega_{x,x}$ and the mode shapes for the multiply tuned case (i.e., $\omega_{x,y} = 1.00$ rad/sec) are shown in Table 4. The non-classical damping character of the multiply tuned system is apparent in the mode shapes, which have imaginary components. Good agreement between approximate and exact values is found in all cases.

The effect of the equipment mass is illustrated in Fig.5.5b, where the frequencies corresponding to $\epsilon = .01, .005$, and .001 are represented by points A, B, and C, respectively, and are shown to be convergent to the subsystem natural frequencies.

5.4.2 Spectral Moments

As stated in the previous chapter the spectral moments can be calculated by evaluating the expressions for the mode shapes and frequencies and substituting the numerical results directly into the modal decomposition method. For the example system in Fig.5.1, the results of this numerical computation are compared with exact results obtained by integrating the complex frequency response function in Figs.5.ba-c. Also, the mean zero-crossing rate, ν_s and the shape factor, δ_s were computed and plotted in Figs.5.7a,b. All plots show good comparison between exact and the proposed approximate results.

It was noted in Section 4.4.2 that the mean zero-crossing rate of a funce PS system was nearly equal to $\nu = \omega_{\mu\nu}/\pi$, where $\omega_{\mu\nu}$ is one of the tuned frequencies. Since the secondary subsystem is always tuned to the primary subsystem, ν would have a constant value of approximately $1.0/\pi \approx 0.318$ rad/sec, which is in agreement with Fig.5.7a. Also, it was observed in Section 5.3 that the complex frequency response function had only a single peak for both singly and multiply tuned configurations of the example system, thus the shape factor would remain small and nearly constant for all values of the secondary frequency ω_{-1} , as shown in Fig.5.7b.

Although the numerical approach to finding the spectral moments is useful in practical applications, it is instructive to derive general algebraic solutions for the spect. I momen's of response of arbitrary SDOE/MDOF PS systems to white-noise input. This is accomplished using the duality relationships and the results in section 4.4.2.

5.4.2.1 Totally Detuned Systems with Well-Spaced Modes

As before, the modal decomposition method simplifies to the SRSS method for totally detuned systems with well-spaced modes. The effective participation factors are found from Eqs. 4.53a.b

$$\psi = q \left[\frac{r}{\omega_{i} - \omega_{i}} + r \right] \qquad i=1, \cdots, n$$
(17a)

$$\boldsymbol{\psi}_{n+1} = \boldsymbol{r}_{n+1} \left[\sum_{j=1}^{n} \frac{\boldsymbol{q} \, \boldsymbol{\omega}_{i}^{2} \boldsymbol{\zeta}_{j}}{\boldsymbol{\omega}_{i}^{2} - \boldsymbol{\omega}_{i}^{2}} + \boldsymbol{q}_{n+1} \right]$$
(17b)

It follows that the spectral moments for response to white-noise are

$$\lambda_{n} = \frac{\pi G_{n}}{4} \left\{ \sum_{j=1}^{n} q^{2} \left[\frac{r_{i+1} \omega_{j}^{2} \zeta_{1}}{\omega_{p1}^{2} - \omega_{j}^{2}} + r \right]^{2} \frac{\omega_{n}^{m-3}}{\xi_{n}} + r_{n+1}^{2} \left[\sum_{j=1}^{n} \frac{q_{i} \omega_{j}^{2} \zeta_{1}}{\omega_{j}^{2} - \omega_{p1}^{2}} + q_{n+1} \right]^{2} \frac{\omega_{p1}^{m-3}}{\xi_{p1}} \right\}$$
(18)

5.4.2.2 Singly T ined Systems

-

Assuming the first secondary mode is tuned to the primary mode, the defuned moment corresponding to Eq.4.56 is

$$\boldsymbol{\lambda}_{abc} = \frac{\pi G_{cc}}{4} \sum_{\boldsymbol{c} \in I_{a}} \boldsymbol{q}^{2} \left[\frac{r_{cc} (\boldsymbol{\omega}_{c}^{2} \boldsymbol{\zeta})}{\boldsymbol{\omega}_{c}^{2} + \boldsymbol{\omega}_{c}^{2}} + r \right]^{2} \frac{\boldsymbol{\omega}_{c}^{n-s}}{\boldsymbol{\xi}_{s}}$$
(19)

and the tuned moment corresponding to Ec.4.57 is

$$\lambda_{im} = \frac{\pi G_{ic}}{8} \frac{q_1 r_{a+1} \zeta_{11}^2 \xi_{a} \omega_{a}^{m-3}}{\xi_{\mu 1} \xi_{s 1} (\beta^2 + 4\xi_{a}^2) - \gamma_{11} \xi_{a}^2}$$
(20)

5.4.2.3 Multiply Tuned Systems

The procedure for finding λ_{μ} for multiply tuned systems remains the same as in Section

4.4.2.3. The defuned component λ_{abc} is computed using Eq.19 and λ_{abc} is evaluated numerically by using the modal decomposition method

5.4.2.4 Systems with Closely Spaced Secondary Modes

The cross modal contribution $\lambda_{1,1}$ from closely spaced secondary subsystem modes is accounted for in the same manner as outlined in section 4.4.2.4. The expression for $\lambda_{1,1}$ is given by Eq.4.59.

5.4.2.5 Systems with Closely Spaced Detuned Secondary Modes

The two methods for considering closely spaced defuned modes derived in Section 4.4.2.5 can be applied to SDOF/MDOF systems. In the first method, the closely spaced modes are considered as a single collective mode. Assuming the first *l* secondary subsystem modes are very closely spaced, the frequency for this collective mode is given by the average ω_i of the *l* secondary frequencies, the damping ratio is given by the average \mathcal{E}_i of the *l* modal damping ratios, and the effective participation factor ψ_i is given by

$$\psi_{\ell} = \sum_{i=1}^{l} q \left[\frac{r_{i,i}(\omega(\zeta))}{\omega_{\ell}^{i} - \omega_{\ell}^{i}} + r \right]$$
(21)

The alternative method is to consider the *l* modes separately. The mode shapes and frequencies are found from the formulae for widely spaced detuned modes, Eqs.8a.b; the effective participation factors are evaluated using Eq.17a; and the effect of close spacing is accounted for by using the approximation Eq.4.66 for the cross-spectral moments.

5.5 Non-Interaction Results

5.5.1 Introduction

Following the methods developed in Chapter 4, the SDOF/MDOF system is reanalyzed without accounting for interaction. The frequency response function and modal decomposition approaches are used and closed form results are obtained which are simpler than the corresponding expressions in the preceding sections.

5.5.2 Frequency Response Function Approach

The frequency response function for the zesponse quantity y(r) is given by the dual form of Eqs.4.72a-c

$$H^{(n+1)}(\omega) \approx \frac{G_1 \prod_{g} g}{d^{(n+1)}(\omega)} \left[\sum_{j=1}^{n} \frac{q_{j+1} m_g r f_j}{g G_1} - \sum_{j=1}^{n} \frac{q_j m_g r}{g} - \frac{q_{j+1} m_{j+1} r_{j+1}}{G_1} \right]$$
(224)
= $\sum_{j=1}^{n} \frac{q_{j+1} m_g r f_j}{g G_1} - \sum_{j=1}^{n} \frac{q_j m_g r}{g} - \frac{q_{j+1} m_{pj} r_{n+j}}{G_j}$

where

$$d^{(\prime\prime\prime)}(\omega) = G^{\prime}(\omega) ||g^{\prime}(\omega)|$$
(22b)

As before, the above expression for $H_1(\omega)$ can be compared with that in Eq.6a which included interaction; the relationship between the two results is given by

$$\lim_{\varepsilon_{\lambda} \to 0} H_{\varepsilon}(\omega) = H^{\varepsilon_{\lambda}(\varepsilon_{\lambda})}(\omega)$$
(23)

This indicates that the results previously obtained in this chapter can be converted to noninteraction results by taking the limit $m \rightarrow 0$.

For the example system in Fig.5.1, a comparison between the transfer function $T_{i_1}(\omega) = [H_i(\omega)]^{-1}$ for interaction and non-interaction analysis is shown in Fig.5.8 for various values of the secondary subsystem mass. As in Chapter 4, the differences are most notable for larger values of m_i and for ω near the tuned mode; for other values of ω , the transfer function is insensitive to interaction. A similar comparison is also made for the spectral moments of the system in Fig.5.9 with varying values for the mass and frequency of the secondary subsystem. As expected, the differences between λ_{i_1} and $\lambda_{i_2}^{(1)}$ are greatest at tuning and diminish at detuning.

5.5.3 Modal Decomposition Method

5.5.3.1 Introduction

Following the analysis in Section 4.5.3, closed form expressions are derived for the mode

shapes, frequencies and the factors a and c from the modal decomposition method. All of the results are obtained by duality.

5.5.3.2 Closed Form Expressions for the Mode Shapes

ı.

The frequencies and mode shapes associated with the secondary subsystem are given by the dual forms of Eqs.4.75 and 76:

$$\boldsymbol{\omega}^{T} = \boldsymbol{\omega}^{T} \left(\sqrt{1 - \boldsymbol{\xi}_{\lambda}^{T}} + i\boldsymbol{\xi}_{\lambda}^{T} \right) \quad i = 1, \dots, n$$

$$\boldsymbol{\phi}^{T} = \begin{bmatrix} 0 \cdots 0 - \frac{G_{1}(\boldsymbol{\omega}^{T})}{f_{1}^{T}(\boldsymbol{\omega}^{T})} & 0 \cdots & 0 \end{bmatrix}^{T}$$

$$\approx \begin{bmatrix} 0 \cdots 0 \frac{2(\beta_{1} + i\boldsymbol{\xi}_{d+1})\boldsymbol{\omega}_{d+1}^{T}}{\boldsymbol{\epsilon}_{1}^{T} \boldsymbol{\zeta}_{1}^{T} \boldsymbol{\omega}_{d}^{T}} & 0 \cdots & 0 \end{bmatrix}^{T}$$
(24a)
$$(24a)$$

where the first non-zero term is at the /th coordinate

Similarly, the corresponding expressions associated with the primary subsystem are given by the dual forms of Eqs.4.78 and 4.79:

$$\boldsymbol{\omega}_{i+1}^{T} = \boldsymbol{\omega}_{i}^{T} \left(\sqrt{1 - \boldsymbol{\xi}_{i+1}^{T}} + i \boldsymbol{\xi}_{i+1} \right)$$

$$(25a)$$

$$\int_{-1}^{T} \left(\boldsymbol{\omega}_{i+1}^{T} \right) = \int_{-1}^{T} \left(\boldsymbol{\omega}_{i+$$

$$\boldsymbol{\phi}_{1,1}^{\prime} = \begin{bmatrix} -\frac{\boldsymbol{\chi}_{11}^{\prime} \boldsymbol{\omega}_{11}^{\prime} \boldsymbol{\chi}_{11}^{\prime}}{\boldsymbol{\chi}_{1}^{\prime} (\boldsymbol{\omega}_{1,1}^{\prime})} & -\frac{\boldsymbol{\chi}_{11}^{\prime} \boldsymbol{\omega}_{11}^{\prime} \boldsymbol{\chi}_{11}^{\prime}}{\boldsymbol{g}_{1}^{\prime} (\boldsymbol{\omega}_{1,1}^{\prime})} & 1 \end{bmatrix}$$

$$= \begin{bmatrix} -\boldsymbol{\zeta}_{11}^{\prime} \boldsymbol{\omega}_{11}^{\prime} & -\frac{-\boldsymbol{\zeta}_{11}^{\prime} \boldsymbol{\omega}_{11}^{\prime}}{2(\boldsymbol{\beta}_{11}^{\prime} + \boldsymbol{\xi}_{11}^{\prime}) \boldsymbol{\omega}_{11}^{\prime}} & 1 \end{bmatrix}^{\prime}$$
(25b)

For defuned modes, these expressions reduce to the results in Section 5.4.1.1. For the singly luned modes, it can be shown that the above expressions are equivalent to the expressions derived in Section 5.4.1.2 in the limit $m_s \rightarrow 0$.

5.5.3.3 Spectral Moments

The key factors a_i and c_j which are used in the modal decomposition method are found simply by applying the duality relationships to Eqs. 4.81a-d:

$$a_{i} \approx 2\omega_{N} \mathrm{Im} b_{i} = \frac{q_{i}\xi_{1,i}r_{n+1}\beta_{1,i}\omega_{N}^{2}}{2(\beta_{1,i}^{2} + \xi_{n,1}^{2})\omega_{n,1}^{2}} + r_{i}q_{i}$$
(26a)

$$a_{n+1} \approx 2\omega_{p1} \text{Im} b_{n+1} = -\sum_{i=1}^{n} \frac{q_i \zeta_{1i} r_{n+1} \beta_{1i} \omega_{n}^2}{2(\beta_{1i}^2 + \xi_{n+1}^2) \omega_{n+1}^2} + r_{n+1} q_{n+1}$$
(26b)

$$c_{i} \approx 2\operatorname{Re}b_{i} = \frac{q_{i}\zeta_{1i}r_{0}+\xi_{d,1}\omega_{0}^{2}}{2\omega_{v}(\beta_{1}^{2}+\xi_{d,1}^{2})\omega_{d,1}^{2}}$$
(26c)

$$c_{n+1} \approx 2\mathbf{R}c\,b_{n+1} = -\sum_{i=1}^{n} \frac{\omega_n}{\omega_{n+1}}c_i$$
(26d)

which can be subsequently substituted into Eqs.4.82a,b to obtain the spectral moments λ_{ij} . The moments calculated by the above expressions are compared with exact results for various values of the secondary masses in Fig.5.10. The results are similar to those observed in Fig.5.9.

Table 5.1. Physical Properties of the SDOF/MDOF Example System

| Subsystem | Parameter Relationships | | | |
|-----------|---|--|--|--|
| Primary | $k_1 = (1.02)^2 m_1 (radians/sec)^2$ | | | |
| Secondary | $k_1, k_2 = (variable)$ | | | |
| | $m_2 = \epsilon m_1$ (ϵ variable) | | | |

Table 5.2. Modal Properties of the Fixed Base Subsystems

| Subsystem | Modal DOF | Frequency (rad/s) | Damping Ratio | |
|-----------|--------------|----------------------|------------------|--|
| Primary 1 | 1 | 1.02 | 0.02 | |
| Secondary | 2 | <i>ν</i> υ, 1 | 0.03 | |
| | 3 | 1.0 | 0.01 | |

| Tab | le | 5.3 | Frequencie | s for | Example | System | (e ₁₁ =0.01) |
|-----|----|-----|------------|-------|---------|--------|-------------------------|
|-----|----|-----|------------|-------|---------|--------|-------------------------|

| 6.1 | Mode | Exact Frequency | | Computed Frequency | | Error |
|-----------|------|-----------------|------------|--------------------|------------|-------|
| (rad/sec) | | Real Part | Imag. Part | Real Part | Imag. Part | ₩ |
| 0.70 | 1 | 0.698 | 0.020 | 0.700 | 0.021 | 0.3 |
| | 2 | 0.991 | 0.011 | 0.990 | 0.012 | 0.1 |
| | 3 | 1.030 | 0.018 | 1.029 | 0.017 | 0.1 |
| 1.00 | 1 | 0.992 | 0.016 | 0.990 | 0.018 | 0.3 |
| | 2 | 0.992 | 0.022 | 0.994 | 0.020 | 0.3 |
| | 3 | 1.035 | 0.020 | 1.033 | 0.021 | 6.1 |

| Mode | DOF | Exact Mode Shape | | Computed | Error | |
|------|-----|------------------|-----------|-----------|------------|------|
| | | Real Part | Imeg Part | Real Part | Imag. Part | 1¥0 |
| 1 | 1 | 0.006 | 0.000 | 0.003 | 0.000 | 0.3 |
| | 2 | 1.060 | 0.000 | 1.000 | 0.000 | 0.0 |
| | 3 | 0.003 | 0.006 | 0.000 | 0.000 | 0.3 |
| 2 | 1 | 0.050 | -0.012 | 0.054 | -0.014 | 0.5 |
| 1 | 2 | -0.034 | 0.006 | 0.005 | 0.001 | 3.2 |
| { | 3 | 1,000 | 0.000 | 1.000 | 0.000 | 0.0 |
| 3 | 1 | -0.185 | -0.048 | -0.172 | -0.045 | 0.6 |
| Í | 2 | 0.197 | 0.031 | 0.017 | 0.004 | 10.3 |
| | 3 | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |

Table 5.4 Mode Shapes of Example System ($\epsilon_{11}=0.01$, $\omega_{12}=1.0$)





Primary Subsystem

Secondary Subsystem



Multiply Tuned System, wsi=1.0 rad/sec



Singly Tuned System, w_{s1}=0.7 rad/sec

Fig.5.2. Distribution of Subsystem Free Vibration Frequencies

- 166 -




-





....







.















- 173 -



Fig5.6a. Spectral Moment $\lambda_0,~\varepsilon_{11}{=}0.01$



Fig5.6b. Spectral Moment A₁, e₁₁=0.01



Fig5.6c. Spectral Moment A2, e11=0.01

•







Fig5.7b. Shape Factor å, e₁₁-0.01







۸0







CHAPTER 6

ANALYSIS OF MULTI-DEGREE-OF-FREEDOM SECONDARY SYSTEMS ATTACHED TO MULTI-DEGREE-OF-FREEDOM PRIMARY SYSTEMS

6.1 Introduction

In this chapter, general PS systems consisting of an MDOF primary subsystem supporting an MDOF secondary subsystem in an arbitrary configuration are studied. Figure 6.1 provides a simple example of such a system. It will be shown that the expressions derived for MDOF/MDOF systems are combinations of the corresponding expressions for MDOF/SDOF and SDOF/MDOF systems analyzed in chapters 4 and 5, respectively.

The steps of the analysis will follow the same methodology used in chapter 4. Most of the results are new; previous research has not accounted for all of the dynamic properties for this general system, as was discussed in detail in the Introduction.

6.2 Definitions

6.2.1 Parameters

The matrices describing the MDOF/MDOF system are a generalization of those developed for the MDOF/SDOF system in Chapter 4. For the individual fixed base subsystems, the primary matrices $\mathbf{K}_{\rho\rho}$, $\mathbf{C}_{\rho\rho}$, and $\mathbf{M}_{\rho\rho}$ remain as before and the secondary matrices $\mathbf{M}_{\mu\nu}$, $\mathbf{C}_{\mu\nu}$, and $\mathbf{M}_{\rho\rho}$ remain as before and the secondary matrices $\mathbf{M}_{\mu\nu}$, $\mathbf{C}_{\mu\nu}$, and $\mathbf{M}_{\mu\nu}$ remain as before and the secondary matrices $\mathbf{M}_{\mu\nu}$, $\mathbf{C}_{\mu\nu}$, and $\mathbf{M}_{\mu\nu}$ remain as before and the secondary matrices $\mathbf{M}_{\mu\nu}$, $\mathbf{C}_{\mu\nu}$, and $\mathbf{M}_{\mu\nu}$ are of order $M \times M$, where M is the number of degrees of freedom of the secondary subsystem.

The displacements of the combined system are defined by the N+M vector

 $\mathbf{x}_0 = [\mathbf{x}_1 \cdots \mathbf{x}_N | \mathbf{x}_{N+1} \cdots \mathbf{x}_{N+M}]^T$ (1)

and the $N+M\times N+M$ mass, damping, and stiffness matrices M_{uv} , C_{uv} , and K_{uv} , respectively, are given by Eq.4.7. As noted in Chapter 4, these matrices are not precisely equal to the true

system matrices, however the approximation is consistent with the analysis that will be performed.

From these matrices, the equations of motion for the MDOF/MDOF system are given by Eq.4.8

$$\mathbf{M}_{xx}\ddot{\mathbf{x}}_{0} + \mathbf{C}_{xx}\dot{\mathbf{x}}_{0} + \mathbf{K}_{xx}\mathbf{x}_{0} = -\mathbf{M}_{xx}\mathbf{r}_{0}\ddot{\mathbf{x}}_{\nu}(t)$$
⁽²⁾

This equation can be transformed to modal coordinates by means of the $N+M\times n+m$ transformation matrix given by Eq.4.11a

$$\mathbf{\Phi} = \begin{bmatrix} \mathbf{\Phi}_{\mu} & \mathbf{0} \\ \mathbf{0} & \mathbf{\Phi}_{\lambda} \end{bmatrix}$$
(3)

where Φ_{λ} is an $M \times m$ matrix consisting of the *m* mode shapes of interest for the fixed base secondary subsystem. The modal coordinate vector z is obtained through the relation $\Phi z = x_0$ and is given by

$$\boldsymbol{z} = [\boldsymbol{z}_{p} \quad \dots \quad \boldsymbol{z}_{pn} \quad \boldsymbol{z}_{v1} \quad \dots \quad \boldsymbol{z}_{vn}] \tag{4}$$

The transformed mass, damping, and stiffness matrices are generalizations of Eqs.4.12a-d:

$$\mathbf{M} = \operatorname{diag}\left[m_{p1} \cdots m_{pn} \ m_{s1} \cdots m_{sn}\right]$$
(5a)

$$\mathbf{C} = \begin{vmatrix} 2m_{\mu_{1}}\omega_{\mu_{1}}\xi_{\mu_{1}} & (sym) \\ 0 & \cdots & 2m_{\mu_{n}}\omega_{\mu_{n}}\xi_{\mu_{n}} \\ c_{11} & \cdots & c_{n1} & 2m_{s1}\omega_{s1}\xi_{s1} \\ \cdots & \cdots & \cdots & \cdots \\ c_{1m} & \cdots & c_{nm} & 0 & \cdots & 2m_{sm}\omega_{sm}\xi_{sm} \end{vmatrix}$$
(5b)
$$\mathbf{K} = \begin{vmatrix} m_{\mu_{1}}\omega_{\mu_{1}}^{2} & (sym) \\ 0 & \cdots & m_{\mu_{n}}\omega_{\mu_{n}}^{2} \\ 0 & \cdots & m_{\mu_{n}}\omega_{\mu_{n}}^{2} \\ k_{11} & \cdots & k_{n1} & m_{s1}\omega_{s1}^{2} \\ \cdots & \cdots & \cdots \\ k_{1m} & \cdots & k_{nm} & 0 & \cdots & m_{sm}\omega_{sm}^{2} \end{vmatrix}$$
(5c)

where

$$\mathbf{c}_{ij} = \mathbf{\phi}_{\rho}^{T} \mathbf{C}_{\rho} \mathbf{\phi}_{xj} \quad k_{ij} = \mathbf{\phi}_{\rho}^{T} \mathbf{K}_{\rho x} \mathbf{\phi}_{xj}$$
(5d)

The equations of motion in terms of the modal coordinates are given by Eq.4.13

$$\mathbf{M}\mathbf{\ddot{z}} + \mathbf{C}\mathbf{\dot{z}} + \mathbf{K}\mathbf{z} - -\mathbf{M}\mathbf{r}\mathbf{\ddot{x}}_{\mathbf{x}}(t) \tag{6}$$

where r is the vector of modal participation factors given by Eq.4.14. The form of the matrices

M, C, and K are more complicated than that for MDOF/SDOF systems, however, this does not pose any problems in the succeeding analysis.

To facilitate the analysis of the system, the terms k_{i1} and c_{i1} are rewritten in a form similar to Eq.4.17

$$k_{ij} = -m_{ij}\omega_{ij}^2\xi_{ij} - c_{ij} \approx -2m_{ij}\omega_{ij}\xi_{ij}\xi_{ij}$$
⁽⁷⁾

where ζ_{ij} is defined to be the ratio

$$\zeta_{ij} = -\frac{k_{ij}}{m_{ij}\omega_{ij}^2} \tag{8}$$

The interpretation of $\zeta_{i,j}$ is a generalization of that given in Section 4.2.1: It is the displacement of the *j*th mode of the secondary subsystem caused by a static unit displacement of the *i*th mode of the primary subsystem.

6.2.2 Notation

The classification of the modes of the MDOF/MDOF system is a generalization of the classification presented in Chapter 4. The definition of tuning remains as before: A primary mode *i* and a secondary mode *j* are considered to be tuned if

$$\left(\frac{\omega_{p_{i}}-\omega_{y_{i}}}{\omega_{a,y}}\right)^{2} < \frac{\xi_{a,y}^{2}}{e} \left[4 + \frac{\gamma_{y_{i}}}{\xi_{y_{i}}\xi_{p_{i}}}\right]$$
(9)

A primary mode which is not tuned to any of the secondary modes is defined to be detuned; the same definition applies to secondary modes.

There are two sets of detuned modes, one for each subsystem:

$$I_{pi} = \begin{cases} \text{set of indecies } i \text{ corresponding to} \\ \text{detuned primary subsystem modes} \end{cases}$$
(10a)
$$I_{pi} = \begin{cases} \text{set of indecies } i \text{ corresponding to} \\ \text{detuned secondary subsystem modes} \end{cases}$$
(10b)

Tuned modes fall into two classifications, as discussed in Section 4.2.2: singly tuned modes and multiply tuned modes. The singly tuned modes form a set of ordered pairs

$$J_{i1} = \left\{ (i,j) : \text{ primary mode } i \text{ is singly tuned to secondary mode } j \right\}$$
(11)

Multiply tuned modes form a collection of sets. Assume there are a total of K clusters of

multiply tuned modes. Then for each $k \leq K$

.

$$I_{\mu 2}^{(k)} = \left\{ \text{ primary modes in the } k-\text{th multiply tuned cluster} \right\}$$
(12a)

$$I_{u2}^{(4)} =$$
 secondary modes in the k-th multiply tuned cluster (12b)

As an example, if the first k primary modes are all tuned to the first l secondary modes, then

$$I_{\mu\nu}^{(1)} = \{1, \dots, k\} = I_{\nu}^{(1)} = \{1, \dots, \ell\}$$
(13)

It should be noted that each primary or secondary mode is included in one and only one of the above sets.

6.2.3 Example System

...

To illustrate the major characteristics of MDOF/MDOF systems and check the accuracy of the formulations derived in this chapter, the example system shown in Fig.6.1 is used. The primary subsystem is identical to the one used in the example system for MDOF/SDOF systems and the secondary subsystem is similar to that used in the example system for SDOF/MDOF systems.

The dynamic properties of the subsystems are described in Tables 1 and 2 and are chosen so that the combined system would exhibit important characteristics found in general MDOF/MDOF systems. For instance, the frequency of the equipment is a variable parameter, which allows for an investigation of tuning. For $\omega_s = .38$ rad/sec, the system is singly tuned as shown in Fig.6.2a, and for $\omega_s = 1.0$ rad/sec, the system becomes multiply tuned as shown in Fig.6.2b. The mass ratio, ϵ_s is also chosen to be a variable parameter, as before. Finally, the damping ratio of the equipment is unequal to the damping ratios of the primary subsystem, thus the combined system is, in general, non-proportionally damped.

The response quantity that will be investigated is the relative displacement between the upper mass of primary subsystem 2 and the adjacent mass of the secondary subsystem.

6.3 Frequency Response Function Approach

6.3.1 The Complex Frequency Response Matrix

The complex frequency response matrix for the MDOF/MDOF system is a generalization of Eq.4.28a and Eq.5.3a

$$\mathbf{H}(\boldsymbol{\omega}) = \begin{bmatrix} G_1(\boldsymbol{\omega}) & (sym) & -1 \\ 0 & \cdots & G_n(\boldsymbol{\omega}) \\ f_{\pm 1}(\boldsymbol{\omega}) & \cdots & f_{n\pm}(\boldsymbol{\omega}) & g_{\pm}(\boldsymbol{\omega}) \\ \cdots & \cdots & \cdots & \cdots \\ f_{\pm n}(\boldsymbol{\omega}) & \cdots & f_{nm}(\boldsymbol{\omega}) & \Theta & \cdots & g_n(\boldsymbol{\omega}) \end{bmatrix}$$
(14a)

The polynomials $G_i(\omega)$ and $g_i(\omega)$ are given by Eqs.4.28c and 5.3b, respectively, and

$$f_{j}(\omega) \approx -\zeta_{ij} m_{ij}(\omega_{ij}^{2} + 2i\omega_{ij}\xi_{ij}\omega)$$
(14b)

for $i=1, \cdots, n$ and $j=1, \cdots, m$.

Unlike the previous chapters, the above inverse has no closed form solution. To find the complex frequency response function $H_1(\omega)$ for a response quantity $y(t) - q^T z(t)$, the equation Eq.4.32 must be solved numerically, which requires the reduction of an n+m order system of equations for each value of the frequency ω . This is considerably more difficult than the evaluation of a rational polynomial, which is the computation required for $H_1(\omega)$ in the MDOF/SDOF and SDOF/MDOF systems. The modal decomposition method to be presented in the succeeding sections does not present such numerical difficulties and is the recommended method for analyzing MDOF/MDOF systems.

6.4 Modal Decomposition Approrch

6.4.1 Mode Shapes and Frequencies

The low-order approximations for the mode shapes and frequencies will be derived for the MDOF/MDOF system. The approach is similar to that of Section 4.4.1. The resulting expressions are a combination of those derived for MDOF/SDOF and SDOF/MDOF systems in Sections 4.4.1 and 5.4.1, respectively. These expressions can be evaluated numerically and used in the modal decomposition analysis to provide accurate measures of the response.

As before, approximate algebraic expressions for the spectral moments for response to white-noise input are obtained to investigate further the dynamic behavior of the system.

6.4.1.1 Detwned Modes

Assume that primary mode 1 is the detuned mode to be analyzed. Following the procedure in 4.4.1.1, the first-order solutions for the frequency and mode shape can be found

$$\omega_1^{(0)} = \omega_{\rho 1} (\sqrt{1 - \xi_{\rho 1}^2} + i \xi_{\rho 1})$$
(15a)

$$\Phi_1^{(1)} = \left[1 \sum_{j=1}^m \frac{f_{1j} f_{2j}}{g_j G_2} \cdots \sum_{j=1}^m \frac{f_{1j} f_{nj}}{g_j G_n} - \frac{f_{11}}{g_1} \cdots - \frac{f_{1m}}{g_m} \right]^T$$
(15b)

where the polynomials are evaluated at $\omega = \omega_1^{(0)}$. Note that the above expression for the mode shape is a combination of the results Eqs.4.41 and 5.9a. The low-order approximations are

$$\omega_1 = \omega_{p1} (1 + i\xi_{p1})$$
(16a)

$$\phi_1 = \left[1 \ 0 \ \cdots \ 0 \ \frac{\zeta_{11} \omega_{11}^2}{\omega_{11}^2 - \omega_{p1}^2} \ \cdots \ \frac{\zeta_{1m} \omega_{mn}^2}{\omega_{mn}^2 - \omega_{p1}^2} \right]^{T}$$
(16b)

Likewise, if secondary mode 1 is detuned, the first-order approximations for the mode shape and frequency are

$$\omega_{n+1}^{(0)} = \omega_{11}(\sqrt{1-\xi_{11}^2}+i\xi_{11})$$
(17a)

$$\Phi_{n+1}^{(1)} = \left[-\frac{f_{11}}{G_1} \cdots - \frac{f_{n1}}{G_n} \prod_{i=1}^n \frac{f_i(f_{i2})}{g_2 G_i} \cdots \prod_{i=1}^n \frac{f_{i1} f_{im}}{g_m G_i} \right]^T$$
(17b)

where the polynomials are evaluated at $\omega = \omega_{n+1}^{(0)}$. In this case, the above expression for the mode shape is a combination of Eqs.4.44 and 5.7b. The low-order approximation is

$$\phi_{n+1}^* = \left[\zeta_{11}\epsilon_{11}\frac{\omega_{n1}^2}{\omega_{n1}^2 - \omega_{n1}^2} \cdots \zeta_{n1}\epsilon_{n1}\frac{\omega_{n1}^2}{\omega_{nn}^2 - \omega_{n1}^2} + 0 \cdots 0\right]'$$
(18a)

$$\omega_{n+1} = \omega_{s1}(1 + i\xi_{s1})$$
(18b)

6.4.1.2 Singly Tuned Modes

Assume that primary subsystem mode 1 and secondary subsystem mode 1 are tuned to each other. Applying the analysis of section 4.4.1.3, the resulting first-order approximations for the frequencies are given by the same expressions as in Eqs.4.46a,b:

$$\boldsymbol{\omega}_{i}^{(0)} = \boldsymbol{\omega}_{a,11} \left\{ 1 + i \boldsymbol{\xi}_{a,11} \pm \frac{1}{2} [\boldsymbol{\gamma}_{11} + (i \boldsymbol{\xi}_{d,11} + \boldsymbol{\beta}_{11})^2]^{1} \right\}$$
(19)

The mode shapes $\phi^{(1)}$ are

$$\Phi_{1}^{(1)} = \left[\alpha_{1}^{(0)} - \frac{f_{11}}{G_{2}} \cdots - \frac{f_{n}}{G_{n}} \right] - \alpha_{1}^{(0)} \frac{f_{12}}{g_{2}} \cdots - \alpha_{n}^{(0)} \frac{f_{1n}}{g_{n}} \right]^{\prime}$$
(20a)

where the polynomials above are evaluated at $\omega = \omega_i^{(0)}$ and $\alpha_i^{(0)}$ are given by Eqs.4.46c, f

$$\alpha_{1}^{(i)} = \frac{1}{\zeta_{11}} \left\{ -\beta_{11} + i\xi_{d,11} \pm \frac{1}{2} [\gamma_{11} + (i\xi_{d,11} + \beta_{11})^{2}]^{2} \right\}$$
(20b)

The low-order approximations are

$$\phi^{*} = \left[\alpha^{(0)} \frac{\zeta_{21} \epsilon_{22} \omega_{31}^{2}}{\omega_{p2}^{2} - \omega_{31}^{2}} \cdots \frac{\zeta_{n1} \epsilon_{n1} \omega_{31}^{2}}{\omega_{p2}^{2} - \omega_{31}^{2}} + \alpha^{(0)} \frac{\zeta_{12} \omega_{32}^{2}}{\omega_{32}^{2} - \omega_{p1}^{2}} \cdots \alpha^{(0)} \frac{\zeta_{10} \omega_{30}^{2}}{\omega_{30}^{2} - \omega_{p1}^{2}} \right]^{\prime}$$
(21)

Note that ϕ_{i}^{+} is a combination of Eqs.4.48b and 5.13b.

6.4.1.3 Multiply Tuned Modes

Assume the first k primary subsystem modes are tuned to the first l secondary subsystem modes, i.e., $I_{u2}^{(1)}$ and $I_{u2}^{(1)}$ are given by Eqs.13. The matrix $\Gamma(\omega) = \mathbf{H}^{(1)}(\omega)$ can be partitioned into submatrices

$$\Gamma(\omega) = \begin{bmatrix} \Gamma_{pi} & 0 & \Gamma_{il}^{-1} & \Gamma_{il}^{-1} \\ 0 & \Gamma_{pi} & \Gamma_{il}^{-1} & \Gamma_{il}^{-1} \\ \Gamma_{ii} & \Gamma_{il} & \Gamma_{si} & 0 \\ \Gamma_{il} & \Gamma_{il} & 0 & \Gamma_{si} \end{bmatrix}$$
(22)

where the subscripts p, s, t, d refer to the primary subsystem, secondary subsystem, tuned modes, and detuned modes, respectively,

$$\Gamma_{p_k}(\omega) = \operatorname{diag}\{ G_1(\omega) \cdots G_k(\omega) \}$$
(23a)

$$\Gamma_{p,\ell}(\omega) = \operatorname{diag} \{ G_{k+1}(\omega) \cdots G_n(\omega) \}$$
(23b)

$$\Gamma_{ii}(\omega) = \operatorname{diag}\{g_1(\omega) + \cdots + g_i(\omega)\} =$$
(23c)

$$\Gamma_{yl}(\omega) = \operatorname{diag}\{g_{l+1}(\omega) \otimes \cdots \otimes g_{ln}(\omega)\}$$
(23d)

where diag $\{\cdots\}$ denotes diagonal matrices, and Σ_{μ} , Γ_{μ} , Γ_{μ} , and Γ_{μ} are full matrices.

Following the analysis in section 4.4.1.4, the initial approximations for the mode shapes are given by

$$\boldsymbol{\phi}_{i}^{(0)} = \left[\boldsymbol{\phi}_{iji}^{(0)} \ 0 \ \boldsymbol{\phi}_{isi}^{(0)} \ 0 \right]^{T}$$
(24)

where the vectors $\phi_{(p)}^{(0)}$ and $\phi_{(b)}^{(0)}$ and the initial approximations for the frequencies are obtained

from the $k+l \times k+l$ order eigenvalue problem

$$\begin{bmatrix} \boldsymbol{\Gamma}_{pr}(\boldsymbol{\omega}_{i}^{(0)}) & \boldsymbol{\Gamma}_{ll}(\boldsymbol{\omega}_{i}^{(0)})^{T} \\ \boldsymbol{\Gamma}_{ll}(\boldsymbol{\omega}_{i}^{(0)}) & \boldsymbol{\Gamma}_{sl}(\boldsymbol{\omega}_{i}^{(0)}) \end{bmatrix} \begin{bmatrix} \boldsymbol{\phi}_{pr}^{(0)} \\ \boldsymbol{\phi}_{tsl}^{(0)} \end{bmatrix} = \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$
(25)

This problem corresponds to Eq.4.49 and is relatively small compared to the order of the combined system $(n+m\times n+m)$. Referring back to the original matrix $\Gamma(\omega)$, the error is

$$\Gamma(\omega) \begin{bmatrix} \phi_{jp}^{(0)} \\ 0 \\ \phi_{jm}^{(0)} \\ 0 \end{bmatrix} = \begin{bmatrix} 0 \\ \Gamma_{nd}^{j\phi} \phi_{jm}^{(0)} \\ 0 \\ \Gamma_{dl} \phi_{jm}^{(0)} \end{bmatrix} = \begin{bmatrix} 0 \\ O(e) \\ 0 \\ O(e^2) \end{bmatrix}$$
(26)

As before, the above error is reduced by introducing detuned components to the mode shapes. The resulting approximation ϕ_1^+ which corresponds to Eqs.4.52a is

$$\phi_{i}^{*} \leftarrow \left[\phi_{ipi}^{(0)} + \phi_{ipi}^{(0)} + \phi_{ibi}^{(0)} + \phi_{ibi}^{(0)}\right]^{\prime}$$
(27a)

where

$$\boldsymbol{\phi}_{dp_{i}}^{(0)} = -\boldsymbol{\Gamma}_{p_{i}}^{-1} \boldsymbol{\Gamma}_{id}^{1} \boldsymbol{\phi}_{ij}^{(0)} \qquad \boldsymbol{\phi}_{db_{i}}^{(0)} = -\boldsymbol{\Gamma}_{id}^{-1} \boldsymbol{\Gamma}_{id} \boldsymbol{\phi}_{ip_{i}}^{(0)}$$
(27b)

Note that $\Gamma_{\mu\nu}$ and $\Gamma_{\mu\nu}$ are diagonal matrices and are easily inverted.

6.4.1.4 Closely Spaced Detuned Modes

In sections 4.4.2.5 and 5.4.2.5 it was noted that the derivation of the mode shapes for closely spaced detuned modes is unnecessary if the quantity of interest is the response of the secondary subsystem. The solutions for the mode shapes derived for widely spaced detuned modes can be used in obtaining results for the system response even if the detuned modes are closely spaced. This method will be applied to MDOF/MDOF systems.

If precise solutions for the detuned modes is required, then considerable amount of computation is required. The analysis of the mode shapes for k very closely spaced detuned primary subsystem modes requires the solution of an $k+m\times k+m$ order eigenvalue problem. Similarly, for *l* detuned very closely spaced secondary subsystem modes the problem is of order $l+n\times l+n$. Since *m* and *n* are large integers, these eigenvalue problems are relatively large.

6.4.1.5 Examples

The complex modal properties of the example system in Fig.6.1 were computed using the formulations developed in this section and were compared with exact results obtained by using a complex eigenvalue solver from the IMSL library. The frequencies are shown in Table 3 and are plotted in Figs.6.3a,b for various values of the equipment frequency ω_{s1} , and the mode shapes for the multiply tuned case (i.e., $\omega_{s1} = 1.00$ rad/sec) are shown in Table 4. The nonclassical damping character of the multiply tuned system is apparent in the mode shapes, which have imaginary components. Good agreement between approximate and exact values is found in all cases.

The effect of the equipment mass is illustrated in Fig.6.3b, where the frequencies corresponding to $\epsilon = .01, .005$, and .001 are represented by points A_i , B_i , and C_i , respectively. Mode 1 is not affected by the mass ratio because it is detuned. The frequencies corresponding to the other modes converge to the subsystem natural frequencies as was observed in the study of the 2-DOF system.

6.4.2 Spectral Moments

The spectral moments can be calculated using the preceding expressions for the modal properties and employing the modal decomposition method developed earlier. For the example system in Fig.6.1, such calculations are compared with exact results obtained by integrating the complex frequency response function in Figs.6.4a-c. Also, the mean zero-crossing rate, ν , and the shape factor, δ , were computed and plotted in Figs.6.5a,b. All plots show good comparison between exact and the proposed approximate results.

It was noted in Section 4.4.2 that the mean zero-crossing rate of a tuned PS system was nearly equal to $\nu = \omega_{\nu}/\pi$, where ω_{ν} is one of the tuned frequencies. For $\omega_{\nu 1} = .38$ rad/sec, it can be seen that ν approaches this theoretical value in Fig.6.5a (compare with Fig.4.8a for the MDOF/SDOF example study). For higher values of $\omega_{\nu 1}$, the tuned modes with frequencies at $\omega_{\nu 2} = 1.0$ rad/sec are dominant in the response motion, and ν has a nearly constant value of approximately $1.0/\pi \approx 0.318$ rad/sec (compare with Fig.5.7a for the SDOF/MDOF example study).

The shape factor is relatively large around $\omega_{si} = .58$ rad/sec. This is due to the fact that the power spectral density function has two peaks, corresponding to the two sets of tuned modes of the system. For other values of ω_{si} , there is only one peak at $\omega = 1.0$ rad/sec and the response process becomes narrow-banded with a small shape factor (see Fig.6.5b).

As in the previous chapters, general low-order algebraic expressions will be derived for the spectral moments for response to white-noise input. These results can be viewed as a combination of the results in Chapters 4 and 5.

6.4.2.1 Well-Spaced Detuned Modes

The treatment of well-spaced detuned modes is essentially the same as in Sections 4.4.2.1 and 5.4.2.1. Let mode *i* and mode n+j correspond to detuned primary and secondary subsystem modes, respectively. The effective participation factors are

$$\boldsymbol{\psi}_{i} = r_{i} \left[\sum_{j=1}^{m} \frac{\boldsymbol{q}_{ij+j} \boldsymbol{\omega}_{ij}^{2} \boldsymbol{\zeta}_{ij}}{\boldsymbol{\omega}_{ij}^{2} - \boldsymbol{\omega}_{ij}^{2}} + \boldsymbol{q}_{i} \right]$$
(28a)

for the primary mode and

$$\psi_{n+j} = q_{n+j} \left[\sum_{i=1}^{n} \frac{r_i \omega_{sj}^2 \zeta_{ij}}{\omega_{pi}^2 - \omega_{sj}^2} + r_{n+j} \right]$$
(28b)

for the secondary mode. Note that the former expression is similar to Eq.5.17b and the latter is similar to Eq.4.53b, which is expected. The response contributed from the detuned modes is foun.' from Eq.2.38, ignoring cross-modal terms due to well-spacing of modes:

$$\lambda_{dm} = \frac{\pi G_0}{4} \left[\sum_{i=l_{pl}} r_i^2 \left[\sum_{j=1}^m \frac{q_{n+j} \omega_{ij}^2 \zeta_{ij}}{\omega_{ij}^2 - \omega_{pi}^2} + q_i \right]^2 \frac{\omega_{pi}^{m-3}}{\xi_{pi}} + \sum_{j \in l_{pl}} q_{n+j}^2 \left[\sum_{i=1}^n \frac{r_i \omega_{ij}^2 \zeta_{ij}}{\omega_{pi}^2 - \omega_{ij}^2} + r_{n+i} \right]^2 \frac{\omega_{ij}^{m-3}}{\xi_{ij}} \right]$$

6.4.2.2 Singly Tuned Modes

For each pair of singly tuned modes there is a tuned contribution to the spectral moment of the form of Eq.4.57. To clarify notation, assume primary mode i and secondary mode j are singly tuned and let

$$\omega_{\mu,ij} = \frac{\omega_{\mu} + \omega_{ij}}{2} \tag{30a}$$

$$\boldsymbol{\xi}_{a,a} = \frac{\boldsymbol{\xi}_{a} + \boldsymbol{\xi}_{sj}}{2} \tag{30b}$$

$$\beta_{ij} = \frac{\omega_{ij} - \omega_{pi}}{\omega_{ijj}}$$
(30c)

Then, the moment term arising from these two modes is

$$\lambda_{im}^{(i,j)} = \frac{\pi G_{gg}}{8} \frac{r_i q_{n+j} \zeta_{ij}^2 \xi_{a,ij} \omega_{a,g}^{m-3}}{\xi_{p_i} \xi_{s_j} (\beta_{ij}^2 + 4\xi_{a,ij}^2) + \gamma_{ij} \xi_{a,ij}^2}$$
(31)

The total contribution from all pairs of singly tuned modes is simply the sum

$$\boldsymbol{\lambda}_{i|m} = \sum_{i_{i},j_{i} \in J_{ij}} \boldsymbol{\lambda}_{jm}^{i_{i},j_{i}}$$
(32)

6.4.2.3 Multiply Tuned Modes

The spectral moments for multiply tuned modes are derived numerically by the modal decomposition method, as before. For each k-th cluster of multiply tuned modes a moment term $\lambda_{12m}^{(4)}$ is found, and, as in Eq.32 above, these moments are summed to find the total moment contribution

$$\lambda_{i2m} \sim \sum_{k=1}^{K} \lambda_{i2m}^{(k)}$$
(33)

6.4.2.4 Closely Spaced Modes

The correlation between closely spaced modes is accounted for in the same manner as detailed in section 4.4.2.4. The resulting expression for the total moment is

$$\lambda_m = \lambda_{dm} + \lambda_{11m} + \lambda_{12m} + \lambda_{cm}$$
(34)

6.4.2.5 Very Closely Spacel Modes

The two methods developed in the previous chapters for analyzing systems with closely spaced detuned modes can be applied directly to MDOF/MDOF systems. If the modes are to be considered separatedly, then the mode shapes and frequencies are found from the analysis of widely spaced detuned modes in Section 6.4.1.1. The effective participation factors are

subsequently evaluated using Eqs.28a or 28b, and the effect of close spacing is accounted for by the approximation Eq.4.66 for the cross spectral moments. If the set of very closely spaced modes are to be represented by one collective mode, the corresponding participation factor would be the sum of the participation factors in Eqs.28a or 28b and the frequency and damping ratio are given by their respective averages.

6.5 Non-Interaction Results

6.5.1 Introduction

The MDOF/MDOF system is reanalyzed without accounting for interaction in both the frequency response and the modal decomposition approaches. Unlike the outcome of Section 6.3, a simple closed form expression is available for the frequency response function $H_y(\omega)$ for the non-interaction case. Also, as in the previous chapters, closed form expressions are obtained for all mode shapes and the factors a_i and c_i which are used in the modal decomposition method.

6.5.2 Frequency Response Function Approach

The procedure developed in Section 3.5 is used to obtain the complex frequency response function $H_{\mu}(\omega)$. The equations of motion are decoupled into two sets of equations. The first set corresponds to the displacement response of the primary subsystem DOF to the base input and is given by

$$\dot{z}_{pi}^{(non)} + 2\omega_{pi}\xi_{pi}z_{pi}^{(noi)} + \omega_{pi}^{2}z_{pi}^{(non)} = -r_{i}\dot{x}_{k} \quad i=1,\cdots,n$$
(35)

and the remaining equations are for the resulting response of the secondary subsystem to the support motions,

$$\ddot{z}_{sj}^{(non)} + 2\omega_{sj}\xi_{sj}\dot{z}_{sj}^{(non)} + \omega_{sj}^{2}z_{sj}^{(non)} = \sum_{i=1}^{n} [\omega_{sj}^{2}z_{pi} + 2\xi_{si}\omega_{sj}\dot{z}_{pi}]\zeta_{ij} - r_{n+j}\dot{z}_{k}$$
(36)

The Fourier transforms $Z_{p_i}^{(nom)}(\omega)$ and $Z_{sj}^{(nom)}(\omega)$ of the subsystem responses $z_{p_i}^{(nom)}(t)$ and $z_{sj}^{(nom)}(t)$ are

$$Z_{\mu\nu}^{(non)}(\omega) = -\frac{X_{R}(\omega)m_{\mu}r_{i}}{G_{i}(\omega)} \qquad i=1,\cdots,n \qquad (37a)$$

$$Z_{sj}^{(non)}(\omega) = \frac{X_{\kappa}(\omega)}{g_{j}(\omega)} \left[\sum_{i=1}^{n} \frac{f_{ij}(\omega)m_{p_{i}}r_{i}}{G_{i}(\omega)} - m_{sj}r_{n+j} \right]$$
(37b)

From these expressions, the Fourier transform $Y^{(mat)}(\omega)$ of the response variable y(t) is obtained

$$Y^{(mon)}(\omega) = \sum_{i=1}^{n} q_i Z_{p_i}^{(mon)}(\omega) + \sum_{i=1}^{m} q_{n+i} Z_{n+i}^{(mon)}(\omega)$$

$$Y^{(mon)}(\omega) = X_{g_i}(\omega) \left\{ \sum_{i=1}^{n} \sum_{j=1}^{m} \frac{q_{n+i} f_{ij}(\omega) m_{p_i} r_i}{g_j(\omega) G_i(\omega)} - \sum_{i=1}^{n} \frac{q_i m_{p_i} r_i}{G_i(\omega)} - \sum_{i=1}^{m} \frac{q_{n+i} m_{n+i} r_{n+i}}{g_j(\omega)} \right\}$$
(38)

It follows that the frequency response function is

$$H_{i}^{(non)}(\omega) \approx \sum_{i=1}^{n} \sum_{j=1}^{n} \frac{q_{n+i} f_{ij}(\omega) m_{ij} r_{i}}{g_{j}(\omega) G_{i}(\omega)} - \sum_{j=1}^{n} \frac{q_{i} m_{jn} r_{i}}{G_{i}(\omega)} - \sum_{j=1}^{n} \frac{q_{n+j} m_{ij} r_{n+j}}{g_{j}(\omega)}$$
(39)

A comparison between the transfer function $T_{11}(\omega) = \{H_1(\omega)\}^2$ for interaction and noninteraction analysis is shown in Fig.6.6 for various values of the secondary system mass. The differences are most notable for values of ω near the tuned mode; the differences increases for larger secondary masses due to the increased effect of interaction. For other values of ω , the transfer function is insensitive to interaction. A similar comparison is made for non-interaction and interaction moments in Fig.6.7 for the same system with verying values for the mass and frequency of the secondary system. The result is similar to the findings in Chapter 3: the difference between λ_0 and $\lambda_0^{(non)}$ are greatest at tuning and diminish at detuning.

6.5.3 Modal Decomposition Method

6.5.3.1 Introduction

The approach taken here is essentially the same as in Section 4.5.3. Expressions for the mode shapes and frequencies are rederived for the non-interacting, combined system and the results substituted into the modal decomposition method of Chapter 2. Closed form expressions for all mode shapes and frequencies, including multiply tuned modes, and the factors a_i and c_i are obtained. The final results for the response of the system are easily obtainable from the expressions for a_i and c_i and the original parameters of the two subsystems.

6.5.3.2 Closed Form Expressions for the Mode Shapes

The original eigenvalue problem

$$\Gamma(\omega_{i})\phi_{i}=0 \tag{40}$$

is reinvestigated. First, the modes associated with the primary subsystem are analyzed. It is intuitively clear that the frequencies associated with these n modes are given by the original primary subsystem frequencies

$$\omega_{i}^{2} = \omega_{\mu\nu}(\sqrt{1 - \xi_{\mu\nu}^{2}} + i\xi_{\mu\nu}) \quad i = 1, \dots, n$$
(41)

The corresponding mode shapes are derived by substituting ω_i^* into Eq.40 and solving the eigenvalue problem. The solutions are

$$\phi_{i}^{*} = \begin{bmatrix} 0 & \cdots & 0 & 1 & 0 & \cdots & 0 & -\frac{f_{i1}(\omega_{i})}{g_{1}(\omega_{i})} & \cdots & -\frac{f_{im}(\omega_{i})}{g_{m}(\omega_{i})} \end{bmatrix}^{T} \\ \approx \begin{bmatrix} 0 & \cdots & 0 & 1 & 0 & \cdots & 0 & -\frac{\zeta_{i1}(\omega_{i1}^{2})}{2(\beta_{i1}+i\xi_{d,i1})\omega_{u,i1}^{2}} & \cdots & -\frac{\zeta_{im}(\omega_{im}^{2})}{2(\beta_{im}+i\xi_{d,im})\omega_{u,im}^{2}} \end{bmatrix}^{T}$$
(42)

where the first non-zero term is at the *i*th coordinate and the parameters $\omega_{a,\mu}$, β_{i1} , and $\xi_{d,\mu}$ are generalizations of the average frequency, detuning, and damping difference parameters of the MDOF/SDOF system

$$\omega_{\alpha,ij} = \frac{\omega_{\mu} + \omega_{ij}}{2} \quad \beta_{ij} = \frac{\omega_{\mu} - \omega_{ij}}{\omega_{\alpha,ij}} \quad \xi_{d,i1} = \xi_{\mu} - \xi_{i1}$$
(43)

The derivation of the modal properties associated with the equipment is similar to the above analysis. The frequency is given by the equipment subsystem frequency

$$\omega_{n+1} = \omega_{xy} (\sqrt{1 - \xi_{xy}^2} + i\xi_{xy})$$
(44)

which, when substituted into Eq.40, yields the following solution for the mode shape

$$\Phi_{n+j} = \left[-\frac{f_{1j}(\omega_{n+j})}{G_1(\omega_{n+j})} \cdots -\frac{f_{nj}(\omega_{n+j})}{G_n(\omega_{n+j})} 0 \cdots 0 1 0 \cdots 0 \right]^{T} \\ \approx \left[\frac{\epsilon_{1j} \zeta_{1j} \omega_{3j}^{2}}{2(\beta_{1j} + i\xi_{d,1j}) \omega_{\sigma,1j}^{2}} \cdots \frac{\epsilon_{nj} \zeta_{nj} \omega_{3j}^{2}}{2(\beta_{nj} + i\xi_{d,nj}) \omega_{\sigma,nj}^{2}} 0 \cdots 0 1 0 \cdots 0 \right]^{T}$$
(45)

These expressions reduce to the results in Section 6.4.1.1 for the detuned cases. For the singly tuned mode, it can be shown that the above expressions are equivalent to the expressions derived in Section 6.4.1.3 for the non-interaction case, where m_{ij} are small.

As in Section 4.5, the above expression appears to be indeterminate since it involves the small parameters ϵ_{ij} , β_{ij} , and $\xi_{ij,ij}$. However, the limit $m_{ij} \rightarrow 0$ is taken after the coefficients a_i and c_i are derived and the problem with the β_{ij} and $\xi_{ij,ij}$ terms are resolved when the modal responses are combined.

6.5.3.3 Spectral Moments

As stated earlier, closed form expressions will be obtained for the factors a_i and c_i , defined in Eq.2.26, which are the key factors of the modal decomposition method. Due to the simplicity of the expressions for the mode shapes and frequencies, the derivation is straightforward.

By following the matrix multiplication in Eqs.2.17a-c and 2.20 and taking the limit $m_{s_i} \rightarrow 0$, the following expression for the factors b_i are obtained which are independent of the mass ratio ϵ_{ij}

$$b_{i} = \frac{m_{i}}{2\omega_{pi}} \left[q_{i} - \sum_{j=1}^{m_{i}} \frac{q_{m+j}\zeta_{ij}\omega_{\lambda_{i}}^{2}}{2(\beta_{ij}+i\xi_{il,j})\omega_{a,j}^{2}} \right] \quad \text{for } i=1,\ldots,n$$
(46a)

$$b_{n+j} = \frac{iq_{n+j}}{2\omega_{sj}} \left[r_{n+j} + \sum_{i=1}^{n} \frac{r_i \zeta_{ij} \omega_{sj}^2}{2(\beta_{ij} + i\xi_{d,j}) \omega_{u,ij}^2} \right] \quad \text{for } j = 1, \dots, m$$
 (46b)

The factor a, and c, are found from Eq.2.26

$$a_{i} \approx 2\omega_{ji} \ln b_{i} = -\sum_{j=1}^{m} \frac{q_{n+j}\zeta_{ij}r_{i}\beta_{ij}\omega_{jj}^{2}}{2(\beta_{ij}^{2} + \zeta_{d,ij}^{2})\omega_{n,ij}^{2}} + r_{i}q_{i}$$
(47a)

$$a_{n+j} \approx 2\omega_{nj} \mathrm{Im} b_{n+j} - \sum_{j=1}^{n} \frac{q_{n+j} \zeta_{ij} r_{i} \beta_{ij} \omega_{ij}^{2}}{2(\beta_{ij}^{2} + \xi_{d,ij}^{2}) \omega_{n,ij}^{2}} + r_{n+j} q_{n+j}$$
(47b)

$$c_{i} = 2\operatorname{Re} b_{i} - \sum_{j=1}^{m} \frac{q_{n+j} \xi_{ij} r_{i} \xi_{d,ij} \omega_{ij}^{2}}{2\omega_{pi} (\beta_{ij}^{2} + \xi_{d,ij}^{2}) \omega_{a,ij}^{2}}$$
(47c)

$$c_{u+j} \approx 2 \operatorname{Re} b_{u+j} - \sum_{i=1}^{n} \frac{q_{u+j} \zeta_{ij} r_i \xi_{d,ij} \omega_{s_i}^2}{2 \omega_{s_i} (\beta_{ij}^2 + \xi_{d,ij}^2) \omega_{u,ij}^2}$$
(47d)

From these expressions, the spectral moments are easily found from Eqs.2.34 and 2.36, which are repeated here:

$$\mathbf{C}_{ij} = \mathbf{a}_i \mathbf{a}_j - \mathbf{D}_{ij} = \mathbf{a}_i \mathbf{c}_j - \mathbf{a}_j \mathbf{c}_i - \mathbf{E}_{ij} = \mathbf{c}_i \mathbf{c}_j$$
(48a)

$$\lambda_m = \sum_{i=1}^n \sum_{j=1}^n (C_{ij} \lambda_{m,ij} - D_{ij} \lambda_{m+1,ij} + E_{ij} \lambda_{m+2,ij})$$
(48b)

The frequencies and damping ratios needed to calculate the cross-spectral moments are given

by the original subsystem parameters as indicated in Eqs.41 and 44. The moments calculated by the above expressions are compared with exact results for various values of the secondary masses in Fig.6.8. The results are similar to those observed in Fig.6.7.

٠

| Table 6.1. Phys | sical Prop | erties a | f the |
|-----------------|------------|----------|-------|
| MDOF/MDC |)F Examp | le Syste | - 110 |

| Subsystem | Parameter Relationships | | | |
|-----------|--|--|--|--|
| Primary | $k_1 = (1.02)^2 m_1 \text{ (radians/sec)}^2$ $k_2 = (.6057)^2 m_1 \text{ (radians/sec)}^2$ $m_2 = m_1$ | | | |
| Secondary | k_3, k_4 (variable) $m_3 = em_1$ (e variable) | | | |

Table 6.2. Modal Properties of the Fixed Base Subsystems

| Subsystem | Modal DOF | Frequency (rad/s) | Damping Ratio |
|---|--------------|----------------------|------------------|
| Primary 1 | 1 | 1.02 | 0.02 |
| Primary 2 | 2 | 0.374 | 0.023 |
| | 3 | 0.98 | 0.06 |
| Secondary | 4 | . | 0.03 |
| , in the second s | 5 | 1.0 | 0.01 |

Table 6.3 Frequencies for Example System (ϵ_{11} =0.01)

| (rad/sec) Mode | Mada | Exact Frequency | | Computed Frequency | | Error |
|----------------|------|------------------|--------------|--------------------|------------|-------|
| | MOUE | Real Part | Imag. Part | Real Part | Imag. Part | % |
| 0.38 | 1 | 0.370 | 0.009 | 0.372 | 0.009 | 0.2 |
| | 2 | 0.382 | 0.011 | 0.383 | 0.010 | 0.1 |
| | 3 | 0.977 | 0.057 | 0.977 | 0.057 | 0.1 |
| | 4 | 0.991 | C.013 | 0.991 | 0.013 | 0.0 |
| | 5 | 1.028 | 0.018 | 1.028 | 0.018 | 0.0 |
| 1.00 | 1 | 0.374 | 0.008 | 0.374 | 0.008 | 0.0 |
| | 2 | 0.976 | 0.055 | 0.976 | 0.055 | 0.0 |
| | 3 | 0.989 | 0.017 | 0.988 | 0.017 | 0.1 |
| ł | 4 | 0.998 | 0.025 | 0.998 | 0.025 | 0.0 |
| L | 5 | 1.035 | 0.020 | 1.033 | 0.021 | 0.1 |

| Made DOF | | Exact Mode Shape | | Computed Mode Shape | | Error |
|-----------|-----------|------------------|-----------|---------------------|--------|-------|
| Milde DUF | Real Part | Imag. Part | Real Part | Imag. Part | % | |
| 1 | | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |
| | 2 | 0.000 | 0.000 | 0.000 | 0.000 | 0.0 |
| { | 3 | 0.000 | 0.000 | 0.000 | 0.000 | 0.0 |
| ļ | 4 | 0.387 | -0.004 | 0.387 | 0.000 | 0.4 |
| L | 5 | 0.387 | 0.002 | 0.387 | 0.000 | 0.4 |
| 2 | 1 | -0.004 | 0.000 | 0.000 | 0.000 | 0.4 |
| [| 2 | 0.130 | -0.157 | 0.123 | -0.163 | 2.1 |
| ţ | 3 | -0.009 | -0.004 | -0.008 | -0.005 | 0.2 |
| | 4 | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |
| | 5 | 0.646 | 0.093 | 0.665 | 0.075 | 2.8 |
| 3 | 1 | -0.002 | -0.001 | -0.003 | 0.001 | 0.3 |
| | 2 | 0.004 | -0.009 | 0.004 | -0.009 | 0.0 |
| 1 | 3 | 0.060 | -0.035 | 0.063 | -0.035 | 0.3 |
| | 4 | -0.215 | 0.620 | -0.229 | -0.600 | 3.4 |
| | 5 | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |
| 4 | 1 | -0.004 | 0.001 | -0.001 | -0.001 | 0.5 |
| | 2 | -0.014 | -0.013 | -0.013 | -0.014 | 0.2 |
| | 3 | -0.023 | -0.041 | -0.022 | -0.942 | 0.2 |
| | 4 | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |
| | 5 | 0.562 | -0.489 | 0.581 | -0.440 | 7.3 |
| 5 | 1 | 0.000 | 0.001 | -0.004 | -0.001 | 0.4 |
| { | 2 | -0.003 | 0.002 | -0.003 | 0.002 | 0.0 |
| | 3 | -0.211 | -0.065 | -0.204 | -0.066 | 0.7 |
| ł | 4 | -0.804 | -0.555 | -0.777 | -0.596 | 3.2 |
| t | 5 | 1.000 | 0.000 | 1.000 | 0.000 | 0.0 |

Table 6.4 Mode Shapes of Example System (ϵ_{11} =0.01, ω_{31} =1.0)



Figure 6.1. Example MDOF/MDOF System

.



Multiply Tuned System, wal-1.0 rad/sec



Multiply and Singly Tuned System, w_{s1}=0.38 rad/sec

Fig.6.2. Distribution of Subsystem Free Vibration Frequencies

__

- 201 -




Fig.6.3b. Complex Frequencies: Multiply Tuned System ω_{s1} =1.0 rad/sec, α_{11} =0.01, 0.005, 0.001

- 203 -

Ę























- 208 -







Fig6.7. Spectral Moment $\lambda_0^{(mon)}$ without Interaction Calculated from $H_{V}^{(mon)}(\omega)$



CHAPTER 7

SUMMARY AND CONCLUSIONS

7.1 Summary of Thesis

The general problem of finding the dynamic characteristics of arbitrary PS systems was investigated. Previous investigators have approximated or have neglected altogether one or more of the properties of such systems which include: interaction effects, correlation between subsystem modal responses and between support motions, non-classical damping, multiplesupport excitations, and single or multiple tuning. In this study, all of these effects are accounted for correctly by analyzing the combined equations of motion. This large system of equations is effectively reduced to a tractable and manageable form through a systematic application of the perturbation theory. Two approaches are provided: one based on modal analysis and the other based on frequency response methods. In the modal approach, simple formulations are provided for the complex-valued mode shapes and frequencies which in all but the multiply tuned case are in closed form. These modal properties are applicable to a variety of response analysis methods; references to some of these methods have been made. For the purpose of this study, a general modal combination rule for systems with complex-valued and closely spaced mode shapes and frequencies was developed for stationary stochastic input specified by its power spectral density function as well as by its response spectrum. These methods are particularly well-suited for the analysis of PS systems.

In the frequency response approach, simple closed form solutions for the frequency response function was obtained for all but the MDOF/MDOF systems, which can be used in cases where the input motion is stationary and is described by a power spectral density function.

For further insight into the behavior of PS systems, algebraic expressions for the response were obtained for the special and important case of white noise input. Several properties of the system were revealed, such as the relation between the mass ratios and interaction, the relation between the proximity of the subsystem frequencies and the corresponding tuning or resonance effects, and the relation between the mass ratios, the difference of subsystem damping values and the corresponding non-classical damping effects. Many other properties are explicitly included in the expressions for the response and were cited in the text.

The PS system was also analyzed without accounting for interaction effects. In the frequency response approach, the equations of motion were decoupled to two sets of equations corresponding to the coordinates of the two subsystems, as has been commonly done in the past. However, in the modal approach, the combined set of equations are retained and closed form expressions for the mode shapes and frequencies are obtained for all cases. These expressions were used in the modal decomposition method and simple closed form formulae were derived for a set of generalized participation factors which can be readily used to find the response of the system.

7.2 Conclusions and Recommendations

The main contribution of this work is the development of an accurate and computationally feasible modal analysis technique for PS system. The expressions that were obtained for the complex mode shapes and frequencies and the modal combination method are easily implemented into a digital computer. Such an implementation was used to derive the figures and tables in this study.

Different formulations can be used for various types of base input; a partial list follows:

- 1. For stationary stochastic input and for input specified by its response spectrum, the modal decomposition method developed in Chapter 2 is used.
- For non-stationary, non-white input, a number of existing methods including those in Refs.14, 30, 8 can be used.
- For deterministic input, the decoupled modal differential equations in Eq.2.25 can be integrated.

In many PS systems, the masses of the secondary subsystem are very small in comparison

to those of the primary subsystem. If the masses are small enough to satisfy the noninteraction criteria, the corresponding non-interaction formulations can be used, which are in closed form and easily calculated. These same formulations can also be used for those systems that do not satisfy the criteria, to provide a simple estimate of the response behavior. The results are conservative, which is often acceptable in preliminary design stages.

References

- 1. International Mathematics and Statistics Library, IMSL Inc., 1979.
- American Society of Mechanical Engineers, ASME Boiler and Pressure Vessel Code, ANSI/ASME BPV-III-1-A Section III, Rules for Construction of Nuclear Power Plant Components, Division 1, Appendix N, July 1981.
- A. D. Aleksandrov, Mathematics: Its Content, Methods, and Meaning, 1, M.I.T. Press, Cambridge, Massachusetts, 1963.
- M. Amin, W. J. Hall, N. M. Newmark, and R. P. Kasawara, "Earthquake Response of Multiple Connected Light Secondary Systems by Spectrum Methods," in Proceedings, ASME First National Congress on Pressure Vessel and Piping Technology. San Francisco, California, May 1971.
- K. J. Bathe and E. L. Wilson, "Stability and Accuracy Analysis of Direct Integration Methods," *Earthquake Engineering and Structural Dynamics*, vol. 1, pp. 283-291, 1973.
- J. M. Biggs and J. M. Roesset, "Seismic Analysis of Equipment Mounted on a Massive Structure," in Seismic Design of Nuclear Power Plants. ed. R. J. Hansen, pp. 319-343, M.I.T. Press, 1970.
- D. E. Cartwright and M. S. Longuet Higgins, "The Statistical Distribution of the Maxima of a Random Function," *Proceedings of the Royal Statistical Society*, vol. 237, pp. 212-232, 1956.
- T. K. Caughey, "Nonstationary Random Inputs and Responses," in Random Vibration, Volume 2, ed. S. H. Crandall, Chapter 3, pp. 56-84, M.I.T. Press, Cambridge, Mass, 1963.
- M. K. Chakravarti and E. H. Vanmarke, "Probabilistic Seismic Analysis of Light Equipment Within Buildings," in *Proceedings*, Fifth World Conference on Earthquake Engineering, vol. II, Rome, Italy, 1973.

- R. W. Clough and J. Penzien, *Dynamics of Structures*, McGraw-Hill Book Co., Inc., New York, N.Y., 1975.
- S. H. Crandall and W. D. Mark, Random Vibration of Mechanical Systems, Academic Press, New York, N.Y., 1963.
- A. F. Curtis and T. R. Boykin, "Response of Two-Degree-of-Freedom Systems to White-Noise Base Excitation," *Journal of the Acoustical Society of America*, vol. 33, pp. 655-663, 1961.
- A. G. Davenport, "Note on the Distribution of the Largest Value of a Random Function with Application to Gust Loading," in *Proceedings*, Institution of Civil Engineers, vol. 28, pp. 187-196, 1964.
- A. Debchandhury and D. A. Gasparini, "State Space Random Vibration Theory," Case Institute of Technology, Dept. of Civil Engineering Research Report ENG 77-19364, October 1980.
- A. Der Kiureghian, "Structural Response to Stationary Excitation," Journal of the Engineering Mechanics Division, vol. 106, no. EM6, pp. 1195-1213, Dec. 1980.
- A. Der Kiureghian, "A Response Spectrum Method for Random Vibration Analysis of MDF Systems," *Earthquake Engineering and Structural Dynamics*, vol. 9, pp. 419-435, 1981.
- A. Der Kiureghian, A. Asfura, J. L. Sackman, and J. M. Kelly, "Seismic Response of Multiply Supported Piping Systems," in *Proceedings*, Seventh International Conference on Structural Mechanics in Reactor Technology, Chicago, Illinois, 1983.
- A. Der Kiureghian, J. L. Sackman, and B. Nour-Omid, "Dynamic Analysis of Light Equipment in Structures: Modal Properties of the Combined System," *Journal of the* Engineering Mechanics Division, vol. 109, no. EM1, Feb., 1983.
- 19. A. Der Kiureghian, J. L. Sackman, and B. Nour-Omid, "Dynamic Analysis of Light Equipment in Structures: Response to Stochastic Input," *Journal of the Engineering*

Mechanics Division, vol. 109, no. EM1, Feb., 1983.

- 20. A. Erdelyi, Asymptotic Expansions, Dover, New York, New York, 1956.
- W. C. Hurty and M. F. Rubinstein, *Dynamics of Structures*. Prentice Hall, Inc., Englewood Cliffs, N.J., 1964.
- T. Igusa and A. Der Kiureghian, "Response Spectrum Method for Systems with Non-Classical Damping," in *Proceedings*, ASCE-EMD Specialty Conference, West Lafayette, Indiana, May 23-25, 1983
- T. Igusa, A. Der Kiureghian, and J. L. Sackman, "An Investigation of the Spectral Density Function and Spectral Moments for Equipment-Structure Systems," unpublished paper.
- K. Kanai, "Semi-Empirical Formula for Seismic Characterization of the Ground," Bulletin of Earthquake Research Institute, vol. 35, University of Tokyo, Tokyo, Japan, June, 1967.
- K. K. Kapur and L. C. Shao, "Generation of Seismic Floor Response Spectra for Equipment Design," in *Proceedings*, Specialty Conference on Structural Design of Nuclear Plant Facilities, ASCE, Chicago, Illinois, December, 1973.
- M.-C. Lee and J. Penzien, "Stochastic Seismic Analysis of Nuclear Power Plants and Piping Systems subjected to Multiple Support Excitations," Report Number UCB/EERC-80/19, Earthquake Engineering Research Center, University of California, Berkeley, Calif., June 1980.
- K. R. Leimbach and J. P. Sterkel, "Comparison of Multiple Support Excitation Solution Techniques for Piping Systems," in *Proceedings*, Fifth International Conference on SMiRT, Berlin, Germany, August 13-17, 1979.
- Y. K. Lin, Probabilistic Theory of Structural Dynamics, McGraw-Hill Book Co., Inc., New York, N.Y., 1967.

- R. H. Lyon, "On the Vibration Statistics of a Randomly Excited Hard-Spring Oscillator," *Journal of the Acoustical Society of America*, vol. 32, pp. 716-719, 1960. Part II: Vol. 33, pp. 1395-1403, 1961.
- S. F. Masri, "Response of a Multi-Degree-of Freedom System to Nonstationary Random Excitation," *Journal of Applied Mechanics*, vol. 45, no. 3, pp. 649-656, September 1978.
- T. Nakhata, N. M. Newmark, and W. J. Hall, "Approximate Dynamic Response of Light Secondary Systems," Structural Research Series Report No. 396, Civil Engineering Studies, University of Illinois, Urbana, Ill., 1973.
- 32. A. H. Nayfeh, Perturbation Methods, Wiley, New York, New York, 1973.
- N. M. Newmark, "Earthquake Response Analysis of Reactor Structures," Nuclear Engineering and Design, vol. 20, no. 2, pp. 303-322 . 1972.
- J. Penzien and A. K. Chopra, "Earthquake Response of Appendage on a Multi-Story Building," in *Proceedings*, Third World Conference on Earthquake Engineering, New Zealand, 1965.
- K. A. Peters, D. Schmitz, and U. Wagner, "Determination of Floor Response Spectra on the Basis of the Response Spectrum Method," Nuclear Engineering and Design, vol. 44, pp. 255-262 . 1977.
- G. C. Ruzicka and A. R. Robinson, "Dynamic Response of Tuned Secondary Systems," Report Number UILU-ENG-80-2020, Department of Civil Engineering, University of Illinois, Urbana, Ill., Nov. 1980.
- J. L. Sackman and J. M. Kelly, "Seismic Analysis of Internal Equipment and Components in Structures," *Engineering Structures*, vol. 1, no. 4, pp. 179-190, July, 1979.
- R. H. Scanlan and K. Sachs, "Earthquake Time Histories and Response Spectra," Journal of the Engineering Mechanics Division, vol. 100, no. EM4, pp. 635-655, August, 1974.

- D. E. Shaw, "Seismic Structura: Response Analysis for Multiple Support Excitation," in *Proceedings*, Third International Conference on Structural Mechanics in Reactor Technology, Paper K5/2, London, England, September 1-5, 1975.
- 40. A. K. Singh, "A Stochastic Model for Predicting Maximum Response of Light Secondary Systems," thesis presented to the University of Illinois at Urbana, Illinois, in 1972, in partial fulfillment of the requirements for the degree of Doctor of Philosophy.
- M. P. Singh, "Generation of Seismic Floor Spectra," Journal of the Engineering Mechanics Division, vol. 101, no. EM5, pp. 594-607, Oct. 1975.
- 42. M. P. Singh, "Seismic Design Input for Secondary Structures," *Journal of the Structural Division*, vol. 106, no. ST2, pp. 505-517, Feb. 1980.
- M. P. Singh, "Seismic Response by SRSS for Nonproportional Damping," Journal of the Engineering Mechanics Division, vol. 106, no. EM6, pp. 1405-1419, Dec. 1980.
- E. H. Vanmarcke, "Properties of Spectral Moments with Applications to Random Vibration," *Journal of the Engineering Mechanics Division*, vol. 98, no. EM2, pp. 425-446., Apr. 1972
- E. H. Vanmarcke, "On the Distribution of the First-Passage Time for Normal Stationary Random Processes," *Journal of Applied Mechanics*, vol. 42, pp. 215-220, Mar. 1975.
- 46. E. H. Vanmarke, "A Simplified Procedure for Predicting Amplified Response Spectra and Equipment Response," in *Proceedings*, Sixth World Conference on Earthquake Engineering, vol. III, New Delhi, India, 1977.
- K. M. Vashi, "Seismic Spectral Analysis of Structural Systems Subjected to Nonuniform Excitation at Supports," in *Proceedings*, Second ASCE Specialty Conference, vol. 1-A, pp. 188-211, New Orleans, Louisiana, December, 1975.

- R. Villaverde and N. M. Newmark, "Seismic Response of Light Attachments to Buildings," Structural Research Series No. 469, University of Illinois, Urbana. Illinois, February, 1980.
- 49. G. B. Warburton and S. R. Soni, "Errors in Response Calculations of Non-Classically Damped Structures," *Journal of Earthquake Engineering and Structural* Dynamics, vol. 5, pp. 365-376, 1977.
- E. L. Wilson, A. Der Kiureghian, and E. P. Bayo, "A Replacement for the SRSS Method in Seismic Analysis." *Earthquake Engineering and Structural Dynamics*, vol. 9, pp. 187-194, 1981.

EARTHOUAKE ENGINEERING RESEARCH CENTER REPORTS

NCTE: Numbers in parentheses are Accession Numbers assigned by the National Technical Information Service; these are followed by a price code. Copies of the reports may be ordered from the National Technical Information Service, 5285 Port Royal Road, Springfield, Virginia, 22161. Accession Numbers should be quoted on orders for reports (PB --- ---) and remittance must accompany each order. Reports without this information were not available at time of printing. The complete list of EERC reports (from EERC 67-1) is available upon request from the Earthquake Engineering Research Center, University of California, Berkeley, 47th Street and Hoffman Boulevard, Richmond, California 94804.

- UCB. EERC-77/01 "PLUSH A Computer Program for Probabilistic Finite Element Analysis of Seismic Boll-Structure Interaction," by M.P. Romo Organista, J. Lysmer and H.J. Seed - 1977 (PB81 177 651)A05
- UCB/EERC-77/02 "Soil-Structure Interaction Effects at the Humboldt Bay Power Plant in the Ferndale Earthquake of June 7, 1975," by J.E. Valera, H.B. Seed, C.F. Tsai and J. Lysmer 1977 (PB 265 795)A04
- UCB/EERC-77/03 "Influence of Sample Disturbance on Sand Response to Cyclic Loading," by K. Mori, H.B. Seed and C.K. Chan = 1977 (PB 267 352)A04
- UCB/EERC-77/04 "Seismological Studies of Strong Motion Records," by J. Shoja-Taheri 1977 (PB 269 655)Al0
- UCB/EERC-77/05 Unassigned
- UCB/EERC-77/06 "Developing Methodologies for Evaluating the Earthquake Safety of Existing Buildings," by No. 1 -8. Bresler; No. 2 - B. Bresler, T. Okada and D. Zisling; No. 3 - T. Okada and B. Bresler; No. 4 - V.V. Bertero and B. Bresler - 1977 (PB 267 354)A08
- UCB/EERC-77/07 "A Literature Survey Transverse Strength of Masonry Walls," by Y. Cmote, R.L. Mayes, S.W. Chen and R.W. Clough 1977 (pB 277 933)A07
- UCB/EERC=77/08 "DRAIN=TABS: A Computer Program for Inelastic Eauthquake Response of Three Dimensional Buildings," by R. Guendelman=Tgrael and G.H. Powell = 1977 (PB 270 693)A07
- UCB/EERC-77/09 "SUBWALL: A Special Purpose Finite Element Computer Program for Practical Elastic Analysis and Design of Structural Walls with Substructure Sption," by D.D. Le, H. Peterson and E.P. Popov - 1977 (PB 270 567)A05
- UCB/EERC-7".10 "Experimental Evaluation of Seismic Design Methods for Broad Cylindrical Tanks," by D.P. Clough (PB 272 280)AL3
- UCB/EERC-77/11 "Earthquake Engineering Research at Berkeley 1976, 1977 (PB 273 507)AC9
- UCB/EERC-77/12 "Automated Design of Earthquake Resistant Multistory Steel Building Trames," by N.D. Walker, Jr. 1977 (PB 276 526)A09
- UCB/EERC-77/13 "Concrete Confined by Rectangular Hoops Subjected to Axial Loads," by J. Vallenas, V.V. Bertero and E.P. Popov 1977 (PB 275 165)A06
- UCB/EERC-77/14 "Seismic Strain Induced in the Ground During Earthquakes," by Y. Sugimura = 1977 (PB 234 201)/04
- UCB/EERC-77/15 Unassigned
- UCB/ZERC-77/16 "Computer Aided Optimum Design of Ductile Reinforced Concrete Moment Resisting Frames," by 3.W. Zagajeski and V.W. Bertero - 1977 (PB 280 137)A07
- UCB/EERC-77/17 "Earthquake Simulation Testing of a Stepping Frame with Energy-Absorbing Devices," by J.M. Kelly and D.F. Tsztoo 1977 (PB 273 506)404
- UCB/EERC-77/18 "Inelastic Gehavior of Eccentrically Braced Steel Frames under Cyclic Loadings." by C.W. Roeder and E.P. Popov - 1977 (PB 275 S26)A15
- UCB/EERC-77/19 "A Simplified Procedure for Estimating Earthquake-Induced Deformations in Dams and Embankments," by F.I. Makdisi and H.B. Seed - 1977 (PB 276 920)A04
- UCB/EERC-77/20 "The Performance of Earth Dams during Earthquakes," by H.B. Seed, F.I. Makdisi and P. de Alba 1977 (PB 276 821)A04
- UCE/EERC-77/21 "Dynamic Plastic Analysis Using Stress Resultant Finite Element Formulation," by P. Lukkunapvasit and J.M. Kelly = 1977 (PP 275 453)A04
- UCB/EERC-77/2' "Preliminary Experimental Study of Seismic Uplift of a Steel Frame," by R.W. Clough and A.A. Huckelbridge 1977 (PB 278 769)AC8
- UCB/EERC-77/23 "Earthquake Simulator Tests of a Nine-Story Steel Frame with Columns Allowed to Uplift," by X.A. Huckelbridge = 1977 (PB 277 944)A09
- UCB/EERC-77/24 "Nonlinear Soil-Structure Interaction of Skew Highway Bridges," by M.-C. Chen and J. Penzien 1977 (PB 276 176)407
- UCS/EERC-77/25 "Seismic Analysis of an Offshore Structure Supported on Pile Foundations," by D.D.-N. Liou and J. Penzlen 1977 (PS 283 180)A06
- UCB/EERC-77/26 "Dynamic Stiffness Matrices for Homogeneous Viscoelaptic Half-Planes," by G. Dasgupta and A.K. Chopra -1977 (FB 279 654)A06

Reproduced from best available copy

| 008 EERC+TT 3) | "An Approach for Improving Seismic + Resistant Benavior of Reinforced Concrete Interior Joints," by 3. Jalunic, V.V. Bertero and E.P. Popov - 1977 (PB 290 370)A06 |
|-----------------|--|
| "CB EERC-73.01 | "The Development of Energy-Absorbing Devices for Aseismic Base Isolation Systems," by J.M. Kelly and D.F. Tsztod - 1978 (PB-284-978)A04 |
| CCB TERC+78 12 | "Effect of Tensile Prestrain on the Tyclic Response of Structural Steel Connections, by J.S. Bouwkamp and A. Mukhopadhyay - 1979 |
| 072 2E8C-74 03 | "Experimental Results of an Barthquake Isolation System using Natural Rubber Brarings," by J.M. Eldinger and J.M. Kelly - 1978 (PB 291 666)404 |
| TOB EFRC-14 14 | "Seismid Benavior of Tall Liquid Storage Tanks," by A. Miwa - 1974 (PB 294 (17)A14 |
| ULB EERC-74 05 | "Hystoretic Benavior of Peinforced Concrete Columns Subjected to High Axial and Cyclic Snear Porcess" by S.W. Cadajeski, V.V. Bertero and J.S. Bouwkamp - 1976 (PB 203 A50(A13 |
| NUR EERNHÜH DE | "There Dimensional Inclusion Frame Elements for the ANSE-I Program," by A. Flani, 2010 Fow and A.B. Fowell - 1978 (PB-295-755)A04 |
| 19 EESC+14 1 | "Studies of Structural Response to Parthquak- spound Unition," by O.A. Lopez and A.F. Doorra - 1474 PR 042 Tabla35 |
| 05 EEFC-16 18 | "A Laboratory Study of the Ply-d-Structure Interaction of Submersed Tanks and Callins in Bartogiakes," by R.S. Byrd + 1979 (PB 194 4) (1940) |
| WB ESBC-74 () | Unassigned |
| NCB SERCETE 1 | "Seismic Performance of Nonstructural and Becondary Structural Flement»," by I. Sakamoto - 1+73 (PB81-154-593)A05 |
| UCB EERC- "S 11 | "Mathemotical Modelling of Hysteresis Loops for Reinforced Concrete Alumns," by J. Nakata, T. Surcul and J. Penzien - 1973 (PB 298-274)A05 |
| UCB. EERC-79 12 | "Damageability in Existing Buildings," by T. Blejwas and B. Bresler - 1978 (PB 80 156 378)A05 |
| UCB EERC-79.13 | "Dynamic Behavior of a Pedestal Base Multistory Building," by R.M. Stephen, E.L. Wilson, J.J. Bouwkamp and M. Button ~ 1978 (PD 286 650)AU" |
| VCB EERC-78 14 | "Saismic Response of Bridges - Jase Studies," by R.A. Imbsen, V. Nutt and J. Penzien - 1978 (PB 206 503)A.) |
| 008 EERC-19-15 | "A Substructure Technique for Nonlinear Static and Dynamic Analysis." by D.C. Row and D.H. Powell - 1978 (FR 288 077)Ald |
| UCB: EERC-T8:16 | "Seismic Rick Studies for San Francisco and for the Greater San Francisco Bay Area," by 0.5. Cliveira - 1978 (PB 81 120 115)A07 |
| UCB/EERC-78/17 | "Strength of Timber Roof Connections Subjected to Cyclic Loads," by P. Gülkan, R.L. Mayes and R.W. Clough - 1978 (HUL-000 1491)A07 |
| UCB/EERC-78/18 | "Response of K-Braued Steel Frame Hudels to Lateral Loads," by J.G. Bouwkump, R.M. Stephen and E.P. Popov - 1978 |
| UCB/EERC-79-19 | "Rational Design Methods for Light Equipment in Structures Subjected to Ground Motion," by J.L. Sackman and J.H. Kally - 1978 (PS 292 357)A04 |
| UCB/EERC-76/20 | "Testing of a Wind Restraint for Aseismic Base Isolation," by J.N. Kelly and D.E. Chitty - 1978 (PS 292 833)R03 |
| UCB/EERC-79/21 | "APOLLO - A Computer Program for the Analysic of Pore Pressure Generation and Dissipation in Horizontal Sand Layers During Cyclic or Earthquake Loading," by P.P. Martin and H.B. Seed - 1978 (PB 292 835)AC4 |
| UCB./JERC-78/22 | "Optimal Design of an Earthquake Isolation System," by N.A. Bhatti, K.S. Pister and E. Polak - 1978 (PE 294 735)A06 |
| UCB/EERC-78/23 | "NASH - A Computer Program for the Non-Linear Analysis of Vertically Propagating Shear Waves in Horizontally Layered Deposits," by P.P. Martin and H.B. Seed - 1978 (PB 293 101)AOS |
| UCB/EERC-76/24 | "Investigation of the Elastic Characteristics of a Three Story Steel Frame Using System Identification," by I. Kaya and H.D. McNiven - 1978 (PS 296 225)A06 |
| UCB/22RC-76/25 | "Investigation of the Nonlinear Characteristics of a Three-Story Steel Frame Using System Identification," by I. Kaya and H.D. McMiven + 1978 (PB 301 363)AD5 |
| | |

-222-

UCB EERC-77/27 "A Practical Soft Story Jarthquake Isolation System," by J.M. Kelly, J.M. Eidinger and C.J. Derham -1977 (PB 276 814)A07

UCB/EERC-77729 "BeiSmic Safety of Existing Buildings and Incentives for Hazard Mitigation in San Francisco: An Exploratory Study," by A.J. Meltsner - 1977 (PB 201 970)A05

UCB EERC-77 29 "DVnamic Analysis of Electrohydraulic Shaking Tables," by D. Rea, S. Abedi-Hayati and Y. Takanashi 1977 (PB 202 5691A04

Reproduced from best available copy

•

UCB/EERC-78/26 "Studies of Strong Ground Motion in Taiwan," by Y.M. Hsiung, B.A. Bolt and J. Penzien - 1978 98 298 4361AC6 VCB MERC+78/27 "Cyclic Loading Tests of Masonry Single Piers: Volume 1 - Height to Width Ratio of 2," by P.A. Hidalgo, R.L. Mayes, H.D. McNiven and R.V. Clough - 1978 (PB 296 211)A07 PCB:SERC+79.29 (Sydlid Loading Tests of Masonry Single Fiers: Molume 2 - Height to Width Ratio of 1," by S.-W.J. Chen. F.A. Hidalon, R.L. Mayes, R.W. Clough and H.D. Motliven - 1978 (PB 295 212)AD9 UCB EERC-79 29 "Analytical Procedures in Soil Dynamics," by J. Lysmer - 1978 'PB 298 445)AD6 UCB/EERC-79/31 "Hysteretic Behavior of Lightweight Reinforced Concrete Beam-Column Subassemblages," by B. Porzani, E.P. Popov and V.V. Bertero - April 1979(PB 298 267) A06 UCB/ESRC-79/02 "The Unvelopment of a Mathematical Model to Predict the Flexural Response of Reinforced Concrete Beams to Cyclic Loads, Using System Identification," by J. Stanton & H. McNiven - Jan. 1979(PB 295 875)A10 (CS:EERC-?9/03 "Linear and Honlinear Earthquake Response of Simple Torsionally Coupled Systems," by C.L. Kan and A.K. Chopra - Feb. 1979(PB 298 262) A06 UCB SERC-79/04 "A Mathematical Model of Masonry for Predicting its Linear Seismic Respinse Characteristics," by Y. Mengi and H.D. McMiwin - Feb. 1979(PB 298 266)A06 UCB/EFRC-13/05 "Mechanical Behavior of Lightweight Concrete Confined by Different Types of Lateral Reinforcement," by M.A. Manrique, V.V. Bertero and E.P. Popov - May 1979(PB 301 114) A0t UDB EERC-79/06 "Static Tilt Tests of a Tall Cylindrical Liquid Storage Tank," by R.W. Clough and A. Niwa - Feb. 1979 PB 301 167) A06 UCB/EEPC-19:07 "The Design of Steel Energy Absorbing Restrainers and Their Incorporation into Nuclear Power Plants for Enhanced Safety: Volume 1 - Summary Report," by P.N. Spencer, V.F. Zackay, and E.R. Parker -Feb. 1979:028 EERC-79/07)A09 100 EEPC-19/08 "The Devign of Steel Energy Absorbing Restrainers and Their Incorporation into Nuclear Power Plants for Ennanced Safety: Volume 2 - The Development of Analyses for Reactor System Piping, ""Simple Systems" by "N.C. Lee, J. Penzien, A.K. Chopri and K. Suzuki "Complex Systems" by S.H. Powell, E.C. Milson, P.W. Clough and D.G. Row - Feb. 1979(UCB/EERC-79703) A10 UCB/EERC- 19 19 "The Design of Steel Energy Absorbing Restrainers and Their Incorporation into Nuclear Power Plants for Enhanced Safety: Volume 3 - Evaluation of Commercial Steels " by W.S. (Wen, R.M.N. Pellovx, R.D. Ritchiel M. Faral, T. Shhashi, S. Toplosky, S.S. Hartman, - F. Sackay and E.R. Parker -Feb. 1979/UCB/EERC-79/09) A04 UTP (ERG-79 1) "The Design of Steel Energy Absorbing Restrainers and Their Incorporation into Nuclear Power Plants for Enhanced Safety: Volume 4 - A Review of Energy-Absorbing Devices," by J.M. Kelly and M.S. Skinner - Feb. 1979(UC8/EERC-70/10)A04 108/FEBC-19-11 "Conservatism In Summation Rules for Closely Spaced Modes," by J.M. Kelly and J.L. Sackman - May 1979 (PB 301 329) A03 UCB-SERG-79-11 "Cyclic Loading Tests of Masonry Single Piers; Volume 3 - Height to Width Ratio of 0.5," by P.A. Hidalgo, R.L. Mayes, H.D. McNiven and R.W. Clough - May 1979(PB 301 321)A08 UCB/EERC-79/1) "Cyclic behavior of Dense Course-Grained Materials in Relation to the Seismic Stability of Dams," by 1.5. Banerjee, H.B. Seed and C.K. Chan - June 1979(PB 301 373)A13 UCB/EERC-79/14 "Seismic Benavior of Peinforced Concrete Interior Beam-Column Subassemblages," by S. Viwathanateps, E.P. Popov and V.V. Bertero - June 1979.PB 301 326)A10 CB/SEPC-79/15 "Optimal Design of Localized Nonlinear Systems with Dual Performance Criteria Under Earthquake Excitations," by M.A. Bhatti - July 1979(PB 80 167 109) A06 UCB/EERC-79/16 "OPTDYN - A General Purpose Optimization Program for Problems with or without Dynamic Constraints," by M.A. Bhatti, E. Polak and K.S. Pister - July 1979(PB 40 167 J91)405 UCB/EERC-79/11 "ANSR-II, Analysis of Nonlinear Structural Response, Users Manual," by D.P. Mondkar and G.H. Powell July 1979 (PB 30 113 301) AOS SCD/EERC-79/18 "Soil Structure Interaction in Different Seismic Environments," A. Gomez-Masso, J. Lysmer, J.-C. Chen and H.B. Seed - August 1979(PB 80 101 520) A04 UCB/EERC-79/19 "ARMA Models for Earthquake Ground Motions," by N.K. Chang, J.W. Kwiatkowski, R.F. Nau, R.M. Oliver and K.S. Pister - July 1979(PB 301 166)A05 UCB/EERC-79/20 "Hysterstic Behavior of Reinforced Concrete Structural Walls," by J.M. Vallenas, V.V. Bertero and E.P. Popov - August 1979(PB 80 165 905) A12 CCB/EERC-79/21 "Studies on High-Frequency Vibrations of Buildings - 1: The Column Effect," by J. Lubliner - August1979 (PB 80 158 553) A03 UCB/EERC-79/22 "Effects of Generalized Loadings on Bond Reinforcing Bars Imbedded in Confined Concrete Blocks," by S. Viwathanateps, S.P. Popov and V.V. Serters - August 1979(PB 81 124 018)A14 UCB/EERC-79/2: "Shaking Table Study of Single-Story Masonry Houses, Volume 1: Test Structures 1 and 2," by P. Gülkan, R.L. Mayes and R.W. Clough - Sept. 1979 (HUD-000 1763)A12 UCB/EERC-79/24 "Shaking Table Study of Single-Story Masonry Houses, Volume 2: Test Structures 1 and 4," by P. Gülkan, R.L. Mayes and R.W. Clough - Sapt. 1979 (HUD-000 1836)Al2 UCB/EERC-79/25 "Shaking Table Study of Single-Story Masonry Houses, Volume 3: Summary, Conclusions and Recommendations," by R.W. Clough, R.L. Mayes and P. Gülkan - Sept. 1979 (MUD- 300 1637)306

| UCB/EERC-79/24 | "Recommendations for a U.SJapan Cooperative Research Program Utilizing Large-Scale Testing Facilities," by U.SJapan Planning Group - Sept. 1979(PB 301 407)A06 |
|--------------------|---|
| UCB. EERC+ 79 /2 7 | "Earthquake-Induced Liquefaction Near Lake Amatitlen, Guatemala," by H.B. Seed, I. Arango, C.K. Chan, A. Gomez-Masso and R. Grant de Ascoli - Sept. 1979(NUREG-CR1341)A03 |
| UCB/EFRC-19/26 | "Infill Panels: Their Influence on Seismic Response of Buildings," by J.W. Axley and V.V. Bertero Sept. 1979(PB-80-163-371)Alo |
| UCB/EERC-79/29 | "3D Truss Bar Element (Type 1) for the ANSR-II Program," by D.P. Mondkar and J.H. Powell - Nov. 1979 (PS 90 169 709).02 |
| UCB EERC-19/30 | "20 Seam-Column Element (Type 5 - Parallel Element Theory) for the ANSR-II Program." by D.G. Row, J.H. Powell and D.P. Mondkaz - Dec. 1979(PB 60 167 224)A03 |
| UCE/EERC+79/31 | "3D Beam-Column Element (Type 2 - Parallel Element Theory) for the ANSR-II Program," by A. Riani, J.H. Powell and D.P. Mondkar - Dec. 1979(PB of 167 216)AO3 |
| UCB/EERC-79/32 | "On Response of Structures to Stationary Excitation," by A. Der Kluuwshiin - Dec. 1979(PB-80166-929)A03 |
| UCB/EERC+79/33 | "Undisturbed Sampling and Cyclic Load Testing of Sands," by 5. Singh, H.B. Seed and C.K. Chan Dec. 1979(ADA 037-298)A07 |
| UCB EERC- 79-34 | "Interaction Effects of Simultaneous Torsional and Compressional Cyclic Leading of Band," by P.M. Griffin and W.N. Houston - Dec. 1979(AEA 192-352)AE5 |
| COBVEERCHBH D1 | "Eiringuake Response of Concrete Gravity Dams Including Hydrodynamic and Foundation Interaction Effects," by A.A. Chopra, P. Chakribarti and S. Quita - Jan. 1980(AD-A087237)Alu |
| UCB/EEFC-80/02 | "Rocking Pesponse of Pigld Blocks to Earthquakes," by C.S. Yim, A.K. Chopra and J. Penzien - Jan. 1960 (PBRC 166 - 021A04 |
| "CB:5ERC+H0:33 | "Optimum Inelastic Design of Seismic-Resistant Reinforced Concrete Frame Structures," ny S.W. Tagajeski and V.V. Bertero - Jan. 1980 1880 184 6351806 |
| CCB SERCHAIND4 | "Effects of Amount and Arrangement of Wall-Panel Reinforcement on Hysteretic Benavior of Reinforces Concrete Walls," by P. 1117a and V.V. Bertero - Feb. 1980(PB8, 122-525/AD9 |
| UCE-EERC+89-05 | "Shaking Table Research on Concrete Dam Models," by A. Niwa and R.W. Clough - Sept. 1980(PB8)122-366/A06 |
| UCE LERGHAD, 16 | "The Design of steel Energy-Absorbing Pestrainers and their Incorporation into Nuclear Power Plants F . Enhanced Safety (Vol LA): Piping with Energy Absorbing Pestrainers: Farameter Study on Small Systems. of 3.H. Powell, D. Duchourlian and J. Simons - June 1980 |
| 105-EERC-30-07 | "Inelastic Torsional Response of Structures Subjected to Earthquake Ground Motions," by Y. Yamazaki April 1980(PB01 1.2 027)A00 |
| UCB/EERC+80/08 | "Study of X-Braced Steel Frame Structures Under Earthquake Simulation," by Y. Ghanaat - April 1980 (PB81-122-335)all |
| 108/EERC-80/19 | "Hybrid Modelling of Soil-Structure Interaction," by S. Supta, T.W. Lin. J. Penzien and C.S. Yen May 1983(P881-122-319)A07 |
| UCB/EEPC+30/10 | "General Appiroability of a Nonlinear Model of a One Story Steel Frame," by B.I. Sveinsson and H.D. McNiven ~ May 1980(PB81 124 377)A06 |
| UCB/EERC-80/11 | 1- Green-Function Method for Wave Interaction with a Submerged Body," by W. Kioka - April 1980 (PB01 122 269)Au7 |
| UCB/SERC-3, 12 | "Hydrodynamic Pressure and Added Mass for Axisymmetric Bodies," by F. Nilrat - May 1980(PB81 122 343)AJB |
| UCB/ESRC-80/13 | "Treitment of Non-Linear Drag Forcus Acting on Offshore Platforms," by B.V. Dao and J. Penzien May 1980(2081-153-413)AC7 |
| UCB/SERC-80/14 | '2D Plane/Axisymmetric Solid Element (Type 3 - Elastic or Elastic-Perfectly Plastic) for the ANSR-II Program," by D.P. Mondkar and S.H. Powell - July 1980(PB61 122 350)AO3 |
| UCB/EERC-00/15 | "A Response Spectrum Method for Random Vibrations," by A. Der Klureghlan - June 1980(PB81122 301)A03 |
| UCB/EERC-80/15 | "Cyclic Inelastic Buckling of Tubular Steel Braces," by V.A. Zayas, E.P. Spov and S.A. Mahin June 1980(PB8) 124 985)Al0 |
| UCB/EEAC-80/17 | "Dynamic Respinse of Simple Arch Dams Including Hydrodynamic Interaction," by C.S. Porter a 1 A.K. Chopra - July 1980(PP61 124 000)A13 |
| UCB/EEPC-60/18 | "Experimental Testing of a Friction Damped Aseismic Base Isolation System .th Fail-Safe Characteristics," by J.M. Kelly, K.E. Beucke and N.S. Skinner - July 1980(1981-148-595)A04 |
| UCB/EERC-80/19 | "The Design of Steel Energy-Absorbing Restrainers and their Incorporation into Nuclear Power Plants for Enhanced Safety [Vol 18]: Stochastic Seismic Anelyses of Nuclear Power Plant Structures and Piping Systems Subjected to Multiple Support Excitations," by M.C. Lee and J. Penzian - June 1980 |
| UCB/EERC-30/20 | "The Design of Steel Energy-Absolbing Restrainers and their Incorporation into Nuclear Power Plants for Enhanced Safecy (Vol 10): Numerical Method for Dynamic Substructure Analysis," by J.M. Dickens and E.L. Wilson - June 1980 |

UCB/EERC-80/21 "The Design of Steel Energy-Absorbing Restrainers and their Incorporation into Nuclear Power Plants for Enhanced Safety (Vol 2): Development and Testing of Restraints for Nuclear Piping Systems," by J.M. Kelly and M.S. Skinner - June 1980

UCB/EERC-80/22 "3D Solid Element (Type 4-Slastic or Elastic-Perfectly-Plastic) for the ANSR-II Program," by D.P. Mondkar and G.H. Powell - July 1980(P881 123 242)A03

UCB/EERC-80/23 "Gap-Friction Element (Type 5) for the ANSR-II Program," by D.P. Nondkar and G.H. Powell - July 1980 (P881 122 285)A03

2

| UCB/EERC-00/24 | "U-Bar Restraint Element (Type 11) for the ANSR-II Program," by C. Oughourlian and G.H. Powell July 1980(PB81 122 293)AO3 |
|------------------|---|
| UCB/EERC-86/25 | "Testing of a Natural Rubber Base Isolation System by an Explosively Simulated Earchquake," by J.M. Kelly - August 1980(PB81 201 360)A04 |
| UCB/EERC-80/26 | "Input Identification from Structural Vibrational Response," by Y. Hu - August 1980(2881-152-308)AG5 |
| JOB. EERC-80/27 | "Cyclic Inelastic Behavior of Steel Offshore Structures," by V.A. Zayas, S.A. Mahin and E.P. Popov August 1980 9881 196 1801A15 |
| UCB/EERC+30/28 | "Shaking Table Testing of a Reinforced Concrete Frame with Biakial Response," by M.G. Oliva October 1980(P881-154-304)Alb |
| 108/EERC+90 29 | "Cynamic Properties of a Ewelve-Story Prefabricated Panel Building," by J.G. Bouwkamp, J.P. Kollegger and R.M. Stephen - October 1980(P282 117 128)A06 |
| UCB EERC-90/30 | "Dy amic Properties of an Eight-Story Prefabricated Panel Building," by J.C. Bouwkamp, J.P. Kollegger and R.M. Stephen - October 1980(PB81 200 313)A05 |
| 008 (SEBC-90 (31 | "Fredictive Dynamic Response of Panel Type Structures Under Earthquakes," by J.P. Kollegger and J.J. Bouwkamp - October 1980(FB81 152 316/A04 |
| 108 SEFC-60/32 | "The Design of Steel Energy-Absorbing Restrainers and their Incorporation into Nuclear Power Plants for Ennanced Safety "Vol 3): Testing of Communical Steels in Low-Typle Torsional Patigue," by P. Scontty, E.R. Parker, E. Jongewaard and M. Droky |
| uca boki+333 | "The Jeslan of Steel Energy-Absorbing Restrainers and their Incorporation into Nuclear Power Plarts for Ennanced Safety (Vol 4): Shaking Table Tests of Piping Systems with Energy-Absorbing Restrainers," by S.F. Stiemer and W.J. Bodden - Sept. 1980 |
| CCB (FEPC-30) 34 | "The Design of Steel Energy-Absorbing Restrainers and their Incorporation into Nuclear Power Plants for Enhanced Safety (Vol[5]: Summary Report," by P. Spencer |
| 125 FERCHEL 35 | "Experimental Testing of an Energy-Absorbing Base Isolation System," by J.M. Yelly, M.S. Skinner and K.E. Beuske - October 1980(1986) 154 0721A04 |
| VIB LERCHBO/36 | "Simulating and Analyzing Artificial Non-Stationary Earthquake Ground Motions," by R.F. Nau, R.M. Cliver and K.S. Pister - October 1980(9881-153-397)A04 |
| COB/SEPC+30/37 | "Earthquake Engineering at Berkeley - 1980," - Sept. 1900(PB61 205 s74)AC9 |
| UCB-EERC-ADV18 | "Irelastic Seismic Analysis of Large Panel Buildings," by V. Schricker and J.H. Powell - Sept. 1981 (PB01 154 338)Aly |
| 1008/ EERC=40-19 | "Oynamic Response of Embankment, Concrete-Gravity and Arch Dams Including Hydrodynamic Interaction," By J.F. Hall and A.K. Chopra - October 1980(PB81 152 324)all |
| UCB/EERC-80/40 | "Inelastic Buckling of Steel-Strute Under Cyclic Load Reversal," by R.G. Black, W.A. Wenger and E.P. Popov - October 1980(P881-154-312)A08 |
| 009/EERC-30/41 | "Influence of Site Characteristics on Building Damage During the October 3, 1974 Lima Earthquake," by P. Repetto, L. Arango and H.B. Seed - Sept. 1980/P:01 161 7391AD5 |
| UCB/EERC+30/42 | "Evaluation of a Shaking Table Test Program on Response Behavior of a Two Story Reinforced Concrete Frame," by J.M. Blondet, R.W. Clough and S.A. Mahin |
| CCB/EERC+50/43 | "Modelling of Soil-Structure Interaction by Finite and Infinite Elements," by F. Medina - December 1980(PB81 229 270)204 |
| UCB/EERC-81/01 | "Control of Seismic Response of Fiping Systems and Other Structures by Base Isolation," edited by J.M. Kelly - January 1981 (PB81 200 735)AG5 |
| UCB/EERC-81/02 | "OPINSR - An Interactive Software System for Optimal Design of Statically and Dynamically Loaded Structures with Nonlinear Response," by M.A. Bhatti, V. Ciampi and K.S. Pister - January 1981 (PB81 218 851)A09 |
| UCB/EERC-81/03 | "Analysis of Local Variations in Free Field Seismic Ground Motions," by JC. Chen, J. Lysmer and H.B. Seed - Vinusky 1981 (AD-A099508)ALD |
| UCB/EEPC-81/04 | "Inelastic Structural Modeling of Braced Offshore Platforms for Seismic Loading," by V.A. Zayas, PS.B. Shing, S.A. Mahin and E.P. Popov - January 1981(PB82-138-777)A07 |
| UCB/EERC-81/05 | "Dynamic Response of Light Equipment in Structures," by A. Der Kiureghien, J.L. Sackman and B. Nour- Omid - April 1981 (PB81 218 497)A04 |
| UCB/EERC-81/06 | "Preliminary Experimental Investigation of a Broad Base Liquid Storage Tank," by J.G. Bouwkamp, J.P. Kollegger and R.M. Stephen - May 1991/PB82 140 385)A03 |
| 'CB/EERC-81/07 | "The Seismic Resistant Design of Reinforced Concrete Coupled Structural Halls," by A.Z. Aktan and V.V. Bertero - June 1981(PB82 113 358)All |
| UCB/22RC-81/08 | "The Undrained Shearing Remistance of Cohemive Soils at Large Deformations," by M.R. Pyles and H.B. Seed - August 1981 |
| UCB/EERC-81/09 | "Experimental Behavior of a Spatial Piping System with Steel Energy Absorbers Subjected to a Simulated Differential Seismic Input," by S.F. St.emer, W.G. Godden and J.H. Kelly - July 1981 |

UCB/EERC-01/10 "Evaluation of Seismic Design Provisions for Mesonry in the United States," by B.I. Sveinscon, R.L. Mayes and H.D. MCNiven - August 1961 (PB82 166 075) AC UCB/EERC-81/11 "Two-Limensional Hybrid Hodelling of Soil-Structure Interaction," by T.-J. Tzong, S. Supta and J. Penzien - August 1981(PB82 142 118)A04 CCB/EERC-01,12 "Studies on Effects of Infills in Seismir Resistant R/C Construction," by 5. Brokker and V.V. Bertero -September 1981 (2582 166 1901A02 UCB/EERC-81/13 "Linear Models to Predict the Nonlinear Seismic Behavior of a One-Story Steph Frame," by H. Valdimarsson, A.H. Shah and H.D. McNiven - September 1981 (PB82 138 793)AC7 UCE ELEC-cl/14 "TLUMES" A Computer Program for the Three-Dimensional Dynamic Analysis of Larth Dans," by 1. Kagawa. L.N. Menia, R.E. Seed and J. Lysmer - September 1981 (PB82 139 940) 406 ULE DEFI-51 11 "Inree Dimensional Dynamic Response Analysis of Earth Dams," by LLH. Megia and H.F. seed - September 1981 (FBG2 137 274)A12 UCP EERC-81/10 "Experimental Study of Lead and Elastomeric Dampers for Base Isolation Systems," by J.K. Kelly and 5.E. Hodder - October 1981 (PB82 166 162)A05 UCH.EEFC-81/17 "The Influence of Base Isolation on the Seismic Response of Light Secondary Equipment," by J.M. Kelly -April 1981 (PB82 255 266) A04 UCB EERC-81 16 "Studies on Evaluation of Shaking Table Response Analysis Procedures," by J. Marcial Blondet - November 1961 (PB82 197 278) A10 UCB EFRC=S1 19 "DELIGHT.STRUCT: A Computer-Aided Design Environment for Structural Engineering," sy N.J. Ballino, N.S. Fister and E. Polak - December 1981 (PB82 218 496)A07 UCH/EERC-81/20 "Optimal Design of Seismic-Resistant Planar Steel Frames," by R.J. Balling, V. Ciargi, K.S. Hister and E. Folak - December 1961 (PB82 220 179)R07 UCB/EERC-82/01 "Dynamic Behavior of Ground for Seismic Analysis of Lifeline Systems," by T. Sato and A. Der Hour music -January 1982 (PB82 218 926)A05 UCE/EERC-82/02 "Shaking Table Tests of a Tubular Steel Frame Model," by Y. Ghansat and R. W. Clough - January 1963 (PB62 220 1611A07 UCB/EERC-82/03 "Behavior of a Piping System under Seismic Excitation: Experimental Investigations of a Spatial Figung System supported by Mechanical shock Arrestors and Steel Energy Absorbing Devices under Seis ic Excitation," by S. Schneider, H.-M. Lae and W. G. Godden - May 1982 (FBB3 172 544)A09 "New Approaches for the Dynamir, Analysis of Large Structural Systems," by E. L. Wilson - June 1982 (PB83 148 0801A05 UCB/EERC-82/04 UCB/EERC-82/05 "Model Study of Effects of Damage on the Vibration Properties of Steel Offshore Platforms," by F. Shahriver and J. G. Bouwkamp - June 198/ (9883 148 742)Alo UCB/EERC-82/06 "States of the Art and Practice in the Optimum Seismic Design and Analytical Response Prediction of F/C Frame-Wall Structures," by A. E. Aktan and V. V. Bertero - July 1982 (PB83 147 736)A05 UCB/EERC-82/0" "Furthar study of the Earthquake Response of a Broad Cylindrical Liquid-Storage Tank Model," by G. C. Manos and R. M. Clough - July 1902 (PB83 147 744)all UCB/EERC-82/08 "An Evaluation of the Design and Analytical Seignic Response of a Seven Story Reinforced Concrete Frame - Wall Structure," by F. A. Charney and V. V. Bertero - July 1982(2005) 157 628)A09 UCB/EER4-82/09 "Fluid-Structure Interactions: Added Mass Computations for Incompressible Fluid," by J. S.-H. Kuo - August 1982 (FB83 156 281)a07 UCE/EERJ-82/10 "Joint-Opening Nonlinear Mechanism: Interface Smeared Crack Model," by J. S -H. Kuo -August 1982 (PB83 149 195)A05 UCB/EERC-02/11 "Dynamic Response Analysis of Techi Dam," by R. W. Clough, R. M. Stephen and J. S.-H. Kuo -August 1982 (PBB3 147 496) A06 UCB/FERC-82/12 'Prediction of the Seismic Responses of R/C Frame-Coupled Wall Structures," by A. E. Aktan, V. V. Bertero and M. Pinzza - August 1982 (PB83 149 203)A09 UCB/EERC-82/13 "Preliminary Report on the SMART 1 Strong Motion Array in Taiwan," by H. A. Bolt, C. H. Loh, J. Penzien, Y. B. Tsai and Y. T. Yeh - August 1982 (PB83 159 400)Alo UCB/TERC-82/14 "Shaking-Table Studies of an Eccentrically X-Braced Steel Structure," by M. S. Yang - September 1962 UCB/ 12RC-82/15 "The Performance of Stairways in Earthquakes," by C. Roha, J. N. Axley and V. V. Bertero - September 1962 (PS#3 157 693) A07 UTB/EERC-82/16 "The penavior of Submerged Multiple Bodies in Earthquakes," by M.-G. Liac - gapt. 1982 (9883 158 709)A07

- UCB/EERC-82/17 "Effects of Concrete Types and Loading Conditions on Local Bond-Blip Relationships, by A. D. Cowell, E. F. Popov and V. V. Bertero - September 1982 (pm83 153 577)A04
- UCB/EERC-82/18 "Nechanical Behavior of Shear Wall Vertical Boundary Nembers: An Experimental Investion of the Man N. T. Wagner and V. V. Bertero October 1982 (PB83 159 764)A05
- UCB/EERC-02/19 "Experimental Studies of Multi-support Seismic Loading on Piping Systems," by J. N. Krlly and A. D. Cowell - November 1902
- UCB/EERC-82/20 "Generalized Plastic Hinge gencepts for 3D Bear-Column Elements," by P. F.-S. Chen and G. H. Powell -November 1982
- UCB/EERC-82/21 "ANSR-III: General Purpose Computer Program for Nonlinear Structural Analysis," by C. V. Oughourlian and G. H. Powell - November 1982
- UCB/EERC-02/22 "Solution Strategies for Statically Loaded Nonlinear Structures," by J. N. Simons and G. H. Powell -November 1982
- UCB/EERC-82/23 "Analytical Model of Deformed Bar Anchorages under Generalized Excitations," by V. Cismpi, R. Eligehausen, V. V. Bertero and E. P. Popov November 1982 (PB8] 169 532)A06
- UCB/EERC-82/24 "A Mathematical Model for the Response of Masonry Walls to Dynamic Excitations," by H. Sucuoğlu, Y. Mengi and H. D. McNiven - November 1962 (PEB3 169 011)A07
- UCB/EERC-82/25 "Earthquake Response Considerations of Broad Liquid Storage Tanks," by F. J. Cambra November 1982
- UCB/EERC-82/26 "Computational Models for Cyclic Plasticity, Rate Dependence and Creep," by B. Mosaddad and G. H. Powell - November 1982
- UCB/EERC-82/27 "Inelastic Analysis of Piping and Tubular Structures," by M. Mahasuverschai and G. H. Powell November 1982
- UCB/EERC-93/01 "The Economic Peasibility of Seismic Pehabilitation of Buildings by Base Isolation," by J. M. Kelly -January 1983
- UCB/EERC-83/02 "Seismic Moment Connections for Moment-Resisting Steel Prames," by E. P. Popov January 1983
- UCB/EERC-83/03 "Design of Links and Beam-Lo-Column Connections for Echentrically Braced Steel Frames," by E. P. Popov and J. U. Melley - January 1983
- UCB/EERC-83/04 "Numerical Techniques for the Evaluation of Soil-Structure Interaction Effects in the Time Domain," by E. Bayo and E. L. Wilson - February 1983
- CCS/EERC-03/05 "A Transducer for Measuring the Internal Porces in the Columns of a Frame-Wall Reinforced Concrete Structure," by R. Sause and V. V. Bertero - Nay 1983
- UCB/EERC-83/06 "Dynamic Interactions between floating Ice and Offshore Structures," by P. Crotaau May 1983
- UCB/EERC-83/07 "Dynamic Analysis of Multiply Tuned and Arbitrarily Supported Secondary Systems," by T. Igusa and A. Der Klureghian - June 1983